INVESTIGATIONS INTO THE MOLECULAR PATHOGENESIS

OF ESSENTIAL THROMBOCYTHAEMIA

Jonathan Richard Lambert

UCL Cancer Institute

University College London

A thesis submitted for the degree of Doctor of Philiosphy

Declaration

I, Jonathan Lambert, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Signed:

Date:

Abstract

In order to explore the phenotypic heterogeneity of the myeloproliferative neoplasm essential thrombocythaemia (ET), the role of the JAK2 mutation V617F in the pathogenesis of the disease was investigated, in particular its relationship to myeloid clonality. The clinical, haematological and molecular characteristics of 133 ET patients were studied. JAK2 V617F was detected in 55 (41%) patients; a clonal X-chromosome inactivation pattern (XCIP) was found in 24 (39%) of the 62 evaluable female patients. There was no association between JAK2 mutational status and XCIP status or thrombotic risk, but higher JAK2 V617F mutant levels were noted in patients who had a thrombosis. A trend towards a higher thrombotic rate was observed in patients whose XCIP was clonal. In 10 untreated JAK2 V617F-positive ET patients, JAK2 WT thrombopoiesis was not suppressed despite the presence of a thrombocytosis, suggesting that the regulation of JAK2 WT thrombopoiesis was abnormal. Eleven patients were screened for the presence of more than one JAK2 V617F-positive population using an exonic SNP located near the mutation. In ten (91%) of these the mutation appeared to have been independently acquired on at least two occasions. Furthermore, XCIP analysis of JAK2 V617Fpositive erythroid colonies from six ET patients revealed that in one patient the V617F-positive populations were not derived from a single clonal population. An association between the reported JAK2 haplotype (known as '46/1') and JAK2 V617F-positive ET patients was observed in the cohort studied. Methylation studies indicated that this haplotype introduced additional methylated sites near to the mutation locus, which may potentially affect conformation of the DNA and mutability of the JAK2 locus. Together, the studies reported in this thesis suggest that JAK2 V617F is not the initiating event at least in some cases of ET, and that its presence does not invariably indicate the presence of a monoclonal disorder.

Acknowledgements

I would like to thank Professors Rosemary Gale and David Linch for their guidance and enthusiasm with this project. With their help I have begun to understand how molecular biology may be used to answer fundamental questions in clinical medicine. In addition, I am very grateful to the other members of the laboratory who have helped me throughout these studies, and without whose advice and friendship my research would have been much slower and less enjoyable. I am also indebted to the patients who have provided samples for the tissue bank over the years, and to the doctors who have referred them. Finally, I cannot begin to express my gratitude to my family and friends for their patience, encouragement and support over the past four years.

Table of contents

Tit	tle page		1
De	eclaratio	n	2
Ał	ostract		3
Ac	knowle	dgements	4
Та	ble of c	ontents	5
Lis	st of figu	res	14
Lis	st of tab	les	17
Cc	ommonly	y used abbreviations	20
1	Introdu	lction	22
	1.1	Essential thrombocythaemia and the myeloproliferative neoplasms	22
	1.2	Physiological megakaryopoiesis and platelet production	26
	1.2.1	MK development and platelet biogenesis	26
	1.2.2	Hormonal regulation of megakaryopoiesis and thrombopoiesis	28
	1.2.3	Molecular signalling pathways involved in thrombopoiesis	30

	1.2.4	Transcriptional control of megakaryopoiesis and thrombopoiesis
	1.3	ET – pathology and clues to pathogenesis
	1.3.1	Cytological and histological changes
	1.3.2	Cytogenetics
	1.3.3	TPO levels and c-Mpl expression
	1.3.4	Growth factor hypersensitivity in ET progenitors and possible causes
	1.3.5	Clonality in ET
	1.3.6	PRV-1 expression
	1.3.7	<i>JAK2</i> V617F50
	1.3.8	Other genetic mutations in ET 54
	1.4	Aims of the project
2	Materia	als and Methods
	2.1	Sample preparation 60
	2.1.1	Reagents
	2.1.2	Buffers62
	2.1.3	Purification of neutrophils, platelets and T-lymphocytes62
	2.1.4	RNA extraction
	2.1.5	Genomic DNA extraction

	2.2 M	olecular biology6	5
	2.2.1	Reagents 6	5
	2.2.2	Buffers	7
	2.2.3	Polymerase chain reaction (PCR)	7
	2.2.3.1	BIOTAQ™ PCR	8
	2.2.3.2	GoTaq® Hot Start PCR6	8
	2.2.3.3	Optimase [®] PCR 6	9
	2.2.3.4	Phusion [®] PCR	9
	2.2.4	Restriction enzyme digestion7	0
	2.2.5	Reverse-transcription PCR (RT-PCR) 7	0
	2.2.6	Agarose gel electrophoresis	'1
	2.2.7	Fragment analysis of PCR products and restriction enzyme digestion products 7	1
	2.2.8	DNA sequencing7	2
	2.2.9	LB-carbenicillin culture medium and plates7	3
	2.2.10	TA cloning7	3
3	Clinical an	d molecular characteristics of the ET patient cohort7	′5
	3.1 Int	roduction	5
	3.1.1	Diagnostic criteria for ET7	6'

3.1.2	Impact of clinical features at presentation on prognosis	77
3.1.3	Impact of molecular characteristics on prognosis	83
3.1.4	Risk stratification and therapy for ET patients	
3.1.5	Choice of cytoreductive agent	
3.1.6	Aims	
3.2	Patients, materials and methods	
3.2.1	Patient recruitment and sample collection	
3.2.2	Clinical and demographic characteristics	
3.2.3	JAK2 V617F status	
3.2.4	JAK2 V617F mutant level quantification in neutrophil gDNA	
3.2.5	X-chromosome inactivation pattern (XCIP) analysis	
3.2.6	MPL mutation screening	101
3.2.6	6.1 Generation of a MPL W515L-positive control by site-directed mutage	nesis . 101
3.2.6	6.2 Screening for MPL exon 10 mutations using dHPLC	104
3.3	Results	106
3.3.1	Overall results and analysis by clinical risk factors at presentation	106
3.3.2	Analysis by XCIP status	108
3.3.3	Analysis of JAK2 V617F status	110

	3.3.4	Correlation between XCIP status and JAK2 mutational status 113	3
	3.3.5	MPL exon 10 mutation screening 11	5
	3.4	Discussion	5
4	JAK2 V	617F-positive and JAK2 WT thrombopoiesis in ET12	B
	4.1	Introduction	8
	4.1.1	The long-term stability of the V617F-positive population in ET	3
	4.1.2	JAK2 V617F mutant levels in neutrophils and platelets	1
	4.1.3	Aims	2
	4.2	Patients, materials and methods	2
	4.2.1	JAK2 V617F mutant levels quantification in neutrophil and platelet RNA 132	2
	4.2.2	Calculation of the numbers of V617F-positive and WT platelets	3
	4.3	Results	3
	4.3.1	Mutant allele levels in neutrophil DNA and RNA and platelet RNA	3
	4.3.2	Calculation of WT and mutant platelet counts13	5
	4.4	Discussion	7
5	Testing	for more than one <i>JAK2</i> V617F-positive population in patients with mutant-	
	positiv	e ET143	3
	5.1	Introduction	3

5.1.1	Is V617F-positive ET a clonal neoplasm initiated by JAK2 V617F?	144
5.1.2	JAK2 V617F and clonality in ET	146
5.1.3	Aims	148
5.2 Pa	atients, materials and methods	148
5.2.1	Patient selection	148
5.2.2	Determining JAK2 status	149
5.2.2.1	Screening for JAK2 V617F using gDNA	149
5.2.2.2	Screening for JAK2 V617F using cDNA	149
5.2.3	JAK2 exon 19 SNP (rs2230724) genotyping	149
5.2.3.1	Exon 19 SNP genotyping using gDNA	149
5.2.3.2	Exon 19 SNP genotyping using cDNA	151
5.2.4	JAK2 exon 19 SNP (rs2230724) genotyping of V617F-positive alleles by PCR-	
	restriction enzyme digestion-PCR	151
5.2.5	JAK2 exon 19 SNP genotyping of V617F-positive alleles by TA cloning	153
5.2.6	Using a JAK2 intron 14 SNP (rs12343687) to genotype V617F-positive alleles.	154
5.2.6.1	Genotyping patients for intron 14 rs12343687 SNP status	155
5.2.6.2	JAK2 intron 14 SNP genotyping of V617F-positive alleles by PCR-restriction	
	enzyme digest-PCR	157
5.2.6.3	JAK2 intron 14 genotyping of V617F-positive alleles by TA cloning	157

	5.2.7	X-chromosome inactivation analysis of erythroid burst-forming units (BFU-Es)
	5.2.7.1	Culturing BFU-Es from peripheral blood158
	5.2.7.2	2 Identification of V617F-positive colonies
	5.2.7.3	Inactivated X-chromosome analysis of V617F-positive colonies
	5.3 R	esults
	5.3.1	JAK2 exon 19 SNP (rs2230724) genotyping in haematologically normal controls
		and ET patients 161
	5.3.2	JAK2 exon 19 SNP genotyping of V617F-positive alleles using SNP rs2230724 162
	5.3.3	JAK2 intron 14 SNP genotyping of V617F-positive alleles using SNP rs12343687
	5.3.4	Genotyping of peripheral blood erythroid burst-forming units (BFU-Es)
	5.3.5	Calculation of neutrophil XCIP based on T-cell and BFU-E XCIPs and mutant levels
	5.4 D	iscussion
6	Investiga	tions into the 46/1 haplotype182
	6.1 lr	ntroduction
	6.1.1	Familial clustering of ET 182
	6.1.2	The 46/1 haplotype 185

6.1.3	Possible mechanisms underlying the association between the 46/1 haploty	/pe
	and MPNs	190
6.1.4	Aims	192
6.2 P	Patients, materials and methods	193
6.2.1	Screening JAK2 V617F-positive patients for the 46/1 haplotype using 5 JAK	2
	SNPs	193
6.2.2	Screening for the presence of the 46/1 haplotype using SNP rs12343867	193
6.2.3	Expression analysis of the 46/1 and non-46/1 alleles	196
6.2.3.	.1 Validation of the technique using neutrophil gDNA	196
6.2.3.	.2 Expression analysis in neutrophil cDNA	197
6.2.4	Analysis of SNP rs505534	198
6.2.5	Methylation analysis	198
6.2.5.	.1 Bisulphite treatment	200
6.2.5.	.2 Qualitative analysis of SNP rs12343867 methylation status	200
6.2.5.	.3 Quantitative analysis of SNP rs12343867 methylation status	201
6.2.5.	.4 Quantitative analysis of methylation status near to SNP rs12340895	202
6.3 F	Results	202
6.3.1	Screening <i>JAK2</i> V617F-positive patients for the 46/1 haplotype	202

	6.3.2	Genotyping normal controls and ET patients for the presence of the 46/1	
		haplotype using SNP rs12343867 20)3
	6.3.3	Expression analysis of the 46/1 and non-46/1 alleles)7
	6.3.4	Analysis of SNP rs505534 20)9
	6.3.5	Methylation analysis of normal controls and ET patients	10
	6.3.	1 Qualitative analysis of SNP rs12343867 methylation status 21	10
	6.3.	2 Quantitative analysis of SNP rs12343867 methylation status	10
	6.3.	3 Quantitative analysis of methylation status near to SNP rs12340895 21	15
	6.3.	4 Comparison of methylation data for SNP rs12343867 and SNP rs12340895.21	15
	6.4	Discussion	17
7	Conclu	ons and future directions 22	26
	7.1	Summary of findings 22	26
	7.2	uture directions	31
	7.3	Overall conclusions	34
Aj	opendix	Primer sequences 23	35
A	opendix	Publications arising from experiments reported in this thesis	37
Re	eference		38

List of figures

Figure 1.1	Haemopoietic cell differentiation pathway27
Figure 1.2	Principal JAK2-mediated MK signalling pathways
Figure 1.3	Peripheral blood smear from a patient with ET and from a haematologically
	normal control
Figure 3.1	Method for PCR of JAK2 exon 14 followed by AfIIII digestion to distinguish
	between JAK2 WT and V617F alleles95
Figure 3.2	Principle of XCIP analysis using HUMARA
Figure 3.3	Fragment analysis chromatogram of HUMARA PCR products from a single patient
Figure 3.4	Site-directed mutagenesis to create a MPL W515L-mutant positive control 102
Figure 3.5	Theoretical melting curves of MPL exon 10 PCR product generated by
	Transgenomic Navigator software 105
Figure 3.6	dHPLC chromatogram traces of MPL exon 10 PCR products amplified using normal
	control DNA and a 10% W515L-positive control as template 115
Figure 3.7	dHPLC chromatogram traces and corresponding direct sequencing chromatogram
	traces of samples with MPL W515 mutations116
Figure 4.1	JAK2 V617F mutant levels in neutrophil DNA and RNA, and platelet RNA in 10 ET
	patients
Figure 4.2	Calculation of JAK2 wild-type and V617F-positive platelets in 10 ET patients 136

Figure 4.3	Correlation between platelet counts and JAK2 V617F mutant level 137
Figure 5.1	Schematic representation of the technique used to determine the JAK2 exon 19
	SNP (rs2230724) genotype of V617F positive-mutant alleles
Figure 5.2	Schematic representation of the TA cloning technique used to determine the JAK2
	exon 19 SNP (rs2230724) genotype of <i>JAK2</i> V617F mutant-positive alleles using
	cDNA
Figure 5.3	Schematic representation of JAK2 exons 14 and 15 showing primers used to
	genotype V617F-positive alleles using intron 14 SNP (rs12343687) 156
Figure 5.4	JAK2 exon 14 ARMS PCR to detect V617F (based on Jones et al, 2005) 160
Figure 5.5	JAK2 exon 19 SNP genotype analysis of V617F mutant alleles in 11 ET patients 166
Figure 5.6	JAK2 exon 19 SNP genotype analysis of V617F mutant alleles in 5 PV patients 167
Figure 5.7	JAK2 intron 14 SNP (rs12343687) genotype analysis of V617F mutant alleles in 6
	ET patients
Figure 6.1	JAK2 SNPs investigated in the initial haplotype studies
Figure 6.2	Schematic representation of JAK2 exons 10 to 16 showing the five SNPs studied
	and their corresponding PCR primers 195
Figure 6.3	Schematic representation of PCR of JAK2 exon 19 followed by BstNI digestion to
	determine expression of A and G alleles at SNP rs2230724 197
Figure 6.4	Effect of bisulphite treatment on possible CpG dinucleotides that involve SNPs
	rs12343867 and rs12340895 (JAK2 intron 14) 199
Figure 6.5	Quantification of the relative levels of SNP rs2230724 alleles A/G

Figure 6.6	Sequences of normal control DNA showing the region around SNP rs12343867
	(C/T)
Figure 6.7	Calibration curve of pyrosequencing analysis of cytosine methylation at SNP
	rs12343867 using artificially created mixes 211
Figure 6.8	Pyrosequencing of C/T SNP rs12343867 following bisulphite treatment of
	neutrophil DNA 212
Figure 6.9	Concordance between duplicate pyrosequencing analyses of the percentage
	methylation at SNP rs1234386 in normal controls and ET patients 213
Figure 6.10	Pyrosequencing analysis of methylation at SNP rs12343867 in normal controls and
	ET patients
Figure 6.11	Pyrosequencing analysis of methylation at SNP rs12340895 in normal controls and
	ET patients

List of tables

Table 1.1	2008 WHO Criteria for the diagnosis of the myeloproliferative neoplasms
Table 1.2	Studies examining TPO levels in ET
Table 1.3	Mutations reported in familial thrombocythaemia
Table 1.4	Studies of XCIP in ET 46
Table 1.5	PRV-1 studies in ET
Table 1.6	Initial studies reporting JAK2 V617F51
Table 1.7	Frequency of JAK2 V617F in ET – published studies to date
Table 1.8	MPL mutations reported in sporadic ET55
Table 3.1	PVSG Criteria for the diagnosis of ET76
Table 3.2	Recent studies analysing life expectancy in ET79
Table 3.3	Studies assessing conventional risk factors for thrombosis in ET
Table 3.4	Published studies reporting presenting characteristics of ET patients according to
	<i>JAK2</i> V617F status
Table 3.5	Studies addressing the association between JAK2 V617F and thrombosis (venous
	and arterial)
Table 3.6	Clinical and demographic characteristics of ET patients included in study 107

Table 3.7	Effect of presenting age and blood counts on occurrence of thrombosis at any
	stage
Table 3.8	Clinical and demographic characteristics of patients studied according to their
	XCIP status 109
Table 3.9	Clinical and demographic characteristics studied according to JAK2 status 111
Table 3.10	JAK2 V617F-positive patients with an interpretable XCIP: the effect of mutant level
	on neutrophil clonality status114
Table 3.11	Summary of studies examining association between myeloid XCIP status and
	thrombotic rate
Table 3.12	JAK2 V617F mutant levels in the MPNs121
Table 5.1	Incidence of JAK2 exon 19 SNP rs2230724 genotypes in normal controls, ET and
	PV patients
Table 5.2	JAK2 exon 19 and intron 14 SNP analysis of PCR products from 11 patients with
	JAK2 V617F mutant-positive ET165
Table 5.3	Summary of XCIP analysis of V617F-positive BFU-Es from 6 ET patients 170
Table 5.4	Prediction of neutrophil XCIP in 6 patients, based on their CD3-positive XCIP,
	mutant levels in neutrophil DNA and the XCIP of V617F-positive colonies 177
Table 6.1	PCR and restriction digestion conditions used to determine the genotype of five
	SNPs within the <i>JAK2</i> gene
Table 6.2	JAK2 SNP genotypes of 11 JAK2 V617F-positive ET patients
Table 6.3	SNP rs12343867 genotypes of the normal controls, ET and PV patients studied 204

Table 6.4	Frequency of rs12343867 alleles in each group studied and comparison with local
	and WTCCC controls 206
Table 6.5	SNP rs505534 genotypes of 114 ET patients 209
Table 6.6	Published studies examining the association between the 46/1 haplotype and ET
	and PV 220

Commonly used abbreviations

AD	Autosomal dominant
AML	Acute myeloid leukaemia
AR	Autosomal recessive
bp	Base pairs
BSA	Bovine serum albumin
CML	Chronic myeloid leukaemia
ddH₂O	Double de-ionised water
dHPLC	Denaturing high performance liquid chromatography
DNA	Deoxyribonucleic acid
	c=complementary
	g=genomic
EEC	Endogeneous erythroid colonies
EPO	Erythropoietin
ET	Essential thrombocythaemia
FCS	Foetal calf serum
Hb	Haemoglobin
HSC	Haemopoietic stem cell
HUMARA	Human androgen receptor assay
JAK2	Janus kinase 2
JΗ	Janus homology
LD	Linkage disequilibrium
LOH	Loss of heterozygosity
MDS	Myelodysplastic syndrome

- MF Myelofibrosis
- MK Megakaryocyte
- MNC Mononuclear cell
- MPL Myeloproliferative leukaemia virus
- MPN Myeloproliferative neoplasm
- PCR Polymerase chain reaction
- PNH Paroxysmal nocturnal haemoglobinuria
- PV Polycythaemia vera
- PVSG Polycythemia Vera Study Group
- RNA Ribonucleic acid
- SNP Single nucleotide polymorphism
- TET2 Ten-eleven translocation 2
- TF Tissue factor
- TPO Thrombopoietin
- WCC White cell count
- WHO World Health Organisation
- WT Wild-type
- XCIP X-chromosome inactivation pattern

1 Introduction

1.1 Essential thrombocythaemia and the myeloproliferative neoplasms

Essential thrombocythaemia (ET) is a chronic haematological disorder characterised by the sustained overproduction of megakaryocytes (MKs) and platelets in the absence of a recognisable physiological stimulus. It is a rare condition, with an annual incidence of approximately 0.8 new cases per 100,000 population in the UK (McNally *et al*, 1997), though this figure may be as high as 2.5 per 100,000 in some regions (Mesa *et al*, 1999), and there is evidence that these rates are increasing (Girodon *et al*, 2009; Johansson *et al*, 2004). The median age at diagnosis is 65 – 70 years, but it may present at any age of life, and there is a female-to-male preponderance of approximately 2-to-1 (Girodon *et al*, 2009; Jensen *et al*, 2000; Landgren *et al*, 2008).

Although it was initially described in 1934 as a haemorrhagic thrombocythaemia (Epstein and Goedel 1934), its most significant clinical sequelae are thrombotic episodes, which may be lifethreatening. Untreated, the rate of thrombotic events in one series was 6.6% per patient-year, though this varied markedly between 1.7% and 31.4% per patient-year depending on a number of risk factors, with a predominance of arterial events (Cortelazzo *et al*, 1990). Haemorrhagic complications are rarer, and are associated with extreme thrombocytosis (platelets greater than 1500x10⁹/L). ET may also transform to the more serious myeloproliferative neoplasm (MPN) myelofibrosis (MF), or to acute myeloid leukaemia (AML), but in the short term the risk of this is small – approximately 5-8% over 10 years, and 1-3% over 10 years, respectively (Cervantes *et al*, 2002; Palandri *et al*, 2009a). For older patients the diagnosis is unlikely to have a significant impact on their life expectancy, but for younger

patients the same may not be true (Bazzan *et al*, 1999; Mesa *et al*, 1999; Passamonti *et al*, 2004; Wolanskyj *et al*, 2006). Cytoreductive therapy reduces the incidence of vascular complications in high-risk patients (Cortelazzo *et al*, 1995) but may increase the risk of malignant transformation (Finazzi *et al*, 2000; Randi *et al*, 2000).

As long ago as 1951, Dameshek recognised the similarities between ET, polycythaemia vera (PV), MF, and chronic myeloid leukaemia (CML), and suggested that they might represent different manifestations of the same 'myeloproliferative syndrome' (Dameshek 1951). He went on to speculate about the existence of a 'myelostimulatory factor', which he believed might be a steroid hormone. Although a number of rarer conditions were later classified as myeloproliferative disorders (renamed MPN by the World Health Organisation, WHO, in 2008), ET, PV, MF and CML still represent the great bulk of this diagnostic category in terms of prevalence and clinical burden.

The finding in 1960 of a recurrent cytogenetic abnormality in most patients with CML – a reciprocal translocation between chromosome 9 and chromosome 22 [t(9;22)(q34.1;q11.2)], known as the Philadelphia chromosome (Nowell and Hungerford 1960) – greatly advanced our understanding of this disorder and led to the recognition that it represents a distinct entity from the other so-called Philadelphia-negative MPNs. It also resulted in significant improvements in the diagnosis and treatment of CML, inspiring intensive efforts to find equivalent abnormalities in ET, PV and MF.

To facilitate research efforts and to help standardise the diagnosis and treatment of patients with MPNs, the Polycythemia Vera Study Group (PVSG) was formed in the 1960s. Its principal role was to coordinate clinical trials of patients with MPNs and, in order to provide consistency between clinical studies, the PVSG introduced formal diagnostic criteria for the three main disorders, which have subsequently been refined by the WHO (Table 1.1). The most recent

Table 1.1 2008 WHO Criteria for the diagnosis of the myeloproliferative

neoplasms

		Polycythaemia vera	Essential thrombocythaemia	Myelofibrosis
Major criteria	1	Hb >18.5 g/dL (men) >16.5 g/dL (women) or other evidence of increased red cell volume* <i>or</i> elevated red cell mass >25% above mean normal predicted value	Sustained platelet count ≥450x10 ⁹ /L	MK proliferation and atypia accompanied by reticulin and/or collagen fibrosis or in the absence of reticulin fibrosis, MK changes must be accompanied by increased marrow cellularity, granulocytic proliferation and often decreased erythropoiesis (i.e. pre-fibrotic PMF)
	2	Presence of JAK2V617F or similar mutation (e.g. JAK2 exon 12)	Bone marrow biopsy specimen showing proliferation mainly of the megakaryocytic lineage with increased numbers of enlarged, mature MKs. No significant increase or left- shift of neutrophil granulopoiesisor erythropoiesis	
	3		Not meeting WHO criteria for CML, PV, PMF, MDS or other myeloid neoplasm	Not meeting WHO criteria for CML, PV, MDS, or other myeloid neoplasm
	4		Demonstration of JAK2V617F or other clonal marker or no evidence of reactive thrombocytosis	Demonstration of JAK2V617F or other clonal marker or no evidence of reactive marrow fibrosis
Minor criteria	1	Bone marrow biopsy showing hypercellularity for age with trilineage growth (panmyelosis) with prominent erythroid, granulocytic, and megakaryocytic proliferation		Leukoerythroblastosis
	2	Subnormal serum Epo level		Increased serum LDH
	3	EEC growth		Anaemia
	4			Palpable splenomegaly

Based on Tefferi and Vardiman (2008).

The diagnosis of PV requires fulfilment of either both major criteria and one minor criterion or the first major criterion and 2 minor criteria. The diagnosis of ET requires fulfilment of all 4 major criteria. The diagnosis of PMF requires fulfilment of all 3 major criteria and 2 minor criteria.

 * Hb or Hct >99th percentile of reference range for age, sex or altitude of residence; or Hb >17 g/dL (men), >15 g/dL (women) if associated with sustained increase of ≥2g/dL from baseline that cannot be attributed to correction of iron deficiency WHO definition of ET requires all four major diagnostic criteria to be fulfilled: the presence of a sustained thrombocytosis, defined as a platelet count of \geq 450x10⁹/L; the presence in the bone marrow of megakaryocytic proliferation with increased numbers of mature forms and no increase in immature granulocytic and erythroid precursors; no evidence of another myeloid disorder, i.e. MF, myelodysplasia (MDS), CML or PV; and either the detection of a specific clonal marker or no evidence of a reactive thrombocytosis.

Despite exhaustive studies, cytogenetic analysis of the Philadelphia-negative MPNs has not yielded the same breakthrough as in CML, and so numerous other approaches have been adopted to elucidate the pathogenic processes underlying them. The discovery that PV was characterised by the presence of endogenous erythroid colonies (EECs) – progenitors which proliferate in vitro in the absence of erythropoietin (EPO) – and that endogenous megakaryocytic colonies were present in some cases of ET, raised the possibility that aberrations in the regulation of haemopoiesis might contribute to their pathogenesis (see section 1.3.4). Consequently, the major haemopoietic growth factors, EPO, granulocyte-colony stimulating factor (G-CSF), and thrombopoietin (TPO) were systemically analysed for the presence of genetic mutations, transcriptional defects and disruption of the normal homeostatic mechanisms. Unfortunately, no abnormality proved sufficiently common or specific to an individual disease to offer a major insight into its pathogenesis or be of diagnostic value, with the exception of suppression of EPO in most cases of PV (Messinezy *et al*, 1995). Similarly, the cell surface receptors with which these growth factors interact were not found to harbour recurrent abnormalities in the majority of patients with acquired MPNs.

In the 1980s and 1990s, this failure to discover a common, specific cytogenetic or molecular marker in the Philadelphia-negative MPNs prompted intense interest in the use of clonal markers as an alternative means to identify the haemopoietic population involved in the disease process (see section 1.3.5). Hypothesising that the disorders were initiated by a single, transforming event in a myeloid precursor which then proliferated to generate a population of

phylogenetically related progeny, it was argued that an affected patient should have evidence of clonal myelopoiesis, and it was hoped that this would be of diagnostic use. It soon became apparent, however, that a proportion of patients with bona fide MPNs had polyclonal or oligoclonal myelopoiesis.

The molecular pathogenesis of the Philadelphia-negative MPNs therefore remained obscure for at least forty years after the discovery of the Philadelphia chromosome. Clinically, the absence of a common cytogenetic or molecular marker hindered both diagnostic and therapeutic advances. Distinguishing individuals with ET and PV from those with reactive conditions was frequently a diagnostic challenge, and without a molecular lesion, designing targeted drugs (an approach which proved so successful in CML) was impossible.

For these reasons, the identification in 2005 of a specific point mutation affecting exon 14 of the *JAK2* gene (*JAK2* V617F) in a large proportion of patients with Philadelphia-negative MPNs was greeted by great excitement in the field (section 1.3.7). This finding, it was hoped, would finally explain the pathogenic process which gave rise to these disorders and revolutionise their management, as the discovery of the Philadelphia chromosome had done in CML.

1.2 Physiological megakaryopoiesis and platelet production

1.2.1 MK development and platelet biogenesis

Platelets play a key role in initiating haemostasis following vascular injury, and may also be involved in wound healing, inflammation and the innate immune response. They are small (3.0μ m by 0.5μ m), discoid anucleate cells, rich in cytoskeletal and signalling proteins, which are produced by budding from their precursor cell, the MK (Hoffbrand *et al*, 2005).

MKs are large (50-100µm diameter) cells, with a polyploid, multilobated nucleus. They arise in the bone marrow from MK progenitor cells (burst-forming unit-MK, BFU-MK, colony-forming unit-MK, CFU-MK, and high-proliferative potential-colony-forming unit-MK). These progenitors are themselves derived from a common MK-erythroid progenitor (MEP) cell, which in turn originates from the haemopoietic stem cell (HSC), via a common myeloid precursor (CMP) (Adolfsson *et al*, 2005; Debili *et al*, 1996; Figure 1.1).



Figure 1.1 Haemopoietic cell differentiation pathway

Schematic representation of the principal stages in MK maturation, with some stages in the maturation pathway of other haemopoietic cells also shown.

Chapter 1

During maturation, MKs undergo a number of specific changes which allow effective platelet production. They greatly amplify their DNA content, up to 128N, by a process of endomitosis, which results in cytoplasmic expansion and ribosomal proliferation, in preparation for the rapid production of platelet-specific proteins (Patel *et al*, 2005). MK maturation is also accompanied by the development of extensive internal membrane systems, continuous with the plasma membrane (Behnke 1968; Nakao and Angrist 1968). These internal membrane systems evaginate to form MK pseudopodial extensions called proplatelets, which then fragment to produce platelets (Behnke 1969; Radley and Haller 1982). Before the proplatelets form, large amounts of platelet-specific proteins are synthesised, and either transported to the MK surface or loaded into granules derived from Golgi complexes (Heijnen *et al*, 1998). At the same time, a dense tubular network and a system of channels called the open canalicular network is created to facilitate granule release (Gerrard *et al*, 1976).

Under steady-state conditions MKs represent approximately one in 2000 nucleated bone marrow cells; each one may release 1000-5000 platelets and in total around 2x10¹¹ platelets are formed every day (Kaushansky 2005; Trowbridge *et al*, 1984). However, in times of increased platelet consumption (e.g. immune thrombocytopenia), the MK mass may increase 3-fold, with a corresponding increase in platelet production (Branehög *et al*, 1975). The regulation of MK proliferation and platelet production is controlled by a complex system of humoral factors, cell surface receptors, intracellular signalling pathways, and nuclear transcription factors.

1.2.2 Hormonal regulation of megakaryopoiesis and thrombopoiesis

Interleukin-3 (IL-3), granulocyte-macrophage colony-stimulating factor (GM-CSF) and G-CSF are all important in the early stages of megakaryopoiesis, but the primary humoral regulator of

Chapter 1

megakaryopoiesis and thrombopoiesis is undoubtedly TPO, formerly known as the c-Mpl ligand. TPO is critical for all stages of megakaryopoiesis and thrombopoiesis, from the survival of HSCs to platelet formation, and even affects the haemostatic activity of mature platelets (Kojima *et al*, 2001). TPO infusions markedly increase MK ploidy and number, and stimulate platelet production (Kaushansky *et al*, 1994) whereas TPO-knockout mice have dramatically reduced numbers of MK progenitors and are severely thrombocytopenic (de Sauvage *et al*, 1996). MKs cultured *in vitro* in the presence of TPO extend numerous proplatelets, which structurally resemble those seen *in vivo*, and generate platelets which are functionally and structurally similar to normal platelets (Choi *et al*, 1995; Cramer *et al*, 1997).

TPO is encoded by a gene on chromosome 3q26.3-3q27, and is initially synthesised as a 353 amino acid propeptide, which is subsequently modified to a heavily glycosylated 332 amino acid functional protein. Under steady-state conditions, TPO is constitutively produced by the liver and kidneys and secreted into the bloodstream (Lok *et al*, 1994). Following binding to its receptor, c-Mpl on the surface of platelets and MKs, TPO is internalised and degraded (Fielder *et al*, 1996). Consequently, its levels generally vary inversely with platelet and MK mass (Emmons *et al*, 1996; Kuter and Rosenberg 1995). Of note, although endothelial cells also express c-Mpl and have a 100-fold greater surface area than platelets and MKs combined, they seem not to play a significant role in the regulation of serum TPO levels (Cardier and Dempsey 1998; Geddis *et al*, 2006).

In certain conditions, however, TPO levels are not inversely proportional to the platelet count. In immune thrombocytopenia, characterised by immune-mediated platelet and MK destruction, TPO levels may be near-normal (Houwerzijl *et al*, 2005; Kosugi *et al*, 1996), possibly due to its uptake by the increased MK mass and the sequestered platelets. In other circumstances, serum TPO concentrations may be influenced by factors outside this autoregulatory loop: reactive thrombocytosis, commonly seen in inflammatory states, is often accompanied by a rise rather than a fall in TPO levels, due primarily to IL-6 (an acute phase

protein, secreted by macrophages, fibroblasts and endothelial cells in response to bacterial infection and tissue damage) upregulating hepatic TPO synthesis (Hollen *et al*, 1991; Kaser *et al*, 2001; Kishimoto 1989). TPO levels may also be inappropriately normal or elevated in ET (discussed in section 1.3.3). Finally, during severe thrombocytopenia, additional TPO may be expressed by bone marrow stromal cells, though the mechanism for this is poorly understood (Sungaran *et al*, 1997).

1.2.3 Molecular signalling pathways involved in thrombopoiesis

TPO binds to c-Mpl on MKs, other haemopoietic precursors, and platelets, inducing conformational changes in the receptor which initiate signal transduction. This TPO/c-Mpl interaction plays a critical role in the regulation of both megakaryocytic and non-megakaryocytic haemopoietic progenitors: c-Mpl knockout mice have a marked reduction in MKs and are severely thrombocytopenic, but are also severely deficient in progenitors committed to all haemopoietic lineages (Alexander *et al*, 1996). Like other type 1 haemopoietic growth factor receptors, such as the erythropoietin receptor (EPOR) and the granulocyte-colony stimulating factor receptor (G-CSFR), c-Mpl is a single chain molecule without intrinsic tyrosine kinase activity but instead binds and relies on the activity of a non-receptor tyrosine kinase, Janus kinase 2 (JAK2).

In its unbound state, c-Mpl is expressed as a monomer on the cell surface (Horan *et al*, 1996; Alexander *et al*, 1996). Ligand binding induces dimerisation of two adjacent receptors, resulting in mutual cross-activation and activation of the bound JAK2 molecules (Kaushansky 2005). The activated JAK kinases initiate downstream signalling, by phosphorylating tyrosine residues on the receptor itself and on intracellular intermediaries, in particular the signal transducers and activators of transcription (STATs). STATs are critical for the survival and

expansion of haemopoietic progenitors (Kirito *et al*, 2002a; Snow *et al*, 2002), both by inducing transcription of anti-apoptotic genes such Bcl-xL (Kirito *et al*, 2002b), and by interacting with the GATA family of transcription factors (TFs) (Ezoe *et al*, 2005).

In addition, JAK kinases can activate Ras and, by phosphorylating phosphoinositide 3-kinase (PI3K), generate phosphoinositol_{3,4,5} triphosphate (PIP₃). Both PIP₃ and activated Ras inhibit the pro-apoptotic TF forkhead box O (FOXO) 3a, via activation of its negative regulator Akt, and thus promote MK survival (Geddis *et al*, 2001; Miyakawa *et al*, 2001; Nakao *et al*, 2008). Phosphorylated Ras can also activate members of the mitogen-activated protein kinase (MAPK) family, in particular the extracellular signal-regulated kinases (ERK) 1 and 2 which induce MK maturation via activation of the Runt-related TF 1 (RUNX1) (Hamelin *et al*, 2006; Rojnuckarin *et al*, 1999).

The resulting signalling cascade results in MK proliferation, growth and maturation, with increased ploidy, and platelet maturation and release. Modulation of this cascade is provided by simultaneous phosphorylation by JAK kinases of the suppressors of cytoplasmic signalling (SOCS) proteins, and SHP1 and SHIP1 phosphatases, which block cell signalling, primarily by inhibiting the phosphorylating activity of the JAK kinases (Kaushansky 2009; Figure 1.2).

1.2.4 Transcriptional control of megakaryopoiesis and thrombopoiesis

A number of TFs control MK development and thrombopoiesis, of which the best characterised are GATA-1, RUNX1, hypoxia-inducible factor (HIF) and nuclear factor erythroid 2 (NF-E2). GATA-1 is a zinc-finger protein which plays a critical role in all stages of megakaryopoiesis, and MK progenitors from GATA-1-deficient mice show maturation arrest at the promegakaryocyte stage (Shivdasani *et al*, 1997; Vyas *et al*, 1999). The role of GATA-1 in MK development is mediated in part by its interaction with another TF, friend of GATA-1 (FOG-1) and GATA-1 mutations which disrupt this association result in thrombocytopenia due to MK maturation arrest, and anaemia (Muntean and Crispino 2005). GATA-1 also binds the TF RUNX1, resulting in activation of MK promoters such as the αllb integrin promoter (Elagib *et al*, 2003), and upregulates expression of key genes involved in megakaryopoiesis and platelet assembly, including those encoding the p45 subunit of NF-E2, c-Mpl, platelet factor 4 and a number of platelet glycoproteins (Tsang *et al*, 1997; Vyas *et al*, 1999).



Figure 1.2 Principal JAK2-mediated MK signalling pathways

Following TPO binding, c-Mpl receptor dimerisation occurs, leading to JAK2 cross-activation and activation of downstream signalling proteins, including STAT, RAS and PI3K. This results in MK proliferation, growth and differentiation via induction of transcription factors such as the GATA family, Bcl-xL and RUNX1 and inhibition of the pro-apoptotic transcription factor FOXO 3a.

Like GATA-1, the TF HIF is important both in early and late megakaryopoiesis. It consists of two subunits: HIF-1 α , whose expression is regulated by TPO, and HIF-1 β , which is stably expressed. By regulating expression of vascular endothelial growth factor (VEGF), which in turn supports the survival of HSCs, HIF-1 α is important in the very early stages of megakaryopoiesis (Kirito *et al*, 2005). Further, HIF-1 α controls production of stromal cell-derived factor-1 (SDF-1), a trophic factor which induces proliferation of MK progenitors, polyploidisation of MKs, cell migration, and platelet production, and is therefore also involved in the later stages of megakaryopoiesis (Ceradini *et al*, 2004; Lane *et al*, 2000).

The heterodimeric leucine-zipper protein NF-E2 is another TF which plays a crucial role in terminal MK differentiation and platelet release, probably by regulating expression of several genes which encode MK proteins required for thrombopoiesis, including thromboxane synthase, β1-tubulin and Rab27b (Schulze and Shivdasani 2005). Loss of NF-E2 function does not markedly affect early MK differentiation, but results in disorganised internal membranes, severely impaired proplatelet formation and platelet release, and profound thrombocytopenia (Lecine *et al*, 1998).

1.3 ET – pathology and clues to pathogenesis

1.3.1 Cytological and histological changes

Although blood from ET patients usually looks macroscopically normal, using even low-power microscopy the thrombocytosis becomes obvious (Figure 1.3). Platelet anisocytosis may be evident, ranging from tiny forms to giant platelets, but frank morphological abnormalities are rare (Hehlmann *et al*, 1988). Histologically, the bone marrow cellularity varies from normal to

moderately hypercellular for age with a marked proliferation of large and giant MKs, which are characteristically arranged in loose clusters. They typically have deeply lobulated and hyperlobated nuclei, and abundant mature cytoplasm. Expansion of the myeloid and erythroid compartments and reticulin fibrosis are not prominent in ET (Thiele *et al*, 1999).

Ultrastructurally, ET platelets show defective pseudopod development (finger-like projections which appear following platelet activation at the site of vessel injury), reduced concentrations of alpha granules and abnormal tubular systems (Barnhart *et al*, 1980; Tablin *et al*, 1989). Such morphological changes might explain the impairment in platelet aggregation frequently seen in ET patients, who characteristically exhibit a reduced response to adrenaline stimulation (Avram *et al*, 2001; Hehlmann *et al*, 1988; Tsantes *et al*, 2010). Electron microscopy of ET MKs reveals a loss of the cytoplasmic organisation seen in normal MKs. In particular, there is a marked disturbance of the membrane systems, which are greatly increased in amount, disordered, vesiculated, and do not form the clearly-delineated platelet fields required to form normal proplatelets (Tablin *et al*, 1989).



Figure 1.3 Peripheral blood smear from a patient with ET and from a haematologically normal control

A. Patient with ET. B. Haematologically normal control

Arrows indicate platelets

Chapter 1

1.3.2 Cytogenetics

Cytogenetic abnormalities are uncommon in ET, present in up to 7% of cases, and none are specific to the disease (Bacher *et al*, 2005; Sessarego *et al*, 1989). The most frequent abnormalities include trisomy 8 and 9, deletions in 13q and 20q (Case Jr 1984; Gangat *et al*, 2009), and abnormalities of chromosome 1 (Gangat *et al*, 2009). Recent single-nucleotide polymorphism (SNP) microarray analysis has also failed to demonstrate the presence of recurrent genomic deletions or amplifications (Kawamata *et al*, 2008; Stegelmann *et al*, 2010), although occasionally acquired loss of heterozygosity affecting the short arm of chromosome 9 (9pLOH) has been identified in ET patients (Kralovics *et al*, 2003; Kralovics *et al*, 2005). The presence of cytogenetic abnormalities appears not to affect prognosis in ET (Gangat *et al*, 2009) and their role in the pathogenesis of the disorder is uncertain.

1.3.3 TPO levels and c-Mpl expression

A number of studies have demonstrated that, in ET, serum TPO levels are similar or elevated compared to normal controls (Table 1.2), which contrasts with the situation in PV, where levels of the principal regulator of erythropoiesis, EPO, are usually suppressed (Johansson *et al*, 2002; Mossuz *et al*, 2004). The reason why TPO is not suppressed in ET has not been convincingly elucidated. In *reactive* thrombocytosis, the increased hepatic TPO synthesis is due to elevated IL-6 levels (Hollen *et al*, 1991; Kaser *et al*, 2001). In ET, however, serum IL-6 levels are either normal (Hollen *et al*, 1991; Panteli *et al*, 2005) or suppressed (Griesshammer *et al*, 1998; Tefferi *et al*, 1994).

One hypothesis is that the c-Mpl-mediated internalisation/degradation mechanism which underpins normal TPO autoregulation is defective in ET. Horikawa *et al* (1997) found that surface c-Mpl expression on platelets from all 17 ET patients studied was dramatically lower

Studies examining TPO levels in ET Table 1.2

		Normal controls		ET patients	
Reference	c	TPO levels mean±SD /pg.mL ⁻¹ *	c	TPO levels mean±SD /pg.mL ⁻¹ *	p-value
(Tahara <i>et al,</i> 1996)	50	0.75±0.29	6	2.8±1.55	<0.05
(Cerutti <i>et al</i> , 1997)	32	156.7+ (62.2-352.7)	32 [‡]	246.2† (93.5-4596)	0.0006
(Horikawa <i>et al</i> , 1997)	21	0.76±0.21	17	1.31 ± 1.64	NS
(Hirayama <i>et al</i> , 1998)	9	22.1 ± 8.2	2	21.1 ± 1.5	SN
(Wang <i>et al</i> , 1998)	17	201 ± 112	20	505 ± 459	<0.05
(Griesshammer <i>et al</i> , 1998)	117	95.3 ± 54.0	25	545 ± 853	<0.001
(Harrison <i>et al</i> , 1999b)	12	122 ± 69	18	162 ± 138	SN
(Li <i>et al,</i> 2000)	18	179 ± 112	23	189 ± 131	NS

* Italicised values indicate units are fmol/L

⁺ Median (range)
[±] 12 of these patients had PV
than on normal control samples, with correspondingly lower c-MpI mRNA expression. Subsequent studies have largely confirmed these findings, but with some variation: Li *et al* (2000) observed a 10-fold reduction in c-MpI surface expression in ET platelets; Harrison *et al* (1999b) reported a less marked (but still significant) difference, with overlap between ET, other MPNs, and reactive thrombocytosis; Moliterno *et al* (1998), however, found no difference between the ET patients and normal controls. A similar downregulation in surface c-MpI expression has also been described in ET MKs (Mesa *et al*, 2002). Of note, in the 23 ET patients reported by Li *et al* (2000) whose TPO levels were similar to normal controls, the reduction in platelet c-MpI expression was accompanied by a four-fold decrease in platelet-mediated TPO clearance. Such a reduction in TPO clearance might explain its elevated levels in some patients with ET, and account for their thrombocytosis. In a murine model, suppression of c-MpI expression resulted in a dramatic thrombocytosis (Tiedt *et al*, 2009).

Another possible explanation for the elevated TPO levels seen in ET is that transcriptional or translational regulation of TPO synthesis is defective. Mutations in the 5'-untranslated region (5'-UTR) or a splice donor site in intron 3 of the *TPO* gene have been identified in four families with familial thrombocythaemia (FT; Table 1.3A). In haematologically normal individuals, a number of upstream open reading frames (uORFs) located within the 5'-UTR inhibit translation of *TPO* mRNA (Ghilardi *et al*, 1998); the mutations observed in FT reduce this uORF-mediated inhibition, resulting in excessive TPO protein synthesis and thrombocytosis. Equivalent abnormalities in patients with sporadic ET have not, however, been identified (Allen *et al*, 2001; Harrison *et al*, 1998).

 Table 1.3
 Mutations reported in familial thrombocythaemia

Reference	Mutation	Location in <i>TPO</i> gene	Inheritance*	Number of cases	Number of kindreds	Serum TPO in affected cases (pg/mL)
(Wiestner <i>et al,</i> 1998)	IVS3+1G>T	Intron 3 splice donor site	AD	11	1	80-1180
(Jorgensen <i>et al,</i> 1998)	IVS3+5A>G	Intron 3 splice donor site	AD	4	Not stated	430-980
(Jorgensen <i>et al,</i> 1998)	3252delG	5′-UTR	AD	5	1	422-1820
(Ghilardi <i>et al,</i> 1999)	516G>T	5'-UTR	AD	4	1	420-570

A. Familial thrombocythaemia due to mutations in the TPO gene

B. Familial thrombocythaemia due to mutations in the MPL gene

Reference	Mutation	Exon Domain†	Inheritance*	Number of cases	Notes
(Moliterno <i>et al,</i> 2004)	K39N (1238G>T)	2 ECD	See note	3 homozygotes 12 heterozygotes	Mild thrombocytosis in heterozygotes, severe in homozygotes. Heterozygote frequency in African-American populations was 7%. Not detected in 250 Caucasian, 40 Hispanic or 39 Asian controls
(Ding <i>et al,</i> 2004)	S505N (1073G>A)	10 TMD	AD	8 heterozygotes	All 8 cases in a single kindred
(El-Harith <i>et al,</i> 2009)	P106L (317C>T)	3 ECD	See note	8 homozygotes 18 heterozygotes	Affected cases all homozygotes of Arabic ancestry (4 families). Of the 18 heterozygotes, 4 had a mild thrombocytosis. Heterozygote frequency amongst an Arabic population studied was 7%
(Teofili <i>et al,</i> 2010)	S505N (1073G>A)	10 TMD	AD	24 heterozygotes	24 cases spread across 8 kindreds. Similar phenotype to sporadic ET (thrombotic and fibrotic risk)
+ ECD indicates extract	i Unicacio: TNAD i	indicator transform	aicmole demois		

ECD indicates extracellular domain; TMD indicates transmembrane domain

* AD indicates autosomal dominant

1.3.4 Growth factor hypersensitivity in ET progenitors and possible causes

Another hypothesis put forward to explain excessive platelet production in ET, especially in patients whose TPO levels are not elevated, is that the progenitors are hypersensitive to TPO, or even proliferate in its absence. The existence of EECs in PV is well described (Prchal and Axelrad 1974; Zanjani *et al*, 1977) and represents a minor diagnostic criterion in the latest WHO classification (Swerdlow *et al*, 2008; Table 1.1). Indeed, the presence of EECs has been described in some ET patients (Juvonen *et al*, 1987; Partanen *et al*, 1983), though its significance is unclear. It is possible that it may be an artefact caused by growth factor secretion by contaminating T-cells, or that the majority of patients with a thrombocytosis and EECs actually have PV masked by iron deficiency (Shih and Lee 1994).

A number of studies have demonstrated the presence of endogenous MK colonies (EMCs) in ET (Battegay *et al*, 1989; Juvonen *et al*, 1987), though the rate varies between 63% (Florensa *et al*, 1995) and 100% of cases (Mi *et al*, 2001). Although EMCs have not been reported in reactive thrombocytosis (Dobo *et al*, 2004; Rolovic *et al*, 1995), they have been detected in between 33% and 95% of PV patients (Escoffre-Barbe *et al*, 2006; Mi *et al*, 2001), indicating that they are not specific to ET. Moreover, it has been suggested that components of the 'serum-free media' used to screen for EMCs (e.g. albumin) may in fact have been contaminated by serum, and that the 'spontaneous' megakaryocytic growth actually represented progenitor hypersensitivity to TPO (Axelrad *et al*, 2000). Indeed, Axelrad *et al* (2000) did demonstrate that megakaryocytic progenitors in ET were on average 53-times more sensitive to TPO than were those from normal controls, findings which have been confirmed by subsequent studies (Kawasaki *et al*, 2001; Mi *et al*, 2001). Axelrad *et al* (2000) went on to speculate that progenitor hypersensitivity to regulatory cytokines might exist in every MPN, and that the lineage of the hypersensitive progenitor determined the phenotype.

The observation that transforming factor beta 1 (TGF- β 1), a platelet and MK-derived cytokine, was involved in the negative regulation of megakaryopoiesis in mice (Ishibashi *et al*, 1987) prompted speculation that defects in this pathway might be responsible for the presence of EMCs in ET. In keeping with this, Zauli *et al* (1993) demonstrated that megakaryocyte progenitors in ET were less sensitive to TGF- β 1 than those from healthy controls. It is possible that this effect is mediated by the TF NFI-B: increased NFI-B has been associated with 9pLOH in some MPN patients (the gene encoding NFI-B is located on chromosome 9p), and in a murine model overexpression of this gene resulted in TGF- β insensitivity (Kralovics *et al*, 2002). A subsequent larger study by the same group, however, refuted the association between 9pLOH and increased NFI-B expression and attributed their earlier findings to chance (Kralovics *et al*, 2003).

Another hypothesis to explain TPO hypersensitivity in ET was that that myeloid progenitors in this disease secreted TPO, generating an autocrine feedback loop. This, however, was refuted by the finding that CD34+ cells only express *TPO* mRNA at very low levels, and that addition of a neutralising antibody to TPO did not affect EMC activity (Taksin *et al*, 1999). The same group and others also investigated the possibility that the activating mutations in c-Mpl observed in FT (Table 1.3B), might also account for the thrombocytosis and autonomous MK proliferation seen in sporadic ET. Two groups therefore sequenced the entire coding region of the gene in a total of 19 patients with non-familial ET, but no mutations were detected (Kiladjian *et al*, 1997; Taksin *et al*, 1999). Subsequent technical advances have allowed larger number of patients to be screened and mutations in the gene encoding c-Mpl have since been identified in approximately 5% of ET patients (see section 1.3.8).

1.3.5 Clonality in ET

Dameshek's original hypothesis for the aetiology of the MPNs suggested that myeloid proliferation occurred in response to a myelostimulatory factor (Dameshek 1951), but he did not speculate on whether the proliferating myeloid cells were derived from a single haemopoietic progenitor or many – i.e. whether it was a monoclonal or polyclonal process. This distinction is of fundamental importance to our understanding of the diseases, since monoclonality is a characteristic feature of neoplastic disorders. Although the Philadelphia chromosome was subsequently detected in most patients with CML (Nowell and Hungerford 1960), in keeping with it being a monoclonal disease, no equivalent abnormality was apparent in the majority of PV and ET patients.

In order to address the issue of clonality in the MPNs, Fialkow *et al* (1967) pioneered the use of X-chromosome inactivation pattern (XCIP) analysis. This technique exploits the cellular mosaicism present in most females (with regard to X-chromosome inactivation) to determine whether a cell population has arisen from a single progenitor without requiring the presence of a specific disease marker. Seminal works by Lyon *et al* (1961) and Beutler *et al* (1962) had shown that in females one X-chromosome is randomly inactivated in every cell during embryogenesis in order to maintain comparable X-linked gene dosage with males. Mohendas *et al* (1981) subsequently demonstrated that the mechanism for this X-chromosome inactivation was methylation. By a process of imprinting, all progeny of these cells inactivate the same X-chromosome as their original progenitor.

In normal polyclonal haemopoiesis one would, therefore, expect some blood cells to inactivate the maternal X-chromosome, and others to inactivate the paternal X-chromosome, since the probability of every stem cell in the original pool inactivating the same X-chromosome by chance is small. By contrast, if a particular population of cells is clonal and has arisen from a single transformed precursor, every cell in that population will inactivate the same X-

chromosome, and the XCIP for the whole population will be 'skewed' if the clonal population is large enough.

In order to determine the XCIP of a tissue or population of cells it is necessary to distinguish between the maternal and the paternal X-chromosome, and between the active and the inactive X-chromosome. The first criterion requires that the individual is heterozygous for an Xlinked gene or allelic marker, such as a tandem repeat sequence. The second involves either measuring the relative expression (at the mRNA or protein level) of each X-linked allele for which the individual is heterozygous, or analysing their DNA methylation patterns.

Using this approach, Fialkow *et al* (1967) investigated the clonality of granulocytes from three CML female patients who were heterozygous for isoenzymes of the X-linked gene glucose-6phosphate dehydrogenase (*G6PD*). In all three patients, only a single G6PD isoenzyme was detectable in their granulocytes whereas both isoenzymes were detected in their fibroblasts, and the authors concluded that this implied that CML was a clonal disease. Whether the same was true for ET was not investigated until 1981, when the same group studied three ET female patients who were also heterozygous for isoenzymes of G6PD. In two patients, only a single G6PD isoenzyme was detectable in all erythrocytes, granulocytes and platelets, whilst in the third patient there was evidence of 95% skewing towards one isoenzyme which had become 100% four months later, a finding which was interpreted as evidence for myeloid clonality in ET. Several other groups subsequently confirmed these observations (Gaetani *et al*, 1982; Singal *et al*, 1983), though patient numbers in these early protein-based studies were limited by the relatively low frequency of G6PD isoenzyme heterozygosity in individuals of non-African ancestry.

Evaluation of much larger cohorts, however, was made feasible by the development of DNA methylation-based approaches to XCIP analysis, allowing approximately 50% of females to be analysed using polymorphic sites within the X-linked genes phosphoglycerate kinase and

hypoxanthine-guanine phosphoribosyltransferase (Vogelstein *et al,* 1985; Vogelstein *et al,* 1987). The proportion of evaluable females was further increased by the analysis of other Xlinked loci, such as the highly polymorphic variable number tandem repeat (VNTR) sequences in the DXS255 locus and the human androgen receptor (*HUMAR*) gene, for which more than 90% of females are polymorphic (Allen *et al,* 1992; Boyd and Fraser 1990; Gale *et al,* 1992; Lucas *et al,* 1989; Tsukamoto *et al,* 1994). In addition, PCR rather than Southern blotting-based analysis allowed assessment of smaller amounts of tissue (Gilliland *et al,* 1991).

An alternative approach to the XCIP analysis involves assaying the transcribed mRNA products of an X-linked gene. Early studies exploited a polymorphism in the *G6PD* gene, different from that used in protein-based clonality studies and more frequently informative, to distinguish between the maternal and paternal X-chromosomes (Curnutte *et al*, 1992; Prchal and Guan 1993). Subsequently a number of genes, including iduronate-2-sulfatase, *MPP1* and *FHL-1*, have proved useful in transcription-based clonality analysis, allowing over 95% of females to be evaluated (Gregg *et al*, 2000; Liu *et al*, 2003; Luhovy *et al*, 1995).

Use of XCIP-based clonality analysis in ET expanded rapidly in the 1990s as a result of these technical advances. Initial studies were often interpreted as confirming the results of the earlier protein-based assays – that ET was a clonal disorder – even though some ET patients appeared to have non-clonal myelopoiesis (Anger *et al*, 1990; Kassar *et al*, 1995; Lucas *et al*, 1989; Tsukamoto *et al*, 1994). With increasing use of the technique, however, a number of limitations and caveats became apparent. First, it cannot be used to investigate males, or indeed females who are not heterozygous for an X-linked marker gene. Since about 70% of ET patients are female (Jensen *et al*, 2000), and approximately 90-95% of these are informative for an X-linked marker, at most about two-thirds of ET patients are evaluable using XCIP analysis.

Second, the XCIP only describes the overall clonality pattern for the entire cell population analysed; even if the XCIP of the clonal cells is grossly imbalanced, the presence of non-clonal cells (where the XCIP would be expected to be more balanced) will diminish the degree of skewing of the overall population. To accommodate this, the minimum X-allele inactivation ratio which most studies use to define clonality is 75%:25% – i.e. the degree of inactivation of one allele is at least 3-fold greater than the other (Vogelstein *et al*, 1987). Even so, a small clone within a large polyclonal population may be missed (Levine *et al*, 2006).

Third, since the XCIP is established early in embryogenesis in a small pool of haemopoietic stem cells, extreme skewing may occur by chance (Gale *et al*, 1991; Gale *et al*, 1993). It has been estimated that degree of X-chromosome inactivation commonly used to define clonality (75%:25%) would arise purely by chance in 18% of individuals if XCI occurred in a haemopoietic stem cell pool consisting of eight cells (Gale and Linch 1994). An apparently clonal pattern in myeloid cells may, therefore, reflect an imbalanced constitutional pattern, and would not necessarily indicate the presence of an abnormal population – a phenomenon termed constitutional skewing. T-cells arise from the same haemopoietic stem cell as myeloid cells and are assumed to have the same constitutive XCIP. They are not believed to be involved with the abnormal clone in ET. Using the patient's T-cell XCIP as a control, therefore, allows individuals with constitutional skewing to be identified.

Finally, elderly women may appear to have a clonal XCIP even in the absence of a haematological disorder, a consequence of 'age-related' skewing. Busque *et al* (1996) observed that 38% of haematologically normal elderly females (\geq 60 years old) had an apparently clonal XCIP, compared to 16% of younger adults, and Gale *et al* (1997) reported equivalent figures of 56% in elderly female controls (\geq 75 years old) compared to 22% in younger women. T-cell clonality does not correlate well with myeloid clonality in elderly women and a balanced T-cell XCIP does not, therefore, reliably exclude the presence of this phenomenon (Gale *et al*, 1997). Consequently, an imbalanced myeloid XCIP in an elderly

woman does not necessarily imply the presence of a clonal myeloid population and such a result cannot be meaningfully interpreted – a significant limitation in a disease which predominantly affects older people. The finding of a balanced XCIP, however, *does* indicate the presence of polyclonal myelopoiesis, regardless of age.

Taking all these caveats into account, in order to fulfil the criteria of clonal myelopoiesis the patient must be younger than 65 years and have a myeloid XCIP which shows more than 75% inactivation of one allele, which is also at least 20% different from the control (T-cell) XCIP.

As these refinements to the interpretation of XCIP analysis became better understood in the late 1990's, a number of studies were published which suggested that a proportion (between 31% and 57%) of ET patients did *not* have evidence of clonal haemopoiesis (Table 1.4). This finding was of great significance since it contradicted the notion that ET was invariably a clonal, neoplastic disorder. The suggestion that ET was a heterogeneous disease was supported by the observation that the rate of vascular events was influenced by clonality status: of the four studies which examined the clinical associations of myeloid clonality, two found that patients with clonal haemopoiesis appeared more likely to suffer a thrombotic complication during their illness than those without (Chiusolo *et al*, 2001; Harrison *et al*, 1999a), whilst the other two found no significant difference (El-Kassar *et al*, 1997; Shih *et al*, 2002).

Ē
⊒.
XCIP
of
Studies
1.4
Ð
Tab

Table includes only those studies which allow the criteria described in Section 1.3.5 to be applied

						Myeloid	XCIP		
Reference	Techninue	Mean age at diagnosis	Definition	Total patients	Polyclonal		Clonal	Interpretable patients	Comments
		in years (range)	of clonality	2	Total <i>n</i>	Total <i>n</i>	≤65 years and polyclonal T-cells <i>n</i>	(of which clonal, %)	
(El-Kassar <i>et</i> al, 1997)	DNA (HUMARA) in T- cells, granulocytes and whole blood RNA (IDS, p55, G6PD) in T-cells, granulocytes, platelets and whole blood	62 (13-77)	Allele ratio <0.20	46	14	32	19	33 (58%)	Two further patients fulfilled the criteria for myeloid clonality if platelets rather than neutrophils were analysed Clonal XCIP associated with increased platelet count at diagnosis and older age (P=0.04 and 0.01 respectively) but no effect on thrombosis
(Harrison <i>et</i> al, 1999a)	DNA (HUMARA & PGK) in T-cells and neutrophils. RNA (IDS & G6PD) in T- cells, neutrophils and platelets	53 (11-89)	>75% expression of 1 allele	46	13	e e	10	23 (43%)	Thrombotic rate higher in clonal than polyclonal patients (60% vs. 15%; P=0.04) No difference in age or platelet count at presentation, follow-up or bleeding risk between clonal and polyclonal groups. No evidence of megakaryocyte-restricted clonality
(Chiusolo <i>et</i> al, 2001)	DNA (HUMARA) in T- cells, granulocytes and	34 (20-63)	>75% expression of	40	15	25	17	32 (53%)	Thrombotic rate higher in clonal than polyclonal patients (41% vs. 7%; P=0.04). The thromboses in the clonal patients were mostly unusual

splanchnic events No difference in age or platelet count at presentation, or duration of follow-up. No evidence of megakaryocyte-restricted clonality	20 of 89 patients excluded because HUMARA homozygous or XCIP was ambiguous Thrombotic rate not significantly higher in clonal than polyclonal patients (33% vs. 7%; P=0.07)	Clinical correlates not reported	T-cell clonality not reported but methods imply that T-cell XCIPs were used to identify cases of constitutional skewing Clinical correlates not reported	
	48 (69%)	13 (69%)	12 (67%)	161 (60%)
	33	6	œ	96
	54	15	œ	167
	15	4	4	65
	68	19	12	252
1 allele	>75% expression of 1 allele	Not stated	Not stated	
	55 (18-92)	49 (24-83)	46 (23-63)	
CD34+ cells RNA (IDS & p55) in platelets	DNA (HUMARA) in T- cells and granulocytes	DNA (HUMARA) in T- cells and granulocytes	RNA (MPP1, IDS, G6PD, BTK and FHL1) in T-cells and granulocytes	
	(Shih <i>et al,</i> 2002)	(Teofili <i>et al,</i> 2002)	(Liu <i>et al,</i> 2003)	Overall

Table 1.4 Studies of XCIP in ET – continued

+ The interpretable number includes only patients with a polyclonal myeloid XCIP and patients with a clonal myeloid XCIP aged ≤65 years with polyclonal T-cell XCIP (i.e. no evidence of possible age-related skewing or constitutional skewing) where the reported data allows such patients to be identified.

1.3.6 PRV-1 expression

Whilst XCIP analysis undoubtedly advanced our understanding of the clonal origins of ET, the specific molecular defects which initiated the MPNs remained obscure. In an attempt to identify these abnormalities, Temerinac *et al* (2000) analysed differences in gene expression between MPN and normal cells. Using subtractive mRNA hybridisation in neutrophils from 19 PV and six ET patients, and 21 normal controls, they demonstrated overexpression of a novel gene which they called polycythemia vera-1 (*PRV-1*) in all PV patients and two ET patients. Since *PRV-1* mRNA was not detectable in any normal controls, it was hoped that *PRV-1* analysis would be useful diagnostically. *PRV-1* encodes a granulocyte cell surface receptor – the antigen CD-177 – the function of which is unknown. Surprisingly, despite overexpression of *PRV-1* mRNA, protein expression was not elevated in most PV patients (Klippel *et al*, 2002).

Although a subsequent study found *PRV-1* overexpression in all 37 ET patients investigated (Teofili *et al*, 2002), later studies found lower rates of *PRV-1* overexpression in ET patients – between 17% and 67% (Griesshammer *et al*, 2004; Kralovics *et al*, 2003; Liu *et al*, 2003; Tefferi *et al*, 2004). Taking all the reports together, 69 of 114 (61%) ET patients were *PRV-1* positive (Table 1.5), although the lack of a standard definition for *PRV-1* positivity or overexpression makes direct comparison between the seven studies difficult. Two studies found that *PRV-1* overexpression was strongly associated with EEC formation, usually a hallmark of PV (Griesshammer *et al*, 2004; Kralovics *et al*, 2003). Combined with the observation that 40% of *PRV-1* positive ET patients went on to develop PV, this association led to speculation that *PRV-1* positive ET patients actually had 'masked' PV (Griesshammer *et al*, 2004). Interest in *PRV-1*, however, was soon largely overshadowed by a more significant discovery.

Ш
2.
studies
PRV-1
1.5
Table

Reference	Technique	Definition of <i>PRV-1</i> positivity	Number of <i>PRV-1</i> positive ET patients	Comments
(Temerinac <i>et al,</i> 2000)	Northern blot	Presence of band on gel	2/6 (33%)	1 of the positive patients was borderline (faint band)
(Teofili <i>et al,</i> 2002)	Northern blot	Presence of band on gel	37/37 (100%)	All clonal patients expressed PRV-1
(Kralovics <i>et al,</i> 2003)	Real-time PCR	Increased <i>PRV-1</i> expression compared to RPL19*	10/15 (67%)	Noted strong association between <i>PRV-1</i> positivity and presence of EECs in individual patients
(Liu <i>et al,</i> 2003)	Real-time PCR	Increased <i>PRV-1</i> expression compared to GAPDH*	2/12 (17%)	PRV-1 expression overall similar to haematologically normal controls
(Griesshammer <i>et al,</i> 2004)	Northern blot	Presence of band on gel	15/30 (50%)	All 15 <i>PRV-1</i> positive patients had EECs and 40% of them developed PV <i>PRV-1</i> positive patients more likely to develop thrombotic and microcirculatory complications
(Tefferi <i>et al,</i> 2004)	Real-time PCR	Increased <i>PRV-1</i> expression compared to GAPDH ⁺	3/14 (21%)	
Overall			69/114 (61%)	
* PRV-1 expression relative t	o a housekeeping g	ene (RPL19 or GAPDH) was determin	ed initially for a cohort of	normal controls to define the normal range

1.3.7 JAK2 V617F

In 2005, five groups independently reported a recurrent G>T transversion in exon 14 of the gene encoding *JAK2*, resulting in a valine-to-phenylalanine substitution in residue 617 (*JAK2* V617F) in patients with MPNs (Table 1.6). Three different routes led the five groups to investigate *JAK2*. James *et al* (2005) had previously observed that impairing *JAK2* activity using a tyrosine kinase inhibitor reduced EEC formation (Ugo *et al*, 2004). Kralovics *et al* (2002; 2003) had reported that approximately 30% of PV patients and one of the 15 ET patients studied had 9pLOH – i.e. acquired uniparental disomy involving the short arm of chromosome 9 as a result of mitotic recombination. In their 2005 study, high-definition mapping identified a 6.3Mbp region on chromosome 9p common to all patients which included *JAK2* (Kralovics *et al*, 2005). The other three groups analysed *JAK2* as part of a systematic tyrosine kinase mutation screen (Baxter *et al*, 2005; Levine *et al*, 2005; Zhao *et al*, 2005).

At least 16 other studies have subsequently examined the frequency of *JAK2* V617F in ET, with rates ranging between 23% and 75%, and an overall mean of 53% (Table 1.7). Some of this variation is due to methodological differences, since direct sequencing is less sensitive than allele-specific polymerase chain reaction (AS-PCR) or PCR-restriction enzyme digestion approaches (Campbell *et al*, 2006b), and some may be due to different stringencies in applying the diagnostic criteria. In most studies which have looked for it, *JAK2* V617F has not been identified in normal controls (Table 1.6), though one study of nearly 4000 Chinese hospitalised patients not known to have an MPN did detect it in about 1% of individuals (Xu *et al*, 2007). This study used nested PCR to increase its sensitivity, however, and it is possible that this increased the risk of false-positivity due to contamination.

Initial studies reporting JAK2 V617F Table 1.6

Reference	Backeround	Technique for mutation		PV		ET		MF	S O	rmal ntrols	Notes
	0	detection	u	V617F+	u	V617F+	u	V617F+	u	V617F+	
(Baxter <i>et al,</i> 2005)	Tyrosine kinase mutation screening	Sequencing confirmed by ARMS PCR	73	71 (97%)	51	29* (57%)	8	16 (50%)	06	0	Mutation identified in multipotent progenitor cells, and present in all EECs in PV. No mutation in T-cells
(James <i>et al,</i> 2005)	Role of <i>JAK2</i> in PV identified by inhibition and knock- down studies	Sequencing	45	40 (89%)	21	9 (43%)	7	3 (43%)	15†	o	Transfection of V617F into cytokine- dependent cell lines resulted in factor independence. Transplant studies in mice resulted in elevated haematocrit
(Kralovics <i>et al,</i> 2005)	Previous identification of 9pLOH in PV and ET patients	Sequencing (neutrophil cDNA)	128	83 (65%)	93	21 (23%)	23	13 (57%)	71^{\pm}	o	Higher rate of fibrosis, haemorrhage and thrombosis in V617F-positive patients. Transfection of V617F into BaF3 cells led to cytokine hypersensitivity and IL-3 independence
(Levine <i>et al,</i> 2005)	Tyrosine kinase mutation screening	Sequencing	164	121 (74%)	115	37 (32%)	46	16 (35%)	269	0	V617F constitutively activated when expressed in 293T cells
(Zhao <i>et al,</i> 2005)	Tyrosine kinase and phosphatase mutation screening	Sequencing	24	20 (83%)	1	I	ı	I	12	0	Transfection of V617F into HeLa cells resulted in increased STAT, Akt and ERK phosphorylation
* The maiority c	$10^{10} + 10^{10} = 70\%$	were only V617F-	onsitive	hv ARMS ar	alvsis	not hv sedu	encina				

20 יטוז גוגעושווש כו

The majority of these (23/29 = 79%) were only vol/r-positive by AKIMS analysis, not the 35 secondary erythrocytosis patients were also tested: all were JAK2 V617F-negative

++

9 CML and 11 secondary erythrocytosis patients were also tested: all were JAK2 V617F-negative

Reference	n	Mutated (n)	Mutation frequency (%)	Method [*]
(Levine <i>et al,</i> 2005)	21	9	43	PCR sequencing
(Kralovics et al, 2005)	93	21	23	PCR sequencing
(Levine <i>et al,</i> 2005)	115	37	32	PCR sequencing
(Baxter <i>et al,</i> 2005)	51	29	57	AS-PCR
(Jones <i>et al,</i> 2005)	59	24	41	ARMS-PCR
(Jelinek <i>et al,</i> 2005)	10	3	30	Pyrosequencing
(Goerttler <i>et al,</i> 2005)	42	14	33	PCR sequencing (cDNA)
(Campbell <i>et al,</i> 2005)	776	411	53	AS-PCR and PCR-enzyme digestion
(Vizmanos et al, 2006)	243	151	62	AS-PCR
(Cheung <i>et al,</i> 2006)	60	29	48	AS-PCR
(Kiladjian <i>et al,</i> 2006)	44	19	43	Real-time AS-PCR
(Lippert <i>et al,</i> 2006)	60	45	75	AS- and q-real time-PCR
(Moliterno <i>et al,</i> 2006)	84	38	45	qPCR
(Gale <i>et al,</i> 2007)	111	43	39	PCR-enzyme digestion
(Larsen <i>et al,</i> 2007)	40	21	53	qPCR
(Kittur <i>et al,</i> 2007)	176	96	55	AS-qPCR
(Alvarez-Larran <i>et al,</i> 2007)	103	44	43	AS-PCR
(Vannucchi <i>et al,</i> 2007)	639	382	60	AS-PCR
(Lieu <i>et al,</i> 2008)	49	29	59	PCR-enzyme digestion
(Shen <i>et al,</i> 2009)	85	53	62	ARMS-PCR
(Palandri <i>et al,</i> 2009b)	275	176	64	PCR-enzyme digestion
(Hussein <i>et al,</i> 2009)	91	43	47	Pyrosequencing
(Pardanani <i>et al,</i> 2010)	226	118	52	AS-PCR
(Wong <i>et al,</i> 2010)	102	35	34	PCR-enzyme digestion
Overall	3555	1870	53	

Table 1.7 Frequency of JAK2 V617F in ET – published studies to date

* AS-PCR = allele-specific PCR

ARMS-PCR = amplification refractory mutation system PCR;

qPCR = quantitative PCR

In ET patients whose myeloid cells carry *JAK2* V617F, the mutation is not detectable in constitutional control tissue, such as buccal or hair follicle cells (Baxter *et al*, 2005; Kralovics *et al*, 2005; Levine *et al*, 2005), in keeping with it being a somatically acquired mutation. Further, the majority of studies suggest that such patients' T-lymphocytes are negative for the mutation, implying it is restricted to the myeloid lineage and that it arose in a myeloid progenitor cell rather than a haemopoietic stem cell (Baxter *et al*, 2005).

As described in Section 1.2.3, JAK2 is a non-receptor tyrosine kinase which binds to type 1 haemopoietic growth factor receptors and initiates downstream signalling when ligand binds to the receptor. JAK2 shares a common domain structure with other JAK family members (JAK1, JAK2 and TYK2): a C-terminal kinase domain (JAK homology 1, JH1), adjacent to which is a pseudokinase domain (JH2) which bears significant homology to JH1 but is enzymatically inactive and is predicted to negatively regulate the activity of JH1 in the absence of receptorbound ligand, an SH2 domain (JH3-4) of unknown function, and an N-terminal FERM domain (JH5-7), which mediates binding to the Box1 domain of the receptor's proximal intracytoplasmic region (Haan *et al*, 2001; Haan *et al*, 2002; Lindauer *et al*, 2001; Pellegrini and Dusanter-Fourt 1997; Richter *et al*, 1998). The V617F mutation is located in the JH2 domain of the molecule, and its acquisition is believed to release the JH1 domains from their JH2mediated inhibition. Indeed structural modelling published in 2001 had predicted that residue 617 lay within one of two small but critical interaction sites through which the JH2 domain mediated this inhibition (Lindauer *et al*, 2001); replacement of valine with bulky phenyalanine would be expected to disrupt its inhibitory effect.

In keeping with this, transfection of *JAK2* V617F into cytokine-dependent cell lines resulted in cytokine independence, with evidence of JAK2 autophosphorylation and constitutive activation of downstream pathways (James *et al*, 2005; Kralovics *et al*, 2005; Levine *et al*, 2005; Zhao *et*

al, 2005). Transplantation of *JAK2* V617F-transfected bone marrow cells into lethally irradiated C57BL6 mice recapitulated the polycythaemic phenotype, often with leukocytosis, thrombocytosis and some myelofibrotic features (Akada *et al,* 2010; James *et al,* 2005; Wernig *et al,* 2006).

The role of JAK2 V617F in ET will be discussed in greater detail in Chapters 3, 4, 5 and 6.

1.3.8 Other genetic mutations in ET

In FT, a number of mutations have been described in the genes encoding TPO and c-Mpl (Table 1.3). Attempts to find similar abnormalities in sporadic ET, however, proved fruitless for many years (Allen et al, 2001; Harrison et al, 1998; Kiladjian et al, 1997; Taksin et al, 1999), partly because technological constraints limited the number of patients who could be screened. In 2006, Pikman et al (2006) sequenced selected exons of the genes encoding c-Mpl, EPOR and G-CSFR in 125 JAK2 WT MPN patients (66 ET, 45 MF and 14 PV) and observed a MPL W515L mutation in four cases of MF. Subsequent studies have demonstrated that MPL W515L mutations do also occur in a small proportion of ET patients and other mutations have been reported (Table 1.8). Overall, MPL mutations are detectable in approximately 4% of ET patients, of whom 90% are JAK2 WT. For this reason, the MPL mutation frequency is markedly higher in JAK2 WT than in V617F-positive patients (6% versus 1%). The commonest mutations are W515L and W515K, which account for 70% and 27% respectively of the mutations so far described, with other mutations being S505N (3%) and W515A (1%). Mutations affecting K39 and P106, reported in FT, have not been described in ET, although only two large-scale studies (Beer et al, 2008; Pikman et al, 2006) have screened for mutations outside residues 505 and 515.

 Table 1.8
 MPL mutations reported in sporadic ET

		AII	ET patients	JAK2 P	! V617F ET atients	JAK2	WT ET patients	Mutatio	ons detecte	d (where st	ated)*
Reference	Technique for <i>MPL</i> mutation detection	2	<i>MPL</i> mutation (% of patients)	2	<i>MPL</i> mutation (% of patients)	2	<i>MPL</i> mutation (% of patients)	W515L	W515K	W515A	S505N
(Pardanani <i>et al,</i> 2006)	Light-Cycler analysis of W515	318	4 (1%)	182	2 (1%)	136	2 (1%)	4	0	0	μ
(Vannucchi <i>et al,</i> 2008a)	Real-time qPCR for <i>MPL</i> W515L/K	994	30 (3%)	576	8 (1%)	418	22 (5%)	18	12	ΤN	NT
(Xu <i>et al,</i> 2008)	AS-PCR for MPL W515L	102	1 (1%)	43	(%0) 0	59	1 (2%)	Ч	NT	NT	NT
(Beer <i>et al</i> , 2008)	Whole gene sequencing in 18 patients; exon 10 sequencing in 88 patients; AS-PCR for S205N and W515L/K in 670 patients	776	32 (4%)	414	1 (0%)	362	31 (9%)	24	'n	o	m
(Schnittger <i>et al,</i> 2009)	Light-Cycler analysis of W515	356	26 (7%)	32	0 (%0) 0	324	26 (8%)	16	б	Ц	μ
(Ruan <i>et al,</i> 2010)	Real-time qPCR for <i>MPL</i> W515L/K	199†	7 (4%)	0	0 (0%)	199	7 (3.5%)	9	7	ΝΤ	NT

Chapter 1

specific PCR for MPL 100 1 (1%) 49 0 (0%) 51 1 (2%) 1 0 NT 0 ISN and W515L/K 2953 107 (4%) 1296 11 (1%) 1657 96 (6%) 70 27 1 3	Light-Cycler ar	alysis of W515	108†	6 (6%)	0	0 (%0)	108	6 (6%)	N/A	N/A	TN	NT
2953 107 (4%) 1296 11 (1%) 1657 96 (6%) 70 27 1 3	specifi 5N anc	c PCR for <i>MPL</i> d W515L/K	100	1 (1%)	49	0 (0%)	51	1 (2%)	1	0	ΤN	0
			2953	107 (4%)	1296	11 (1%)	1657	96 (6%)	70	27	1	3

Table 1.8 MPL mutations reported in sporadic ET – continued

- * NT = not tested; N/A = not available
- These studies screened only JAK2 WT patients.

Residues 505 and 515 lie within the transmembrane domain of c-Mpl, and mutations are believed to disrupt a motif which normally prevents autonomous receptor activation, thereby allowing receptor dimerisation in the absence of ligand (Glembotsky *et al*, 2010). *MPL* W515Ltransduced cell lines show cytokine-independent growth, hyper-responsiveness to TPO and constitutive activation of the JAK/STAT signalling pathway (Pikman *et al*, 2006). In a murine bone marrow transplant assay, expression of *MPL* W515L resulted in thrombocytosis, leukocytosis, MK hyperplasia, and splenomegaly due to extramedullary haemopoiesis, a phenotype similar to pre-fibrotic MF (Pikman *et al*, 2006).

Mutations in other genes have been reported infrequently in ET patients. In 2009, Delhommeau *et al* (2009) reported on the presence of mutations in the ten-eleven translocation 2 (*TET2*) gene for in a variety of myeloid disorders. They initially studied six patients with AML who were known to have rearrangements involving chromosome 4q24. FISH analysis revealed a commonly deleted region of chromosome 4 in three of these patients and in a further three MDS patients analysed; this region contained only the *TET2* gene. They then screened five MPN patients for copy number alterations using comparative genomic hybridisation and SNP arrays and in three of them identified a 325kb common region of LOH on chromosome 4q; again this region contained only the *TET2* gene. Direct sequencing of *TET2* in eight of the patients with deletions or LOH at 4q24 revealed mutations in six of them: one frame shift, one in-frame deletion, two nonsense mutations, and two amino acid substitutions.

Delhommeau *et al* (2009) then screened 203 MPN patients (of whom 186 were *JAK2* V617Fpositive) for *TET2* abnormalities, and of these 27 (13%) were found to harbour *TET2* alterations. Thirty-one different *TET2* mutations and one whole gene deletion were detected. The mutations were scattered widely throughout the gene, without any apparent 'hotspots'. Twenty-five of the 27 *TET2*-mutated patients were *JAK2* V617F-positive and in these individuals colony analysis suggested that acquisition of *TET2* mutations occurred prior to acquisition of *JAK2* V617F. Five patients had more than one mutation and cloning in one case

confirmed that both *TET2* genes were affected. Delhommeau *et al* (2009) therefore speculated that *TET2* was a tumour suppressor gene – i.e. loss of function due to biallelic mutations or a single allele mutation and deletion of the non-mutated allele resulted in disease. The precise function of *TET2* remains obscure but *TET1*, with which it shares two domains, is known to play a role in the epigenetic regulation of embryonic stem cells (Tahiliani *et al*, 2009).

A subsequent study identified *TET2* mutations in only three of 57 (5%) ET patients (Tefferi *et al,* 2009) and again suggested *TET2* may be associated with *JAK2* V617F. The over-representation of V617F-positive patients in the Delhommeau *et al* (2009) study may therefore help explain the discrepancy in *TET2* mutation frequency between the two reports.

A single study has recently identified mutations in the additional sex combs-like 1 (*ASXL1*) gene, which is thought to be involved in regulating chromatin remodelling and is mutated in approximately 10% of MDS patients and 40% of chronic myelomonocytic leukaemia cases (Carbuccia *et al*, 2009). Of 35 ET patients studied, three had mutations in the gene (all different), one of whom also carried *JAK2* V617F and a *TET2* mutation. *ASXL1* mutations were also present in three patients with MF. Like TET2, the function of ASXL1 is unknown, but the authors suggested it too may be involved in epigenetic regulation and have a tumour suppressor role.

1.4 Aims of the project

The studies reported in this thesis set out to examine the molecular origins of ET, and in particular to investigate the role of *JAK2* V617F in the pathogenesis of the disorder and its relationship to the concept of clonality in ET. One of the major assertions made in relation to the 2008 WHO classification of MPNs has been examined, specifically that ET, like the other MPNs, is a clonal, neoplastic disorder.

The clinical features of a cohort of ET patients and their correlation with a range of molecular characteristics have been analysed (Chapter 3). Regulation of *JAK2* WT and V617F-positive platelets in V617F-positive ET has been examined in Chapter 4, in order to establish whether the V617F-positive platelets expand at the expense of WT platelets, as might be expected in a neoplastic disorder. A further series of experiments, reported in Chapter 5, sought to elicit evidence for more than one clonal population in V617F-positive ET, and thereby determine whether acquisition of *JAK2* V617F occurred more than once. Finally, inherited factors predisposing to the development of V617F-positive ET, specifically the 46/1 haplotype, are examined in Chapter 6.

2 Materials and Methods

2.1 Sample preparation

2.1.1 Reagents

- Ammonium chloride (Sigma-Aldrich, Gillingham, UK)
- Calcium- and magnesium-free phosphate buffered solution (PBS) (Invitrogen, Paisley, UK)
- Chloroform (VWR International Ltd, Lutterworth, UK)
- Double de-ionised water (ddH₂0)
- DTAB (dodecyl-trimethyl-ammonium bromide) (Sigma-Aldrich, Gillingham, UK)
- Dynal CD3 beads (Invitrogen, Paisley, UK)
- EDTA (ethylenediaminetetraacetic acid) (Sigma-Aldrich, Gillingham, UK)
- Ethanol 100% (VWR International Ltd, Lutterworth, UK)
- Ficoll-Paque[™] PLUS (GE Healthcare, Buckinghamshire, UK)
- Foetal calf serum (FCS), sterile filtered (Invitrogen, Paisley, UK)
- Giemsa stain 0.4% w/v in buffered methanol solution, pH 6.9 (Sigma-Aldrich, Gillingham, UK)
- May-Grünwald stain 0.25% w/v in methanol (Sigma-Aldrich, Gillingham, UK)

- Plastic whole blood tube with spray-coated K₂EDTA, 4mL (BD, Oxford, UK)
- RPMI-1640 medium with L-glutamine (Invitrogen, Paisley, UK)
- Sodium chloride (VWR International Ltd, Lutterworth, UK)
- Tris (Hydroxymethyl) methylamine (VWR International Ltd, Lutterworth, UK)
- TRIzol[™] (Invitrogen, Paisley, UK)

2.1.2 Buffers

DNA lysis buffer: 20g DTAB (final concentration 8%), 22g NaCl (final concentration 1.5 M), 25 mLs of 1M Tris Cl pH 7.8 (100 mM). Made up to 250mL with ddH₂0.

2.1.3 Purification of neutrophils, platelets and T-lymphocytes

Neutrophils, platelets and T-lymphocytes were purified from peripheral blood prior to DNA and RNA extraction (Figure 2.1). Peripheral blood was collected into 4mL tubes containing EDTA and centrifuged for 15 minutes at 200g at room temperature. The upper three-quarters of the plasma layer was harvested as platelet-rich plasma, pooled and spun for 15 minutes at 2000g. The supernatant was discarded and the pellet washed in 10mL calcium- and magnesium-free PBS containing 10mmol/L EDTA and centrifuged for 15 minutes at 2000g at room temperature. The washing process was repeated once more and the resulting platelet suspension counted using a Sysmex KX21N whole blood cell analyser (Sysmex, Milton Keynes UK). Aliquots of 5x10⁹ platelets were then centrifuged for 15 minutes at 2000g at room

temperature and the supernatant was carefully removed. The pellet was resuspended in 1mL TRIzol reagent by repeated pipetting and stored at -80°C for RNA extraction.

The sample remaining in each tube after removal of the platelet rich plasma, comprised mostly of erythrocytes and leukocytes, was made up to 4mL with calcium- and magnesium-free PBS containing 10mmol/L EDTA, mixed and centrifuged for 5 minutes at 2000g. The buffy coat, comprised mostly of leukocytes, was recovered from each tube and pooled. The samples remaining in the 4mL tubes were re-mixed, spun again for 5 minutes at 2000g, and any additional buffy coat recovered and pooled with the first harvest. The resulting buffy coat suspension was made up to 14mL with calcium- and magnesium-free PBS containing 10mmol/L EDTA, layered onto 12mL Ficoll-Paque[™] PLUS and centrifuged for 20 minutes at 800g at room temperature, with minimum acceleration and braking. Mononuclear cells were then harvested from the interface, resuspended in 15mL RPMI containing 2% FCS (RF2%), and spun for 5 minutes at 800g at room temperature. The supernatant was carefully removed and the pellet resuspended in 10mL RF2%.

The suspension was counted using the Sysmex analyser to obtain the total leukocyte count. Tlymphocytes were then purified from the mononuclear cell suspension using anti-CD3-coated magnetic beads (4x10⁵ beads/µL suspended in PBS containing 0.1% bovine serum albumin, BSA, and 0.02% sodium azide). For every 1x10⁶ leukocytes in the mononuclear cell suspension, 2.5µL (1x10⁶) of magnetic beads were used. These were first washed by adding them to 10mL RF2% in a reaction tube, shaking the mixture, and placing the tube against a magnet for 3 minutes. Without removing it from the magnet, the tube was inverted to discard the supernatant, leaving only the washed beads in the tube which was then removed from the magnet. The mononuclear cell suspension was added to the washed beads, the mixture shaken, and incubated for one hour at 4°C with end-over-end rotation to allow the beads to bind the CD3-positive T-lymphocytes. Following this, the CD3-negative fraction was removed by placing the tube against the magnet for 3 minutes then pouring off the supernatant by

inverting the tube whilst holding it against the magnet. The CD3-positive cells and beads remained in the tube and were resuspended in 10mL fresh RF2% and the process repeated twice, to produce a suspension of purified T-lymphocytes (bound to the beads) in 10mL RF2%.

To allow morphological analysis of their purity, aliquots of approximately 50,000 Tlymphocytes were diluted to a total volume of 300µL with RF2% and used to prepare cytospins. The cytospin slides were stained using the May-Grünwald Giemsa technique: slides were placed in May-Grünwald stain for 5 minutes at room temperature, transferred to dilute Giemsa stain (1:20 in ddH₂0) for 20 minutes at room temperature, then rinsed with ddH₂0 and allowed to air dry.

For RNA extraction, aliquots of 5x10⁶ T-lymphocytes were spun for 5 minutes at 300g at 4°C, the supernatant removed, the pellet resuspended in 1mL TRIzol reagent by repeated pipetting, and stored at -80°C. For DNA extraction, aliquots of 15x10⁶ T-lymphocytes were spun for 5 minutes at 800g at room temperature, the supernatant removed, the pellet resuspended in 2.5mL RF2%, and 5mL DTAB lysis buffer added. The mixture was shaken and incubated for 5 minutes at 68°C then stored at -20°C for at least 12 hours prior to DNA extraction.

To obtain purified neutrophils, residual Ficoll-Paque[™] PLUS was removed after the MNCs had been harvested, the pellet disrupted by vortexing, resuspended in 2.5mL PBS and 22.5mL 0.8% ammonium choride solution added to lyse red blood cells. The suspension was vortexed and incubated on ice for 15 minutes, then spun for 5 minutes at 800g, the supernatant removed, and the pellet resuspended in 10mL RF2%. The neutrophil suspension was counted using the Sysmex analyser.

Cytospin slides of the purified neutrophils were prepared as described above for Tlymphocytes. The purified neutrophils were stored in TRIzol reagent for RNA extraction and in DTAB lysis buffer for DNA extraction, as described above.

2.1.4 RNA extraction

Samples stored in TRIzol reagent were thawed at room temperature and 0.2mL chloroform was added. The mixture was shaken vigorously, incubated at room temperature for 3 minutes, and spun for 15 minutes at 12,000g at 4°C. The aqueous upper phase of the resulting mixture was transferred to a tube containing 0.5mL isopropyl alcohol, shaken, incubated at room temperature for 10 minutes, then spun for 10 minutes at 12,000g at 4°C to precipitate the RNA. The supernatant was carefully removed and 1mL 75% ethanol added. The mixture was vortexed, spun for 5 minutes at 7,500g at 4°C, the supernatant removed, and the RNA pellet allowed to air-dry for 10 minutes then suspended in 500µL ddH₂0. RNA samples were stored at -80°C.

2.1.5 Genomic DNA extraction

Samples stored in DTAB were thawed at room temperature and 7.2mL chloroform was added. The mixture was shaken vigorously and spun for 15 minutes at 3000g at room temperature. The aqueous upper phase of the resulting mixture was transferred to a tube containing 8mL absolute ethanol, shaken, then spun for 5 minutes at 1000g at room temperature to precipitate the DNA. The supernatant was carefully removed and 2mL 75% ethanol added. The mixture was shaken, spun for 5 minutes at 1000g at room temperature, and the supernatant removed. The DNA pellet was allowed to air dry for 15-20 minutes, then suspended in 50-300µL ddH₂O depending on the size of the pellet and incubated overnight at 4°C with rotation. The DNA concentration was measured using a NanoDrop ND-1000 UV-Vis spectrophotometer (Thermo Fisher Scientific, Wilmington, USA), and working dilutions of 50ng/µL made. All DNA

samples were stored at 4°C.

2.2 Molecular biology

2.2.1 Reagents

- Acetonitrile (Transgenomic Ltd, Glasgow, UK)
- Agarose (Bioline, London, UK)
- 10x Bioline ammonium buffer (Bioline, London, UK)
- Bioline magnesium chloride 50mM (Bioline, London, UK)
- BIOTAQ[™] DNA polymerase 5u/µL (Bioline, London, UK)
- Bovine serum albumin (BSA) 10mg/mL (New England Biolabs, Hitchin, UK)
- Bromophenol blue (VWR International Ltd, Lutterworth, UK)
- Carbenicillin (Invitrogen, Paisley, UK)
- N,N-Dimethylformamide ≥99.8% (Sigma Aldrich, Gillingham, UK)
- DNA size standard PA400 (Beckman Coulter UK Ltd., Buckinghamshire, UK)
- dNTPs 100mM each (Bioline, London, UK)
- Ethidium bromide 10mg/mL (Invitrogen, Paisley, UK)
- Glycogen 20mg/mL (Sigma Aldrich, Gillingham, UK)
- GoTaq[®] DNA polymerase 5u/µL (Promega, Madison, USA)

- GoTaq[®] Flexibuffer (Promega, Madison, USA)
- GoTaq[®] magnesium chloride 25mM (Promega, Madison, USA)
- Luria-Bertani (LB) broth tablets (Sigma Aldrich, Gillingham, UK)
- Nusieve low melting point agarose (VWR International Ltd, Lutterworth, UK)
- Olignucleotide primers (Integrated DNA Technologies, Leuven, Belgium)
- Optimase[®] reaction buffer (Transgenomic Ltd, Glasgow, UK)
- Optimase[®] magnesium sulphate 25mM (Transgenomic Ltd, Glasgow, UK)
- Optimase[®] DNA polymerase (Transgenomic Ltd, Glasgow, UK)
- Orthoboric acid (VWR International Ltd, Lutterworth, UK)
- peqGOLD MicroSpin Cycle-Pure PCR Purification Kit (Peqlab, Sarisbury Green, UK)
- 5x Phusion[®] HF buffer containing MgCl₂ 7.5mM (New England Biolabs, Hitchin, UK)
- Phusion[®] Hot Start DNA Polymerase 2u/μL (New England Biolabs, Hitchin, UK)
- QIAquick[™] Gel Extraction Kit (Qiagen, Crawley, UK)
- Restriction enzymes and buffers (New England Biolabs, Hitchin, UK)
- Sample loading solution (SLS) (Beckman Coulter UK Ltd., Buckinghamshire, UK)
- SDS (sodium dodecyl sulphate) (Sigma Aldrich, Gillingham, UK)
- 3M sodium acetate pH 5.2 (Sigma-Aldrich, Gillingham, UK)
- Superscript[™] III First-Strand Synthesis SuperMix for qRT-PCR kit (Invitrogen, Paisley, UK)

- TOPO TA Cloning[®] Kit with One Shot[®] MAX Efficiency[™] DH5α-T1R E. coli competent cells (Invitrogen, Paisley, UK)
- Tri-ethylene ammonium acetate (TEAA) (Transgenomic Ltd, Glasgow, UK)
- Tris (Hydroxymethyl) methylamine (VWR International Ltd, Lutterworth, UK)
- X-Gal (Sigma Aldrich, Gillingham, UK)

2.2.2 Buffers

- 10x TBE (pH8.3), for 1 litre: Tris 108.9g, orthoboric acid 55.7g, EDTA 7.4g
- Loading buffer: 30% glycerol, 0.025% bromophenol blue in 1xTBE

2.2.3 Polymerase chain reaction (PCR)

Four different DNA polymerases were used in the PCR experiments reported in this thesis. The general reaction conditions for each polymerase are described below, and variations from these for specific situations are described in the appropriate chapters. For every DNA polymerase, a master mix containing all necessary reagents except DNA template was prepared and aliquoted according to the number of samples prior to the addition of DNA template. Thermal cycling was performed using a DNA Engine Tetrad 2 Peltier Thermal Cycler (BioRad Laboratories, Hemel Hempstead, UK) using a ramping speed of 1.0°C/S unless otherwise specified.

All oligonucleotide primer sequences are listed in Appendix 1.

2.2.3.1 BIOTAQ[™] PCR

BIOTAQ[™], a non-proofreading Taq polymerase, was used for the majority of the experiments reported in this thesis which did not require an enzyme with 3' to 5' exonuclease proofreading properties.

A master mix was prepared containing 1x reaction buffer [final concentrations: 16mM (NH₄)₂SO₄, 67mM Tris-HCl, 0.01% Tween-20], 1mM MgCl₂, 0.5µM each oligonucleotide primer, 0.2mM each dNTP, 0.05unit/µL polymerase, and made up to the desired volume with ddH₂O. To this mix, 2.5ng/µL template DNA was added. Cycling conditions were 30 seconds denaturation at 95°C, 30 seconds primer annealing at a temperature specific to the primers used, and 30 seconds primer extension at 72°C, repeated between 22 and 40 times, depending on the nature of the PCR (see specific methods). A final step of 5 minutes at 72°C was included to ensure complete extension of the PCR products.

2.2.3.2 GoTaq[®] Hot Start PCR

GoTaq[®] is also a non-proofreading Taq polymerase, but differs from BIOTAQ[™] polymerase in that it is bound to a proprietary antibody which inactivates it until the reaction mixture is heated to 94°C or above for 2 minutes, whereupon its activity is restored ('hot start' activity). This minimizes non-specific DNA amplification at lower temperatures. GoTaq[®] polymerase was therefore used when BIOTAQ[™] polymerase generated multiple non-specific PCR products.

A master mix was prepared containing 1x Promega flexibuffer (proprietary formulation), 2mM MgCl₂, 0.05unit/ μ L GoTaq[®] Hot Start polymerase, 0.5 μ M each oligonucleotide primer, 0.2mM each dNTP, and made up to the desired volume with ddH₂0. To this mix, 2.5ng/ μ L template DNA was added. Following an initial incubation for 2 minutes at 95°C to activate the polymerase enzyme, between 22 and 35 cycles of PCR were performed, each cycle comprising a 30 second denaturation step at 95°C, a 30 second primer annealing step at a temperature

specific to the primers used, and a 45 second primer extension step at 72°C. After the last cycle, a final step of 5 minutes at 72°C was included to ensure complete extension of the primers.

2.2.3.3 Optimase® PCR

Optimase[®] is a DNA polymerase with 3' to 5' exonuclease proofreading activity which was used when high-fidelity PCR amplification was required, most commonly to generate products for analysis by denaturing high-performance liquid chromatography (dHPLC).

A master mix was prepared containing 1x reaction buffer (proprietary formulation), 1.5mM MgSO₄, 0.05unit/µL Optimase[®] polymerase, 0.5µM each oligonucleotide primer, 0.2mM each dNTP, and made up to the desired volume with ddH₂0. To this mix, 2.5ng/µL template DNA was added. Following an initial denaturation step of 5 minutes at 95°C, between 22 and 35 cycles of PCR were performed, each cycle comprising a 30 second denaturation step at 95°C, a 30 second primer annealing step at a temperature specific to the primers used, and a 60 second primer extension step at 72°C. After the last cycle, a final step of 5 minutes at 72°C was included to ensure complete extension of the primers.

2.2.3.4 Phusion® PCR

Phusion[®] was used to generate PCR products from very low concentrations of template DNA or cDNA (e.g. from old or degraded samples), especially when longer PCR products (>500bp) were required. Like Optimase[®], it is a proofreading polymerase, making it suitable for applications such as dHPLC, and like GoTaq[®], it has 'hot start' activity. It was generally used when no satisfactory products could be obtained using one of the other DNA polymerase enzymes. A master mix was prepared containing 1x HF buffer (proprietary formulation; contains MgCl₂ at a final concentration of 1.5mM), 0.02unit/µL Phusion® polymerase, 0.5µM each oligonucleotide primer, 0.2mM each dNTP and made up to the desired volume with ddH₂0. To this mix, 2.5ng/µL template DNA was added. Following an initial denaturation/activation step of 2 minutes at 98°C, between 32 and 40 cycles of PCR were performed, each cycle comprising a 42 second denaturation step at 98°C, a 42 second primer annealing step at a temperature specific to the primers used, and a 60 second primer extension step at 72°C. After the last cycle, a final step of 5 minutes at 72°C was included to ensure complete extension of the primers.

2.2.4 Restriction enzyme digestion

Restriction enzymes were obtained from New England Biolabs. Genomic DNA or PCR products were added to a mixture of the appropriate 1x reaction buffer, restriction enzyme, ddH_20 and, if necessary, 100µg/ml BSA, according to the manufacturer's instructions, and incubated at the recommended temperature for 2-12 hours, according to manufacturer's guidance.

2.2.5 Reverse-transcription PCR (RT-PCR)

Complementary DNA (cDNA) was generated from RNA by reverse transcription, using the SuperScript[™] III First-Strand Synthesis SuperMix for qRT-PCR kit. RNA concentration was measured using the NanoDrop ND-1000 UV-Vis spectrophotometer, and 100ng-1µg of RNA was used as template. The RNA sample was made up to 8µL with ddH₂O and added to a mastermix containing 2µL RT enzyme mix (a proprietary formulation of SuperScript[™] III reverse transcriptase and RNaseOUT[™] recombinant ribonuclease inhibitor) and 10µL 2x RT

reaction mix (final concentration 5mM MgCl₂, 1.25ng/ μ L random hexamers, 0.5mM dNTPs, 1.25 μ M oligo(dT)₂₀ primers). The reaction mixture was mixed gently and incubated for 10 minutes at 25°C, followed by 30 minutes at 50°C, 5 minutes at 85°C, then chilled on ice. Following the addition of 1 μ L *E. coli* RNaseH (2U/ μ L), the mixture was incubated for 20 minutes at 37°C. The resulting cDNA was stored at -80°C until needed and used in a 1 in 4 dilution as the template for PCR reactions.

2.2.6 Agarose gel electrophoresis

PCR products were visualised using agarose gels (1.5%-3.5%). Agarose (0.53-1.23g, depending on desired gel concentration) was added to 35mL 1x TBE, dissolved by heating in a microwave oven, and 3.5µL 1mg/mL ethidium bromide added. The gel was cooled, poured into a mould, allowed to set and immersed in 1x TBE containing 100ng/mL ethidium bromide running buffer. PCR product or restriction enzyme digest was mixed with loading buffer (30% glycerol, 0.025% bromophenol blue in 1xTBE), loaded into the wells and electrophoresed by the application of a potential difference across the gel. The use of a DNA ladder allowed estimation of the PCR product size. The gel was then placed on a UV transilluminator to visualise PCR products, and digital images made.

2.2.7 Fragment analysis of PCR products and restriction enzyme digestion products

Fragment analysis was used to quantify the relative amount of multiple products generated by PCR or following restriction enzyme digestion.

PCR was performed using one fluorescently-labelled primer and one unlabelled primer. The DNA polymerase used varied according to the exact application, as did the number of cycles

and the final concentration of each primer. Specific details are described in the appropriate chapter. Where appropriate, PCR products underwent restriction enzyme digestion as described in Section 2.2.4.

To perform fragment analysis, 2µL of the PCR or restriction enzyme digestion product were mixed with 0.5µL size standard and 38µL sample loading solution (deionised formamide), and run on the Beckman Coulter CEQ 8000 DNA Genetic Analysis System (Beckman Coulter UK Ltd., Buckinghamshire, UK). The instrument denatured the sample and separated the different PCR products according to fragment length by capillary electrophoresis. Elution of each product was detected by laser-induced excitation of its fluorescent label, and indicated by a peak at the appropriate elution time on a graph generated by the supplied software. By reference to the size standards, each product's size was estimated by the software based on its elution time. The area under the curve for each peak reflected the relative amount of each product, derived by the software from the amplitude and duration of its fluorescence. The relative amount of each product was calculated by expressing the area under the appropriate peak as a proportion of the total area under all the peaks.

2.2.8 DNA sequencing

For DNA sequencing, PCR products were prepared and then purified using the peqGold microspin cycle-pure kit. One volume PCR product was mixed with 10 volumes buffer CP (proprietary formulation) then vortexed and the mixture applied to a PerfectBind DNA spin-column assembled in a 2mL collection tube and spun for 2 minutes at 10,000g at room temperature. The spin-column was transferred to a clean 1.5mL collection tube, 20µL DNA Elution Buffer (proprietary formulation) applied to the column matrix and incubated for 5 minutes at room temperature, then spun for 1 minute at 8,000g at room temperature to elute the purified PCR
products into the collection tube. These were sequenced by the UCL Wolfson Institute for Biomedical Research DNA sequencing service using an Applied Biosystems 3730*xl* 96-capillary DNA analyser (Life Technologies, California, USA) with BigDye[®] 3.1 chemistry.

2.2.9 LB-carbenicillin culture medium and plates

For LB-carbenicillin culture medium, 12 LB medium capsules were dissolved in 500mL ddH₂0 and autoclaved, then cooled to approximately 60°C, and carbenicillin added to a final concentration of 100µg/mL. For LB-carbenicillin-agar plates, 7.5g agar was added to the mix prior to autoclaving (final concentration 1.5% w/v); after the addition of carbenicillin the medium was poured into Petri dishes, allowed to set for one hour, and stored at 4°C. Prior to bacterial spreading, plates were incubated at 37°C for 30 minutes to dry, and spread with 40µL X-gal solution (40mg/mL X-gal in dimethylformamide), then incubated at 37°C for a further 30 minutes.

2.2.10 TA cloning

TA cloning allows individual PCR products to be analysed by ligating them into a plasmid which is then transfected into competent bacteria and these are then cultured on LB-carbenicillinagar plates. Multiple, discrete bacterial clones are generated, each containing a PCR product.

The cloning studies reported in this thesis used the Invitrogen TOPO TA Cloning[®] Kit with One Shot[®] MAX Efficiency[™] DH5α-T1R *E. coli*. The plasmid vector in this kit has single 3'-thymidine (T) overhangs, bound covalently to topoisomerase I, which allow efficient ligation of PCR products generated using non-proofreading Taq polymerases as these add a single deoxyadenosine (A) to the 3' end of PCR products. Cloning was performed according to

manufacturers' instructions. Briefly, 1 to 4 μ L of fresh PCR product were added to 1 μ L salt solution and 1 μ L TOPO vector, made up with de-ionised water to 6 μ L, and incubated for 30 minutes at room temperature. A vial of One Shot® Chemically Competent *E. coli* was thawed on ice, 2 μ L of the ligated PCR product added, and incubated on ice for 30 minutes. Following incubation at 42°C for 30 seconds without shaking, the vial was transferred to ice, 250 μ L room temperature SOC medium added, and incubated at 37°C for 1 hour with shaking (200rpm). Between 5 μ L and 50 μ L cell suspension was then spread on pre-warmed LB-agarosecarbenicillin plates coated with X-gal (section 2.2.9), and the plates incubated at 37°C overnight. Transformed colonies containing an inserted PCR product were identified by their white or light-blue colour due to disruption of the lacZ α gene in the pCR®2.1-TOPO® plasmid. Individual colonies were picked and expanded by incubation in LB-carbenicillin medium at 37°C for 12 hours.

3 Clinical and molecular characteristics of the ET patient cohort

3.1 Introduction

Despite substantial improvements in our understanding of ET since it was first described over 75 years ago (Epstein and Goedel 1934), the diagnosis and management of the condition remain challenging. Distinguishing between ET and other causes of thrombocytosis, such as other MPNs, infection, bleeding, malignancy, iron deficiency, autoimmune inflammatory disorders and hyposplenism, is often difficult and in many cases remains a process of exclusion. Similarly, prognostication in ET can be challenging since ET is a heterogeneous disorder and patients vary markedly in their clinical course. For some, ET represents a relatively benign condition with a comparatively low risk of thrombosis and exerts little, if any, effect on their life expectancy. For others, the disorder is associated with major morbidity – such as life-threatening arterial and venous thromboses or haemorrhage – and premature death. Since, in general, more aggressive disease justifies more intensive treatment, estimating at diagnosis the risk of later complications is an essential part of deciding on appropriate therapy. Considerable efforts have therefore been made to improve the diagnostic approach to individuals with thrombocytosis, and then, in those with ET, to identify clinical and pathological features which predict outcome.

3.1.1 Diagnostic criteria for ET

Criteria for the diagnosis of ET were first proposed in 1975 by the PVSG (Berlin 1975). These required the presence of a sustained platelet count exceeding 1000×10^9 /L, a bone marrow showing megakaryocytic hyperplasia without significant fibrosis or myelodysplasia, and the absence of polycythaemia, CML, iron deficiency or a reactive cause for the thrombocytosis such as infection, malignancy or an inflammatory process. Subsequent revisions of the PVSG criteria have reduced the platelet threshold to 600×10^9 /L (Table 3.1; Murphy *et al*, 1986; Murphy *et al*, 1997). Despite claims that a threshold of 400×10^9 /L would allow more ET patients to be detected and treated (Sacchi *et al*, 2000), this cut-off was rejected, largely due to concerns that some of the many patients with a reactive thrombocytosis of between 400×10^9 /L and 600×10^9 /L would be erroneously diagnosed with ET in the absence of a robust means of distinguishing between the two conditions. Consequently the initial WHO diagnostic criteria stipulated that patients must have a minimum platelet count of 600×10^9 /L to be diagnosed with ET (Jaffe 2001).

Table 3.1 PVSG Criteria for the diagnosis of ET

All seven criteria must be fulfilled to make a diagnosis of ET

1	Platelet count greater than 600 x 10 ⁹ /L
2	Haematocrit less than 40% or normal RBC mass
3	Stainable iron in the marrow or normal RBC mean corpuscular volume (If these measurements
	suggest iron deficiency, PV cannot be excluded unless a trial of iron therapy fails to increase the
	RBC mass into the polycythaemic range.)
4	No Philadelphia chromosome or BCR-ABL gene rearrangement
5	Collagen fibrosis of the bone marrow absent or less than one third of the biopsy area without
	both marked splenomegaly and a leukoerythroblastic blood film
6	No cytogenetic or morphologic evidence for a myelodysplastic syndrome
7	No cause for a reactive thrombocytosis

From Murphy *et al* (1997)

The discovery of JAK2 V617F and MPL mutations in approximately 50% and 5% respectively of ET patients (Tables 1.7 and 1.8) had a significant impact on the diagnosis of ET. They both facilitated the distinction between reactive thrombocytosis and ET and enabled identification of a population of individuals with a mild thrombocytosis (below 600x10⁹/L) who carried these mutations and suffered the same thrombotic complications as ET patients with higher platelet counts (Michiels et al, 2007). As a result, a reduction of the platelet threshold was once again proposed (Tefferi *et al*, 2007) and on this occasion a lower value of 450×10^9 /L was adopted by the WHO in their revised 2008 criteria, shown in Table 1.1 (Swerdlow et al, 2008). The 2008 WHO classification requires the presence of MK hyperproliferation in the bone marrow, with large, mature MKs and normal erythroid and myeloid precursors, and the exclusion of other MPNs. Unlike previous criteria, it does not stipulate that reactive causes of a thrombocytosis must be systematically excluded if a molecular marker (e.g. JAK2 V617F) is detected. The 2008 WHO classification is likely to become the most widely accepted diagnostic criteria for the diagnosis of ET, and in the UK it has recently been incorporated into the British Committee for Standards in Haematology guideline for the investigation of individuals presenting with a thrombocytosis, although the diagnostic criteria in this guideline are subtly different from the WHO ones in that they do not require that a bone marrow biopsy be performed if a molecular marker is detected (Harrison et al, 2010).

3.1.2 Impact of clinical features at presentation on prognosis

In the 2008 WHO monograph, ET, along with the other conditions hitherto called myeloproliferative *disorders*, was reclassified as a neoplasm (Swerdlow *et al*, 2008). Nonetheless, in many patients – particularly older ones who comprise the majority of affected individuals – the diagnosis of ET has little or no effect on life expectancy (Table 3.2). Bazzan *et al* (1999) found that the life expectancy of older ET patients (>55 years at diagnosis) was

similar to that of age-matched healthy controls, whereas that of younger patients was shortened compared to controls. Although Mesa *et al* (1999) reported that the overall survival of a small cohort of ET patients was poorer than that of age-matched controls, Passamonti *et al* (2004) found that ET patients of all ages had a similar life expectancy to healthy controls. Certainly, in the first 10 years following diagnosis, survival appears to be equivalent to controls although it may be worse thereafter (Wolanskyj *et al*, 2006). The different findings of these groups probably reflect differences in the composition of the patient cohorts they studied, and a number of biological characteristics at diagnosis have now been identified which are associated with inferior survival in ET (Table 3.2).

The most frequent complications observed in individuals with ET are thromboembolic, with a historical rate of approximately 7% per patient-year following diagnosis (Cortelazzo *et al*, 1990), and a 10-year cumulative incidence (CI) of thrombosis of up to 42% (Wolanskyj *et al*, 2006), although this varies widely (see below). Treatment with cytoreductive agents has been shown to reduce the rate of thromboembolic events in ET patients (see Section 3.1.4), but there is concern that some of these agents may also increase the risk of leukaemic transformation. For this reason, considerable efforts have been made to predict the relative risks of thrombotic complications in order to select those patients who are most likely to benefit from cytoreductive medications, and conversely to identify patients for whom such treatment is not indicated (Table 3.3).

Two independent clinical risk factors – increasing age and prior history of thrombosis – are now widely accepted to be associated with an increased rate of thromboembolic complications in ET patients, although several other factors have also been proposed. In their retrospective study of 100 ET patients, some of whom were treated with busulphan, Cortelazzo *et al* (1990) observed that the average thrombotic rate in events per patient-year varied markedly according to age: in individuals younger than 40 years it was 1.7%; in those aged 40-60 years it was 6.3%; and in those older than 60 years it was 15.1%. The equivalent

Reference	n	Median follow up (years)	Median survival (years)	Risk factors associated with inferior survival	Overall survival compared to aged-matched controls
(Bazzan et al, 1999)	187	4.1	Not reported	None identified	Significantly worse for patients <55 years old at diagnosis: mortality rate was 4.2x that of matched controls, mostly due to thrombosis. Note only 58% of patients were on anti-platelet agents
					Life expectancy similar to controls for patients > 55 years old at diagnosis
(Passamonti	425	0.2	22.6	Prior thrombosis	
et al, 2004)	435	9.5	22.0	Male sex	Similar regardless of age
(Chim <i>et al,</i>	224	Not	12	Age >60 at diagnosis	
2005)	231	reported	13	Cigarette smoking	Not reported
				Age >60 at diagnosis	
				Prior thrombosis	
(Wolanskyj <i>et al,</i> 2006) 322 13.6	18.9	Diabetes mellitus	decade after diagnosis, but		
				Cigarette smoking	significantly worse thereafter
				WCC >15 at diagnosis	
(Dan <i>et al,</i>	201	1.2	Not	Age >60 at diagnosis	
2006)	201	4.5	reported	Prior thrombosis	Similar to controls
				Age >60 at diagnosis	
				Prior thrombosis	
(Gangat <i>et</i>	6 0-	_		Diabetes mellitus	
al, 2007)	605	7	18	Cigarette smoking	Not reported
				WCC >15 at diagnosis	
				Anaemia	

Table 3.2Recent studies analysing life expectancy in ET

		P-valı	ue for associati	on between	specified ris	k factor and p	ost-diagnosi:	s thrombotic	rate*	Comments
rence	2	Age > 60 years	Prior thrombosis	Diabetes mellitus	Hyper- tension	Hyper- lipidaemia	Cigarette smoking	Platelet count	White cell count	
telazzo (1990)	100	<0.001	<0.0005	I	NS	NS	NS	I	I	P values for comparison against 200 normal controls
ombi (a)	103	NS	<0.001	I	I	I	I	NS	I	Age analysed as a continuous variable
ses <i>et</i> 999)	148	0.0007	<0.0001	NS	0.01	0.0002	NS	SN	Ι	Platelet count analysed as < or > 1000x10 ⁹ /l In multivariate analysis, risk of thrombosis was predicted by: age >60 (P=0.002), prior thrombosis (P=0.001), and hyperlipidaemia (P<0.001)
tunen , 2001)	132	NS	I	NS	NS	NS	0.02	SN	I	Age analysed as a continuous variable. Male sex also associated with an increased thrombotic risk (P=0.0001)
(2)	89	<0.001	I	NS	0.002	NS	I	SN	NS	Age analysed as a continuous variable In multivariate analysis, only increasing age associated with an increased thrombotic rish (P=0.03)
n <i>et</i> 005)	231	NS	NS	I	I	I	NS	NS	NS	WCC categorised as < or > 11x10 ⁹ /L Platelet count categorised as < or >

Table 3.3Studies assessing conventional risk factors for thrombosis in ET

Chapter 3

<pre>#P value depended on WCC cut-off value (P=0.04 for 7.2x10⁹/L, 0.02 for 8.7x10⁹/L, and 0.008 for 10.4x10⁹/L)</pre>	0.04- 0.008‡	NS Itinued	- In ET – con	- - -	- actors for th	- - onal risk f	0.008+ ng conventi	008†	0.0 dies	657 0.0 Studies
[†] Age > 60 and/or prior thrombotic event combined as 'high-risk' category [‡] P value depended on WCC cut-off value (P=0.04 for 7.2x10 ⁹ /L, 0.02 for 8.7x10 ⁹ /L)	0.04- 0.008‡	NS	I	I	I	I		0.008+	0.008+	657 0.008† 0.008†
In multivariate analysis thrombotic risk was predicted by age > 60 (P=0.004), prior thrombotic event (P=0.004) and WCC \ge 8.7x10 ⁹ /L (P=0.001)	0.01	NS	I	I	I	1		0.02	0.02 0.02	439 0.02 0.02
WCC was categorised as < or > 8.7x10 9 /L										
In multivariate analysis thrombotic risk was predicted by age >60 (P<0.001) and WCC > 15x10 ⁹ /L (P=0.02)										
WCC was categorised as < or > $15x10^9$ /L	0.005	I	NS	NS	0.02	0.03		<0.001	0.002 <0.001	322 0.002 <0.001
Prior thrombotic complications included only prior arterial events										
Splenomegaly at diagnosis associated with a reduced thrombotic risk										
1200×10°/L										

P value is for univariate comparison against ET patients in the same study without the specified risk factor unless stated otherwise in the comments column. Dash (-) indicates analysis not performed. NS = not significant. *

rate in a cohort of 200 haematologically normal controls was 1.2%. The thrombotic risk was significantly higher than the controls in the oldest age group whereas in the two younger categories the difference was not significant. A number of later reports also found that increasing age was associated with an increased thrombotic risk (Besses *et al*, 1999; Carobbio *et al*, 2007; Carobbio *et al*, 2008a; Shih *et al*, 2002; Wolanskyj *et al*, 2006) although none of these had control arms and instead compared older ET patients with younger ones. Cortelazzo *et al* (1990) also noted that ET patients who had had a prior thrombotic event were at significantly higher risk of a subsequent thrombosis than those who had never had a thrombosis (OR = 13, P < 0.0005), a finding corroborated by several other studies (Besses *et al*, 1999; Carobbio *et al*, 2007; Colombi *et al*, 1991b; Wolanskyj *et al*, 2006).

The role of vascular risk factors such as hypertension, diabetes mellitus (DM), hyperlipidaemia and cigarette smoking in predicting the thrombotic risk in ET patients is uncertain. Whilst Besses *et al* (1999) found that hypertension and hyperlipidaemia, but not DM or cigarette smoking, were associated with an increased thrombotic rate, other studies have reported contradictory results. Some have noted an increased thrombotic risk associated with cigarette smoking alone (Jantunen *et al*, 2001), hypertension alone (Shih *et al*, 2002), or hypertension and DM (Wolanskyj *et al*, 2006). Cortelazzo *et al* (1990) did not detect any association between vascular risk factors and thrombotic risk.

Surprisingly, patients with the highest platelet counts seem not be at greatest risk of thrombotic complications (Table 3.3). Indeed a recent study has suggested that moderately elevated platelet counts (650-1000x10⁹/L) are associated with the highest risk of thrombosis (Carrobbio Blood 08), whereas patients with extreme thrombocytosis are most likely to suffer haemorrhagic complications due to consumption of higher molecular-weight multimers of von Willebrand factor (Budde *et al*, 1984; van Genderen *et al*, 1994; vanGenderen *et al*, 1996). Recently, leukocytosis has been identified in a number of retrospective studies as a risk factor for thrombosis (Barbui *et al*, 2009; Carobbio *et al*, 2008a; Carobbio *et al*, 2008b; Landolfi *et al*,

2007; Wolanskyj *et al*, 2006), although other studies have suggested it is of more limited prognostic value, being predictive only in younger patients (De Stefano *et al*, 2010), or of no value at all (Gangat *et al*, 2009). Other proposed risk factors for thrombosis include leukocyte activation and elevated haematocrit (Falanga *et al*, 2000; Wolanskyj *et al*, 2005).

Patients with ET are at increased risk of developing MF and AML compared to healthy controls, athough the absolute risk remains low. The largest long-term follow-up study observed 435 patients (Passamonti *et al,* 2004) and found that the 15-year CI of developing MF and AML was 4% and 2% respectively, markedly lower than the incidence of thrombosis and solid tumours (17% and 8% respectively). In a study of 386 patients, Palandri *et al* (2009a) reported the 10-year CI of fibrotic and leukaemic transformation as 5% and 3% respectively, and Wolanskyj *et al* (2006) reported equivalent figures of 4% and 1% in their study of 322 patients. Owing to the rarity of these events, there is little data relating to the risk of fibrotic or leukaemic transformation, although one study found that increasing age, anaemia, and a platelet count exceeding 1000×10^9 /L at diagnosis were associated with a higher incidence of leukaemic transformation (Gangat *et al*, 2007).

3.1.3 Impact of molecular characteristics on prognosis

The role of molecular markers in predicting outcome in ET is less well-defined than in many other haematological disorders. Nonetheless, a number have been proposed for use in prognostication, specifically the presence of myeloid clonality by XCIP analysis, and *JAK2* V617F and *MPL* W505 mutational status.

XCIP analysis exploits the cellular mosaicism present in most females with regard to Xchromosome inactivation to determine whether a cell population has arisen from a single progenitor – i.e. whether the population is monoclonal (see Section 1.3.5). If all the constituent

cells have inactivated the same X-chromosome, this is suggestive of a monoclonal population, assuming there is no evidence of constitutional or age-related skewing. By contrast, if there is a mix of inactivated X-chromosomes, this is suggestive of a polyclonal population. The XCIP may also be predictive of outcome. Of the four studies which reported on the clinical significance of the presence of a clonal myeloid population (Table 1.4), two observed that it was associated with an increased thrombotic risk (Chiusolo *et al*, 2001; Harrison *et al*, 1999a). It is worth noting, however, that in total these two studies reported 16 patients who suffered thrombotic events, but in 14 of these (88%) the first event occurred at or prior to diagnosis; consequently, these patients would be deemed high risk on clinical grounds alone, so whether XCIP status provides additional useful prognostic information is uncertain.

Approximately 50% of ET patients carry the *JAK2* V617F mutation (see Section 1.3.7), a finding which has both facilitated the diagnosis of ET and prompted comparison between those patients who harbour the mutation and those who do not. A number of studies have shown that, compared to *JAK2* WT ET patients, those who are V617F-positive tend to have higher haemoglobin concentrations and lower platelet counts (Table 3.4). In these regards they more closely resemble PV patients, who are almost invariably V617F-positive, than *JAK2* WT ET patients. It has, therefore, been argued that V617F-positive ET might represent a *forme fruste* of PV. Loss of the full PV phenotype may be due to modifiers such as sex, iron stores, erythropoietin levels, and the V617F mutant level (Campbell and Green 2006). PV is generally believed to confer a higher thrombotic rate and poorer prognosis than ET (Cervantes *et al*, 2008; Passamonti *et al*, 2004) and the suggestion that V617-positive ET resembled it immediately led to speculation that mutant-positive ET patients may have a worse outcome than WT ones. Despite extensive research, it remains uncertain whether this is truly the case (Table 3.5)

Published studies reporting presenting characteristics of ET patients according to JAK2 V617F status Table 3.4

A. Sex and age

concepto d	Total	JAK2 WT	V617F+		Female s	sex n (%)		2	Median age at c	Jiagnosis <i>year</i> s	
	u	u (%)	n (%)	Overall	JAK2 WT	V617F+	ď	Overall	JAK2 WT	V617F+	Р
(Baxter <i>et al</i> , 2005)	51	22 (43%)	29 (57%)	27 (53%)	12 (55%)	15 (52%)	NS	55	54	55	NS
(Campbell <i>et al,</i> 2005)	776	362 (47%)	414 (53%)	464 (60%)	206 (57%)	258 62%)	0.1	56	52	60	<0.0001
(Cheung <i>et al,</i> 2006)	60	31 (52%)	29 (48%)	36 (60%)	17 (55%)	19 (65%)	NS	44	41	46	NS
(Kittur <i>et al,</i> 2007)	176	80 (45%)	96 (55%)	110 (63%)	47 (59%)	63 (66%)	0.35	57	48	62	0.03
(Vannucchi <i>et al,</i> 2007)	624*	257 (41%)	368 (59%)	428 (69%)	169 (66%)	259 (70%)	NS	50	47	52	<0.05
(Palandri <i>et al</i> , 2009b)	275	100 (36%)	175 (64%)	115 (42%)	41 (41%)	74 (42%)	NS	60	55	63	0.0015
(Pardanani <i>et al,</i> 2010)	226	108 (48%)	118 (52%)	145 (64%)	60 (56%)	85 (72%)	0.01	56	50	61	0.008
Overall ⁺	2263	977 (43%)	1271 (56%)	1325 (59%)	552 (56%)	773 (61%)	[1/7]	54	50	58	[5/7]

A further 14 who were 'homozyous' for JAK2 V617F (mutant level > 50%) were not included in the authors' main analysis and are not shown here *

⁺ Values shown are means for the whole series, weighted according to the size of the study; values shown under the P-value for each parameter indicate the number of studies reporting a significant association with JAK2 mutation status out of the total studies shown Table 3.4 Published studies reporting presenting characteristics of ET patients according to JAK2 V617F status – continued

B. Full blood counts

	TotoT	TWCAN	1617E+	White	cell count at	t diagnosis	(10 ⁹ /L)	Haem	oglobin at e	diagnosis (1	0 ⁹ /L)	Platel	et count at (diagnosis (1	0 ⁹ /L)
Reference	n	u 1	u	Overall*	JAK2WT	V617F+	٩	Overall*	JAK2WT	V617F+	Ч	Overall*	JAK2WT	V617F+	٩
(Baxter <i>et al</i> , 2005)	51	22 (43%)	29 (57%)	10.3	9.4	10.9	NS	14.1	14.1	14.1	NS	968	1024	926	NS
(Campbell <i>et al,</i> 2005)	776	362 (47%)	414 (53%)	10.0	9.3	10.6	<0.0001	14.0	13.5	14.5	<0.0001	961	1030	902	<0.0001
(Cheung <i>et al,</i> 2006)	60	31 (52%)	29 (48%)	0.6	7.9	10.1	NS	13.8	13.0	14.2	0.04	872	954	867	NS
(Kittur <i>et al,</i> 2007)	60	80 (45%)	96 (55%)	10.1	9.4	10.4	NS	13.7	13.1	14.0	0.04	840	577	785	0.01
(Vannucchi <i>et al,</i> 2007)	176	257 (41%)	368 (59%)	10.0	8.9	10.2	0.008	13.9	13.2	14.2	<0.0001	1000	1130	975	0.001
(Palandri <i>et al,</i> 2009b)	624†	100 (36%)	175 (64%)	8.9	8.4	9.3	NS		Not rep	orted [‡]		850	939	787	NS
(Pardanani <i>et al,</i> 2010)	275	108 (48%)	118 (52%)	8.7	8.3	0.6	NS	14.2	13.6	14.7	<0.0001	744	825	712	<0.0001
(Baxter <i>et al,</i> 2005)	226	22 (43%)	29 (57%)	9.7	8.6	10.1	<0.0001	13.8	13.2	14.1	0.0001	1000	1109	266	0.03
Overall [§]	2263	977 (43%)	1271 (56%)	9.5	8.8	6.6	[3/8]	14.0	13.4	14.4	[6/7]	305	666	869	[2/8]
* In four studies (italicised), overall val	ues were nc	ot reportec	l, so they h	ave been	calculated	here as me	eans based	on the me	dians repo	rted for JA	K2 WT and	V617F pat	tients,

weighted according to the size of each group

A further 14 who were 'homozyous' for JAK2 V617F (mutant level > 50%) were not included in the authors' main analysis and are not shown here +

Haematocrit was similar in the JAK2 WT and V617F-positive patients (0.41 vs. 0.43, P>0.05)

ഗ

Values shown are means for the whole series, weighted according to the size of the study; values shown in square brackets under the P-value for each parameter indicate the number of studies reporting a significant association with JAK2 mutation status out of the total studies shown

Reference	n†	Patient group	%JAK2 V617F+	Median f/up (mo)	Association between JAK2 V617F and thrombosis? [‡]	Details	Confounding factors associated with JAK2 V617F status	Independent association between <i>JAK2</i> V617F and thrombosis?	Comments
(Kralovics <i>et</i> al, 2005)	244	128 PV, 93 ET, 23 IMF	48%	NR	Yes	Higher overall [§] rate of thrombosis in V617F+ group (26% vs 15%)*	Age, disease duration	QN	No separate analysis for ET subgroup
(Wolanskyj <i>et</i> al, 2005)	150	ET	49%	137	No	No difference in rate of thrombosis at any time (45% vs 38%) or after diagnosis (33% vs 29%).	Age Hb, WCC	No	Not powered to detect small differences in thrombotic rate (eg. 5% and 15%)
(Campbell <i>et</i> <i>al</i> , 2005)	776	ET	53%	NR	Yes	Higher rate of VTE <i>prior</i> to diagnosis in V617F+ group (11 vs 2 events)*. Trend towards higher rate of VTE after diagnosis (12 vs 4 events). No difference in arterial events.	Age, Hb, neutrophil count	QN	Patients mostly derived from MRC PT-1 trial
(Cheung <i>et</i> al, 2006)	60	ET	48%	NR	Yes	Higher overall rate of thrombosis in V617F+ group (62% vs 26%)**.	ЧН	QN	
(Heller <i>et al,</i> 2006)	50	EL	48%	62	Yes	Higher overall rate of thrombosis in V617F+ group (11 vs 1 events)**	Age, Hb,	Yes	Relatively young cohort (78% <60 years). Most thrombotic events occurred prior to diagnosis of ET
(Finazzi <i>et al,</i> 2007)	179	ET	58%	66	Yes	Higher overall rate of thrombosis in V617F+ group (33% vs 17%)*	нь, wcc	QN	77 further patients had PV, in whom the thrombotic rate was 43% (not significantly different to V617F+ ET group)
(Ohyashiki <i>et</i> al, 2007)	49	ET	63%	NR	Yes	Higher overall rate of thrombosis in V617F+ group (29% & vs 6%)	нь, wcc	QN	Also analysed 34 PV patients: no association between homo- or heterozygosity for <i>JAK2</i> V617F and thrombotic risk
(Carobbio <i>et</i> al, 2007)	277	ET	55%	NR	No	No difference in post-diagnosis rate of thrombosis (hazard ratio = 1.4, 95% CI 0.7 to 3.0)	Hb, WCC	No	

Studies addressing the association between JAK2 V617F and thrombosis (venous and arterial) Table 3.5 Chapter 3

Significance on multivariable analysis was lost if history of prior thrombosis was included.	Major thrombotic events only included		Those with thrombosis were significantly older and had significantly higher WCC than those without	Thrombotic risk at 5+ years post- diagnosis associated with mutant level >25%. At equivalent mutant levels, the thrombotic risk was similar for ET and PV		Independent thrombotic risk factors were haemoglobin and age
Yes ²	No	Yes	DN	Yes (see comments)	ND	No
Age, Hb, WCC	НЬ	Age, Hct	нь, wcc	Age, WCC, Hb	Hb, WCC, splenomegaly	нь, wcc
Higher rate of VTE after diagnosis in V617F+ group (11% vs 3%)*. No difference in pre-diagnosis VTE (11% vs 6%), or arterial events at any time (38% vs 29%).	No difference in post-diagnosis rate of thrombosis (9 vs 13 events).	Increased rate of arterial and venous thrombosis in ET patients with mutant level >50% (n=14) compared with all other ET patients, at diagnosis and during follow-up*.	Higher overall rate of thrombosis in V617F+ group (43% vs 11%)*	Thrombotic risk highest in PV, intermediate in V617F-positive ET, lowest in WT ET (RR 2.3, 1.5 and 1.0 respectively).	Higher overall rate of thrombosis in V617F+ group (OR 2.2; 95%Cl 1.1-4.4), due to markedly higher rate in patients <60 years old and those with mutant level >50% (n=8)	
Yes	N	Yes	Yes	Yes	Yes	Yes
59	120	60	35	49	NR	NR
55%	43%	73%	66%	57%	63%	57%
ET	ET <40 years old	323 PV, 639 ET	ET	867 ET 415 PV	ET	Ы
176	103	962	53	1282	132	141
(Kittur <i>et al,</i> 2007)	(Alvarez- Larran <i>et al,</i> 2007)	(Vannucchi <i>et</i> <i>al</i> , 2007)	(Hsiao <i>et al,</i> 2007)	Carobbio <i>et</i> al, 2009	(De Stefano <i>et al,</i> 2009)	(Chim <i>et al,</i> 2010)

Studies addressing the association between JAK2 V617F and thrombosis (venous and arterial) – continued Table 3.5

Studies addressing the association between JAK2 V617F and thrombosis (venous and arterial) – preceding pages Table 3.5

- * p<0.05; **p<0.005; 'OR' denotes odds ratio; 'CI' denotes confidence intervals; 'ND' denotes not done; 'NR' denotes not reported.</p>
- number of evaluable patients in whom JAK2 V617F status was known and clinical data was available +
- [‡] on univariate analysis
- [§] 'overall' refers to thrombosis at presentation or follow-up

A number of studies have suggested that, in ET, the presence of *JAK2* V617F is associated with an increased risk of thrombotic complications, either at diagnosis (Campbell *et al*, 2005), after diagnosis (Kittur *et al*, 2007), or at any point (Cheung *et al*, 2006; Finazzi *et al*, 2007; Heller *et al*, 2006; Ohyashiki *et al*, 2007) although in some reports the increased risk was confined to younger patients (De Stefano *et al*, 2009) or those with mutant levels greater than 50%, which are found in 2-6% of ET patients (De Stefano *et al*, 2009; Vannucchi *et al*, 2007). In the majority of these studies, however, there was no evidence of an independent association between *JAK2* mutation status and thrombotic risk once adjustment was made in multivariate analysis for confounding factors, typically greater age and white cell count which are both associated with V617F-positivity. Further, several studies have failed to find any association whatsoever between *JAK2* V617F-positivity and increased thrombotic complications (Alvarez-Larran *et al*, 2007; Caramazza *et al*, 2009; Wolanskyj *et al*, 2005).

A recent meta-analysis of 21 studies with 3150 ET patients concluded that the overall thrombotic rate was higher in V617F-positive patients (OR 1.92, 95% CI 1.45-2.53), but did not adjust for confounding factors, or distinguish between thrombotic events occurring prior to presentation and those occurring thereafter (Lussana *et al*, 2009). Clearly, it is the risk of future events which are most germane to management decisions at diagnosis, and whether *JAK2* V617F status influences this remains uncertain.

The clinical significance of the presence of *MPL* W515 mutations is difficult to evaluate due to the small numbers of ET patients in whom they have been detected. The largest published study found that, amongst 994 ET patients, 30 (3%) carried *MPL* W515 mutations and these patients had an increased rate of arterial events at diagnosis and of microcirculatory disturbance, but multivariate analysis was not attempted (Vannucchi *et al*, 2008a). Another major study detected 32 patients with *MPL* mutations in a cohort of 776 patients (4%) and reported an association between *MPL* mutations and an increased risk of venous events after

entry into the study and death compared to *JAK2* WT/ *MPL* WT patients but these associations were lost after adjusting for age and prior history of thrombosis (Beer *et al*, 2008).

3.1.4 Risk stratification and therapy for ET patients

All patients with ET are routinely prescribed low-dose aspirin, which has been shown to reduce the incidence of non-fatal thrombosis in PV (Landolfi *et al*, 2004), although it has never been prospectively studied in ET. Aspirin may also relieve the vasomotor symptoms associated with ET such as headaches and erythromelalgia (erythaema and a burning sensation of the hands or feet). Other therapeutic interventions in ET include addressing concomitant vascular risk factors such as hypertension, hyperlipidaemia, diabetes mellitus, obesity and smoking.

Cytoreductive medication is generally reserved for those patients at high risk of thrombotic or haemorrhagic complications. Most national or international guidelines define high risk for thrombosis as age greater than 60 years or a history of thrombotic events (Barbui *et al*, 2004; Harrison *et al*, 2010; Tefferi 2008). Thrombotic events are generally deemed to include myocardial infarction, angina pectoris, stroke, transient ischaemic attack, deep venous thrombosis, pulmonary embolism, peripheral arterial thrombosis and retinal artery occlusion, although severe erythromelalgia unresponsive to anti-platelet therapy may also be included (Barbui *et al*, 2004). Patients at high risk of bleeding are those with a history of major haemorrhage or who have extreme thrombocytosis, although the platelet threshold used for implementation of cytoreduction varies between 1000 x10⁹/L and 1500 x10⁹/L (Barbui *et al*, 2004; Harrison *et al*, 2005; Harrison *et al*, 2010; Tefferi 2008).

It is unclear whether any ET patients who are younger than 60 years old and do not have a prior history of thrombotic events or an extreme thrombocytosis would also benefit from cytoreduction. The 'intermediate risk' arm of the MRC Primary Thrombocythaemia 1 (PT-1)

Study is currently investigating the role of cytoreduction in patients aged 40-60 years without high-risk features. Some experts have recommended that patients within this age group with additional vascular risk factors such as diabetes or hypertension requiring therapy, familial thrombophilic defects and leukocytosis should also be considered for cytoreductive therapy (Barbui *et al*, 2004; Carobbio *et al*, 2008a; Harrison *et al*, 2005).

3.1.5 Choice of cytoreductive agent

The first-line of cytoreductive therapy is usually hydroxycarbamide (formerly called hydroxyurea), which exerts its cytotoxic effects by inhibition of ribonucleotide reductase and hence DNA synthesis, and has been shown to reduce the incidence of thrombotic complications in 'high-risk' ET patients (Cortelazzo et al, 1995). Anagrelide, a newer agent, is a phosphodiesterase inhibitor which selectively blocks megakaryocyte maturation without affecting neutrophil production. It is licensed in the EU for the treatment of ET patients who are intolerant of, or resistant to, hydroxycarbamide and is most commonly used when the dose of hydroxycarbamide required to control the patients' platelet counts renders them neutropenic. Interferon alpha (IFNa) inhibits TPO-induced megakaryopoiesis by inducing SOCS-1 expression, thereby suppressing proliferation of CFU-MKs and the more immature pluripotent haemopoietic progenitors (Lengfelder et al, 1996), resulting in reduction of platelet counts. Since it is not believed to be teratogenic, IFN α is the preferred treatment for ET patients requiring cytoreduction in pregnancy. Alkylating agents such as chlorambucil and busulphan and radio-isotopes such as ³²Phosphorus also have cytoreductive properties but are now rarely used in the treatment of ET as they increase the rate of progression to acute leukaemia, especially if used in combination with hydroxycarbamide (Finazzi et al, 2000; Passamonti et al, 2004; Sterkers et al, 1998). By contrast, no intervention in ET has been shown to reduce the risk of leukaemic transformation.

The high risk arm of the PT-1 study compared cytoreductive agents in 809 ET patients at high risk of thrombosis and found that hydroxycarbamide was superior to anagrelide in preventing arterial events but less effective in preventing venous events. Hydroxycarbamide was also associated with a lower incidence of myelofibrotic transformation, intolerable side effects and serious bleeding (Harrison *et al*, 2005). For this reason, some experts recommend hydroxycarbamide as the first-line cytoreductive agent in high-risk ET patients (Harrison *et al*, 2010; Tefferi 2008), although (unproven) concerns about the long-term leukaemogenic potential of hydroxycarbamide mean that others favour anagrelide or IFNα for younger patients (Barbui *et al*, 2004; Emadi and Spivak 2009).

3.1.6 Aims

In this chapter, the clinical, haematological and molecular characteristics at disease presentation of the cohort of ET patients used in subsequent studies have been analysed and the results correlated with their clinical outcome.

3.2 Patients, materials and methods

3.2.1 Patient recruitment and sample collection

Patients were recruited from the University College Hospitals London Essential Thrombocythaemia clinic and from haematology clinics in the UK between May 1996 and January 2010, and provided written, informed consent in keeping with the Declaration of Helsinki. Prior to 2008, the 1986 PVSG Criteria (Murphy *et al*, 1986) were used for the diagnosis of ET; from 2008 onwards, the WHO Criteria were used (Swerdlow *et al*, 2008).

Peripheral blood samples (approximately 25mL) were collected into EDTA and cell purification performed as described in Section 2.1.1. For the majority of patients, neutrophil DNA and RNA, T-lymphocyte DNA and RNA, and platelet RNA samples were prepared and stored in a tissue bank. At the start of the research reported in this thesis, this tissue bank contained samples from 114 ET patients. During the period of the research, samples from 19 new patients and follow-up samples from 8 existing patients were added to the bank.

3.2.2 Clinical and demographic characteristics

Patients were assessed at diagnosis for the presence of splenomegaly, bleeding and thrombotic manifestations (as defined in Section 3.1.4), and their blood counts recorded. Where possible, patients were followed up to record the use of cytoreductive and anti-platelet agent(s), the development of haemorrhagic or thrombotic complications, the onset of myelofibrosis or acute myeloid leukaemia, or death.

Clinical and blood count data was obtained from a departmental clinical database created prior to the start of this research for 59 patients, although in some cases the data here was incomplete, and from medical case notes for 62 patients. For 12 patients no presentation clinical or blood count data was available.

3.2.3 JAK2 V617F status

Patient neutrophil DNA was screened for the presence of *JAK2* V617F using PCR to amplify exon 14 of the *JAK2* gene followed by restriction enzyme digestion with *Af*/III to distinguish between wild-type and mutant alleles (Figure 3.1) as previously published (Gale *et al,* 2007). All oligonucleotide primer sequences are listed in Appendix 1.



Figure 3.1 Method for PCR of JAK2 exon 14 followed by AfIIII digestion to distinguish between *JAK2* WT and V617F alleles

Bold indicates V617F mutation site, underlining indicates location of mismatch (T>C) introduced by primer. **Afl*III recognition site is ACGTG

BIOLINE[™] PCR was performed as described in Section 2.2.3.1 using primers flanking the *JAK2* V617F mutation site in exon 14. Since no restriction enzymes were available which could discriminate between WT and mutant-positive PCR products directly, a mismatched forward primer, *JAK2*ex14/(MM)F, was designed. This introduced a mismatch immediately upstream of the mutation site to create a recognition site for *Af*/III which would cut the WT allele whilst leaving the mutant allele uncut. Using this mismatched forward primer and an intronic reverse primer, *JAK2*ex14/R, a 157bp PCR product spanning the mutation site in exon 14 was amplified. Thirty-five cycles of PCR were performed with an annealing temperature of 60°C, and PCR products were separated by agarose gel electrophoresis (2% gel) and visualised by UV transillumination.

A digestion mix was then prepared containing 5 units Af/III enzyme, 1x manufacturer's buffer 3 and 100µg/mL bovine serum albumin (BSA), made up to 3µL with ddH₂O, and to this 7µL PCR product was added and incubated for 6 hours at 37°C. Following 3.5% agarose gel electrophoresis, digestion products were visualised by UV transillumination. Af/III cut the wildtype allele to 129bp and 28bp but did not cut the mutant allele (157bp).

Screening for *JAK2* V617F had already been performed for 114 patients at the start of the research reported in this thesis, and was performed for a further 19 patients during the course of this research.

3.2.4 JAK2 V617F mutant level quantification in neutrophil gDNA

JAK2 exon 14 PCR and *Af*/III digestion was performed as described in Section 3.2.3, except that a fluorescently-labelled exonic reverse primer (*JAK2*ex14/RF) was used. Only 30 cycles of PCR were performed in order to minimise the formation of heteroduplex products, since they would not be cut by *Af*/III and would therefore mimic homoduplex V617F-positive products. Denaturation, annealing and extension times were 60 seconds in each cycle. Following digestion with *Af*/III, 2µL of each product underwent fragment analysis as described in Section 2.2.7. The wild-type and the mutant peaks were at 129bp and 157bp respectively. Mutant allele level was calculated by expressing the area under the 157bp peak as a proportion of the total area under both peaks. Each sample was analysed three times and the mean value of the three results calculated. *JAK2* mutant level analysis had already been performed for 42 patients prior to the start of the research reported in this thesis, and was performed for a further six patients during the course of this research.

3.2.5 X-chromosome inactivation pattern (XCIP) analysis

In order to determine an XCIP, it is necessary to distinguish between the two X-chromosomes and between the active and the inactivated X-chromosome (Figure 3.2). To distinguish between the two X-chromosomes, the subject must be heterozygous for a polymorphic Xchromosome gene which is known to be differentially methylated on the active and inactive Xchromosomes. In these studies, the human androgen receptor gene, *HUMAR*, was studied, since it is heterozygous in 90% of females (Allen *et al*, 1992), with different alleles distinguishable on the basis of size. This is due to the presence of a highly polymorphic variable number tandem repeat (VNTR) sequence of 11-31 trinucleotide (CAG) repeats in the first exon of the *HUMAR* gene.

The active and inactive chromosomes are distinguished on the basis of differential methylation using a methylation-sensitive restriction enzyme (*Hpa*II) followed by PCR. Since the inactivated *HUMAR* allele is methylated, subjecting the DNA to digestion with *Hpa*II will cut the active (unmethylated) allele just upstream of the VNTR region. If PCR is then performed using primers located outside the cutting site, only the intact inactive (methylated) allele will be amplified. If some cells in a population inactivate one X-chromosome and others inactivate the other X-chromosome, two different-sized PCR products will be generated, assuming the individual is heterozygous at the *HUMAR* locus. The relative amount of the two products is proportional to the relative degree of methylation of the two alleles within the cell population, and can be quantified by fragment analysis by calculating the areas under the corresponding peaks.



Figure 3.2 Principle of XCIP analysis using HUMARA

The *HUMAR* allele on the left is methylated (indicated by M), whereas the *HUMAR* allele on the right is not. Methylation-sensitive digestion with *Hpa*II leaves the methylated allele intact (left) but cuts the unmethylated allele (right). PCR with primers which anneal outside the methylation sites allows efficient amplification of the intact allele (left), whereas the cut allele is not amplified (right).

VNTR indicates variable number tandem repeat

However, the relative amounts of the final PCR products does not depend solely on the amount of each allele present in the template DNA following *Hpa*II digestion. If equal amounts of two different-sized alleles are used as template in the same PCR, the smaller allele may be preferentially amplified if a large number of cycles is used (Walsh *et al*, 1992). For this reason, only 22 cycles of PCR were performed (Gale *et al*, 1996). Furthermore, in order to identify and control for any preferential amplification of the smaller allele due to its size rather than differential methylation, the PCR was performed twice for every DNA sample: in one, *Hpa*II-digested DNA was used as template, in the other non-*Hpa*II digested DNA was used. In the latter, equal amounts of the two PCR products were expected. The results from the two PCRs were then compared and the effect of the *Hpa*II digestion was determined. In this way the

degree of methylation of each *HUMAR* allele in the original sample was calculated. This analysis was performed on both neutrophil and T-cell DNA to evaluate methylation in myeloid cells and lymphoid cells respectively. The latter are not believed to be involved in MPNs and were therefore used to reflect the patient's constitutional XCIP.

Methylation-sensitive digestion: Two reactions were set up, one with and one without *Hpall*. For the former, 4μL gDNA was added to a reaction mixture containing 2.5 units *Rsal*, 12.5 units *Hpall*, 1x manufacturer's buffer 1, in a total reaction volume of 5μL. For the second reaction, 2μL gDNA was added to a reaction mixture containing 2.5 units *Rsal*, 1x manufacturer's buffer 1, and made up to a total reaction volume of 5μL with ddH₂O. *Rsal* was included in both reactions as it cut the genomic DNA outside the primer annealing sites, regardless of methylation status, and thereby improved PCR efficiency. Both reactions were incubated at 37°C overnight.

PCR: A PCR reaction mix was prepared containing 1x Promega buffer, 0.2mM each dNTP, 1ng/μL each *HUMAR* primer (*HUMAR*/U, a fluorescently-labelled forward primer, and *HUMAR*/D, an unlabelled reverse primer), and made up to 14μL with ddH₂O. This was added to the digest products, followed by a drop of mineral oil and incubated at 95°C for 5 minutes. Samples were then held at 85°C whilst Taq polymerase mixture (1 unit Promega aTaq made up to 1μL with ddH₂O) was added, then 22 cycles PCR performed (45 seconds 95°C, 45 seconds 68°C, 30 seconds 72°C) followed by a final extension of 15 minutes at 72°C. PCR products (2μL each) were then analysed by fragment analysis as described in Section 2.2.7.

Calculation of X-chromosome inactivation status: For polymorphic individuals, two major peaks were expected – one which represented the shorter *HUMAR* allele, A, and one which represented the longer *HUMAR* allele, B (Figure 3.3). In addition, each major peak frequently had one or more 'slippage' peaks, each being smaller than the major peak in 3bp increments, caused by Taq slippage during the PCR, and the area of the largest slippage peak was added to

the corresponding major peak. Using the fragment analysis software, the total area of alleles A and B was calculated for the reaction without *Hpa*II (A^- and B^- respectively) and for the reaction containing *Hpa*II (A^+ and B^+ respectively). The relative percentage inactivation of allele A (%A) and allele B (%B) is given by the following equations:

$$%A = \frac{(A^+ / A^-)}{(A^+ / A^-) + (B^+ / B^-)}$$
 and $%B = \frac{(B^+ / B^-)}{(A^+ / A^-) + (B^+ / B^-)}$



Figure 3.3 Fragment analysis chromatogram of HUMARA PCR products from a single patient

Chromatograms of PCR products from neutrophil gDNA which has undergone restriction enzyme digestion in the absence (left) and presence (right) of *Hpa*II.

In each chromatogram, the major peaks representing the shorter and longer *HUMAR* alleles are labelled A and B respectively. A number of smaller slippage peaks are visible to the left of the corresponding major peak. For each allele the area of the largest slippage peak was added to that of the corresponding major peak (as bracketed) for use in the HUMARA calculation.

XCIP analysis data was already available for 94 patients at the start of the research reported in this thesis, and was performed for a further 11 patients during the course of this research.

3.2.6 MPL mutation screening

Mutations in *MPL* most commonly occur in exon 10 and affect residue W515. *MPL* exon 10 was therefore screened for the presence of mutations by denaturing HPLC.

3.2.6.1 Generation of a MPL W515L-positive control by site-directed mutagenesis

A mutant-positive control sample was generated by site-directed mutagenesis (Figure 3.4). Two short-length W515L-positive products were generated from normal control gDNA template using mismatched primers *MPL*W515L/F and *MPL*W515L/R (Figure 3.4A). A 149bp *MPL* W515L-positive PCR product was amplified using primers *MPL*ex10/F, which annealed just upstream of *MPL* exon 10, and *MPL*W515L/R, and a 130bp *MPL* W515L-positive PCR product was amplified using primers *MPL*ex10/R2, which annealed just downstream of *MPL* exon 10, and *MPL*W515L/F (Figure 3.4B). These two products were generated in separate reactions, using 35 cycles of BIOTAQ[™] PCR as described in Section 2.2.3.1, and an annealing temperature of 60°C. The two PCR products were electrophoresed on a 1.5% low-melting point agarose gel, visualised by UV transillumination, cut out from the gel and purified using the QIAquick Gel Extraction Kit (Qiagen, Crawley, UK) as per manufacturers' instructions.

Equal volumes of each purified PCR product were then mixed together. A 252bp W515Lpositive PCR product was amplified by BIOTAQ[™] PCR using 1µL of the PCR product mixture as template in a 20µL total reaction volume with primers *MPL*ex10/F and *MPL*ex10/R2, an annealing temperature 60°C and 25 cycles of PCR. Other conditions were as described in Section 2.2.3.1. The 252bp W515L-positive PCR product was then purified using the PEQLab

Primer MPL W515L/F		5′-TGCTGCTGAGG T<u>T</u>G CAGTTTCCTGC-3′
WT forward strand	5′-	ctgctgctgctgagg tgg cagtttcctgcacactacag-3′
WT reverse strand	3′-	GACGACGACGACTCC ACC GTCAAAGGACGTGTGATGTC-5′
Primer MPL W515L/R		3'-ACGACTCC AAC GTCAAAGGACGTG-5'



Figure 3.4 Site-directed mutagenesis to create a *MPL* W515L-mutant positive control

- A. Genomic sequence of part of *MPL* exon 10 (codon 515 in bold), showing WT forward and reverse strands, and the two mismatched primers which introduce the G>T substitution (mismatch underlined)
- B. Generation of mutant-positive product

А

Gold PCR purification kit, according to manufacturers' instructions, and the presence of the W515L mutation confirmed by direct sequencing.

Mutant levels are reported to be relatively low in ET, with a median value of approximately 15-25% for both JAK2 V617F (Vannucchi 08; Girodon 08) and MPL W515K/L (Beer 08; Jones 10). In order to ensure that the screening technique was sufficiently sensitive to identify the majority of mutant-positive samples, a positive control with a low MPL W515L level (approximately 10%) was included with every run. This control was produced by mixing nine volumes of WT gDNA with one volume of the 252bp W515L-positive PCR product of an equal concentration. In order to equalise the concentrations of the W515L-positive PCR product and the WT gDNA, the former was diluted in ddH₂O (1 in 10, 1 in 100, 1 in 1000, 1 in 10,000). The dilution which was closest in concentration to the WT gDNA was identified by using each one, as well as WT gDNA, as template in a MPL exon 10 PCR. An Optimase® PCR was performed using conditions as specified in Section 2.2.3.3, with primers spanning MPL exon 10, MPLex10/F3 (forward) and MPLex10/R2 (reverse). A 'touchdown' PCR technique was used, based on manufacturer's recommendations, which involved using an initial annealing temperature 7°C above the mean melting temperature of the two primers (65°C), and gradually reducing the annealing temperature with successive cycles: two cycles, $T_a=72^\circ$ C; two cycles, $T_a=71^\circ$ C; two cycles, T_a=70°C; three cycles, T_a=69°C; 13 cycles, T_a=68°C; 13 cycles, T_a=66°C.

Visual inspection of the PCR products by UV transillumination following agarose gel electrophoresis revealed that the 1 in 10,000 dilution of the W515L-positive PCR product and the WT gDNA generated similar strength products. WT gDNA and the 1 in 10,000 dilution of the W515L-positive product were therefore mixed together in a 9:1 ratio to generate the 10% (approximately) *MPL* W515L-positive control.

3.2.6.2 Screening for MPL exon 10 mutations using dHPLC

The Transgenomic WAVE® system (Transgenomic Ltd, Glasgow, UK) of dHPLC was used to screen for mutations of *MPL* exon 10. This technique exploits differences in the physical properties of heteroduplex and homoduplex PCR products to detect the presence of somatic mutations. Heteroduplex products are generated in a PCR when the template DNA contains a mixture of WT and mutant alleles. One strand of the heteroduplex product contains the WT allele and the other contains the mutant allele (or rather its complement). This mismatch introduces instability into the PCR product, and consequently the two strands are less strongly bound together at this site than at the equivalent site in homoduplex PCR product which is composed of two perfectly complementary DNA strands, either both WT or both harbouring the same mutation. Denaturing the PCR product by heating it to 95°C then allowing it to reanneal at progressively lower temperatures maximises heteroduplex formation.

In dHPLC, the PCR products are pumped through a polystyrene-divinylbenzene copolymer column at a temperature which has been calculated to partially denature them. The denaturation profile of a PCR product depends on its sequence and can be predicted using the manufacturer's Navigator software. The partially-denatured products bind to the column and are eluted off by progressively increasing concentrations of acetonitrile. Since the DNA strands of the heteroduplex product are less strongly bound than those of the homoduplex product, they will be elute off the column earlier. A UV detector generates a chromatogram based on the timing and intensity of DNA elution. The presence of heteroduplex products are therefore detectable as a peak arising before the main homoduplex peak.

In order to screen the cohort of ET patients for *MPL* exon 10 mutations, a 226bp fragment spanning *MPL* exon 10 was amplified using an Optimase[®] PCR as described in Section 3.2.6.1 and analysed by dHPLC. PCR products were electrophoresed on a 2% agarose gel and visualised by UV transillumination, then denatured by incubation for 5 minutes at 95°C and

slowly cooled: 1 minute at 93.5°C, and then 1 minute at progressively lower temperatures in 1.5°C increments down to 39.5°C. Denatured products were run on the Transgenomic WAVE[®] instrument at 62.5°C and 65.8°C, which were the optimal column temperatures for partial denaturation of the amplicon, calculated using Transgenomic Navigator software (Figure 3.5).

Samples with abnormal WAVE patterns were further analysed by direct sequencing of *MPL* exon 10. *MPL* exon 10 PCR products were generated as described in Section 3.2.6.1, purified using the peqGold micro-spin cycle-pure kit, and sequenced by the UCL Wolfson Institute for Biomedical Research DNA sequencing service (Section 2.2.8) using primers *MPL*ex10/3F and *MPL*ex10/R2.



Figure 3.5 Theoretical melting curves of *MPL* exon 10 PCR product generated by Transgenomic Navigator software

A 226bp PCR product amplified from gDNA using primers *MPL*ex10/F3 (forward) and *MPL*ex10/R2 (reverse) is shown, with the predicted helical fraction after running on the dHPLC instrument at 62.5°C and 65.8°C.

3.3 Results

3.3.1 Overall results and analysis by clinical risk factors at presentation

The clinical and demographic characteristics of the 133 patients studied are shown in Table 3.6. Of the 133 patients, 28 (21%) were male. At diagnosis, the median age was 48 years (range 8-84 years), 34 patients were aged 60 or over, and the median platelet count was 805x10⁹/L (range 453-4875 x10⁹/L). At the time of analysis (February 2010), the median time since diagnosis was 12.6 years (range 0.7-31.4). Follow-up data was available on 104 patients and 29 patients were lost to follow-up. The median patient age at the time of analysis was 62.8 years (range 24.6-97.8). Eighty-two patients (78% of those evaluable) received cytoreductive medication during follow-up.

Overall, 37 patients (36% of those evaluable) had a thrombotic episode, of whom 20 had suffered a thrombosis prior to diagnosis, 11 post diagnosis, and six suffered thromboses both pre and post diagnosis. The 37 patients who had a thrombosis (at any stage) were significantly older at diagnosis than those who did not (53 versus 44 years; P=0.03, t-test), but had similar presenting blood counts: median platelets 819 versus 802×10^9 /L; haemoglobin 13.6 versus 13.9g/dL; white cell count 9.5 versus 9.2 $\times 10^9$ /L (Table 3.7). There was no significant association between suffering a thrombosis prior to or at diagnosis and suffering one after diagnosis (P=0.10, Fisher's exact test). Six patients (6%) had haemorrhagic complications. Six patients (6%) transformed to MF, of whom one later developed AML and died, one underwent a stem cell transplant, one was awaiting a transplant at the time of analysis and three were lost to follow up. Six patients died (6%), one due to AML, one pneumonitis, one gliobastoma multiforme, one dementia, and two unknown causes.

Table 3.6	Clinical and demographic characteristics of ET patients included in
	study

	Data available <i>n</i>	Value
Total patients		133
Sex, n (% of total*)	133	
Male		28 (21%)
Female		105 (79%)
Median age at presentation, years (95% CI)	121	48.0 (43.4-50.0)
Median platelet count at presentation, x 10 ⁹ /L (95% CI)	117	805 (874-1073)
Median haemoglobin at presentation, g/dL (95% CI)	77	13.7 (13.3-14.0)
Median white cell count at presentation, x 10 ⁶ /L (95% CI)	50	9.4 (9.3-11.63)
Median time since diagnosis, years (95% CI)	121	12.6 (12.1-14.7)
Palpable splenomegaly, n (% of total*)	80	7 (9%)
Cytoreductive therapy ⁺ , n (% of total*)	104	
None		22 (21%)
Hydroxycarbamide		70 (67%)
Anagrelide		20 (19%)
Interferon-alpha		12 (12%)
Other (³² P, chlorambucil or busulphan)		7 (7%)
Complications, n (% of total*)	104	
Thrombosis prior to or at presentation		26 (25%)
Thrombosis post-presentation		17 (16%)
Thrombosis (at any stage)‡		37 (36%)
Bleeding (at any stage)		6 (6%)
Death (all causes)		6 (6%)
Myelofibrotic transformation		6 (6%)
Leukaemic transformation		1 (1%)

* Where percentages of the total are shown, these are the percentage of the number of patients for whom data on that parameter was available.

⁺ Patients are counted for each therapy used and so may appear more than once.

‡ Six patients had thromboses both pre- and post diagnosis

(t-test)

0.03

0.3

0.4

0.3

at any stage			
		P-value	

Median (95% CI)

44 (40.0-48.6)

13.9 (13.3-14.3)

9.2 (8.9-11.2)

802 (870-1143)

Thrombosis

Median (95% CI)

53 (47.8-56.5)

13.6 (12.8-14.0)

9.5 (8.7-13.4)

819 (796-1011)

п

37

32

21

37

No thrombosis

п

84

45

29

81

Table 3.7 Effect of presenting age and blood counts on occurrence of thrombosis

The presenting age and blood counts were only available for a proportion of patients (see Table 3.6)

3.3.2 Analysis by XCIP status

Presenting variable*

Age, years

Haemoglobin, g/dL

White cell count, x10⁹/L

Platelet count, x10⁹/L

During the research reported in this thesis, XCIP analysis was performed on samples from 10 patients, of whom two were uninterpretable due to constitutional skewing, and two were not informative at the HUMAR locus. Of the six patients with interpretable XCIPs, three (50%) had clonal myelopoiesis (i.e. greater than 75% expression of one allele, a difference between neutrophil and T-cell XCIP of greater than 20%, and patient age less than 65 years) and three (50%) had polyclonal myelopoiesis. Data was available for a further 92 patients from analysis performed prior to the start of the research reported in this thesis. Overall, therefore, XCIP results were available for 102 of the 105 female patients in the study. No PCR products were obtained in the remaining three patients.

In 36 of the 102 patients, XCIP results were not interpretable, either due to age (n=21) or constitutional skewing (n=15), and in four more patients they were not informative at the
HUMAR locus, leaving 62 patients with evaluable XCIPs. Of these, 38 (61%) were polyclonal, and 24 (39%) were clonal (Table 3.8). Clonal patients tended to be younger at presentation than polyclonal patients, 35.5 versus 43.0 years, although the difference was not quite significant (P=0.06, t-test). The median presentation platelet count was higher in clonal patients than polyclonal patients, 950 versus 780x10⁹/L, and this difference just reached statistical significance (P=0.05, t-test). The median presenting haemoglobin, 12.6 versus 13.7g/dL, was similar in the two groups (P=0.51, t-test).

Table 3.8Clinical and demographic characteristics of patients studied according
to their XCIP status

	Clonal	Polyclonal	P-value
Total patients, n (% of evaluable patients)	24 (39%)	38 (61%)	-
Median age at presentation, years (95% CI)	35.5 (29.4-43.1)	43.0 (39.0-50.9)	0.06
Median platelet count at presentation, x 10 ⁹ /L (95% CI)	950 (804-1617)	780 (717-965)	0.05
Median haemoglobin at presentation, g/dL (95% CI)	12.6 (11.8-14.2)	13.7 (12.7-14.1)	0.5
Median time since diagnosis, years (95% CI)	13.4 (11.0-17.5)	11.7 (10.3-14.0)	0.2
Cytoreductive therapy, n (% of evaluable patients*)			
No	2 (12%)	5 (17%)	1 00
Yes	15 (88%)	24 (83%)	1.00
Complications, n (% of evaluable patients*)			
Thrombosis prior to or at presentation	7 (47%)	6 (21%)	0.09
Thrombosis post-presentation	3 (20%)	3 (10%)	0.4
Thrombosis (at any stage)†	8 (53%)	8 (28%)	0.1
Bleeding (at any stage)	0	1	
Death (all causes)	1	1	
Myelofibrotic transformation	3	0	
Leukaemic transformation	1	0	

* Clinical data was available for 15 clonal patient and 29 polyclonal patients

 Three patients had thromboses both pre- and post-presentation (two with clonal XCIP and one with polyclonal XCIP)

The rate of thrombotic events was higher in clonal patients than in polyclonal patients, both prior to presentation and overall, but this difference did not reach statistical significance (47% versus 21%, P=0.09; and 53% versus 28%, P=0.11, Fisher's exact test). There was no difference in the rate of thrombosis after presentation, 20% versus 10% (P= 0.39, Fisher's exact test).

3.3.3 Analysis of JAK2 V617F status

During the research reported in this thesis, 20 ET patients were screened for the presence of the *JAK2* V617F mutation, of whom ten (50%) were mutant-positive. There was no difference in median age at presentation in mutant-positive patients compared to WT ones (53 years versus 48 years, P=0.43, t-test), in the median presenting platelet count ($734x10^{9}$ /L versus 788 $x10^{9}$ /L, P=0.51, t-test), or the rate of thrombotic complications (50% versus 30%, P=0.65, Fisher's exact test). The median mutant level in the V617F-positive patients was 15% (range 5%-21%).

JAK2 V617F screening had been performed for a further 113 patients prior to the start of the research described in this thesis. In total, therefore, data was available for 133 ET patients, of whom 55 (41%) were mutant-positive (Table 3.9). *JAK2* mutant levels were available in 48 V617F-positive patients (in the other seven patients no PCR products were obtained), and in these the median mutant level was 17% (range 5-100%). Patients who harboured the *JAK2* V617F mutation were significantly older than those who were *JAK2* WT, median age 53.0 versus 39.5 years (P<0.0001, t-test). Mutant-positive patients also had a significantly lower presenting platelet count, median 772 versus 950x10⁹/L, (P = 0.011, t-test) and a significantly higher presenting haemoglobin, 14.2g/dL versus 13.0g/dL (P=0.0002, t-test).

Table 3.9 Clinical and demographic characteristics studied according to JAK2

status

	Data available <i>n</i>	JAK2 WT*	JAK2 V617F†	P-value
	(<i>JAK2</i> WT : V617F)			
Total patients	133	78 (58%)	55 (41%)	-
Sex	133 (78 : 55)			
Male		16 (21%)	12 (22%)	1.00
Female		62 (79%)	43 (78%)	1.00
Median age at presentation, years	121 (69 : 52)	39.5 (36.4-45.0)	53.0 (50.4-59.0)	<0.0001
Median platelet count at presentation, x 10 ⁹ /L	117 (68 : 49)	950 (937-1264)	772 (747-871)	0.011
Median haemoglobin at presentation, g/dL	77 (35 : 42)	13.0 (12.4-13.4)	14.2 (13.8-14.7)	0.0002
Median white cell count at presentation, x 10 ⁶ /L	50 (18 : 32)	10.0 (8.9-13.1)	9.0 (8.8-11.6)	0.52
Median time since diagnosis, years	121 (69 : 52)	12.5 (11.6-14.9)	12.7 (11.6-15.6)	0.770
Palpable splenomegaly	80 (36 : 44)	3 (9%)	4 (10%)	1.00
Cytoreductive therapy	104 (53 : 51)			
None		14 (26%)	8 (16%)	
Hydroxycarbamide		30 (57%)	40 (78%)	
Anagrelide		14 (26%)	6 (12%)	
Interferon alpha-2a		4 (8%)	8 (16%)	
Other (³² P, busulphan or chlorambucil)		5 (9%)	2 (4%)	
Cytoreductive therapy	104 (53 : 51)			
No		14 (26%)	8 (16%)	0 221
Yes		39 (74%)	43 (84%)	0.231

Continuous variables are shown using median (95% confidence intervals), P-values using the t-test. Categorical variables are shown using n (percentage of total), P-values using Fisher's exact test.

- * Where percentages are given, these represent the percentage of the total number of *JAK2* WT patients for whom data for that parameter was available (shown in second column)
- ⁺ Where percentages are given, these represent the percentage of the total number of *JAK2* V617F patients for whom data for that parameter was available (shown in second column)

000.000				
Complications	104 (53 : 51)			
Thrombosis prior to or at presentation		10 (19%)	16 (31%)	0.256
Thrombosis post- presentation		6 (11%)	11 (22%)	0.191
Thrombosis (at any stage)‡		14 (26%)	23 (45%)	0.099
Bleeding (at any stage)		6 (11%)	2 (4%)	
Death (all causes)		0 (0%)	6 (12%)	
Myelofibrotic transformation		3 (6%)	3 (6%)	
Leukaemic transformation		0 (0%)	1 (2%)	

Table 3.9 Clinical and demographic characteristics studied according to JAK2 status – continued

[‡] Four JAK2 V617F-positive patients and two JAK2 WT had thromboses pre- and post diagnosis

There was no difference in sex-distribution, median follow up, prevalence of splenomegaly or usage of cytoreduction between the two groups. There was no difference in the incidence of thrombotic complications pre- or post-presentation. Although, the incidence of thrombosis at any time did appear higher in the V617F-positive group than in the *JAK2* WT group, 45% versus 26%, the difference did not reach statistical significance (P= 0.1, Fisher's exact test). Patients who had had a thrombosis (at any time) did, however, have a significantly higher mean mutant level than those who did not have a thrombosis, 11% versus 5% (P=0.02; t-test).

Mutant levels greater than 25% have been associated with a particularly high thrombotic risk (Antonioli *et al*, 2008; Carobbio *et al*, 2009). Of the 37 patients who had a thrombosis, six (16%) had mutant levels greater than 25%, compared to three (5%) of the 67 patients who did not have a thrombosis (P<0.0001, Fisher's exact test).

All six patients who died were *JAK2* V617F-positive, resulting in a significantly higher mortality rate in the V617F-positive group, 12% versus 0% (P=0.01, Fisher's exact test).

3.3.4 Correlation between XCIP status and JAK2 mutational status

Of the 22 JAK2 V617F-positive patients with an interpretable XCIP, 16 (73%) had a polyclonal XCIP and six (27%) had a clonal XCIP. Of the 40 JAK2 WT patients with an interpretable XCIP, 22 (55%) had a polyclonal XCIP and 18 (45%) had a clonal XCIP. There was no association between XCIP status and JAK2 genotype (P=0.19, Fisher's exact test).

An analysis was performed to determine whether, in female V617F-positive patients, the size of the mutant-positive population could account for the difference between the T-cell and neutrophil XCIP. This analysis was based on the assumption that the mutant-positive population was clonal and all of its member cells therefore shared the same inactivated Xchromosome carrying either allele A or B. Scott *et al* (2006) have reported that in ET, mutantpositive cells are heterozygous for the mutation. Therefore, if a patient has a particular mutant level (x%), the percentage of mutant-positive cells (y%) will be double this and the remainder will be WT (z%). Thus y = 2x, and z = 100-2x.

Using the constitutive (i.e. T-cell) XCIP of A_T % : B_T %, the predicted neutrophil XCIP is given by one of two possibilities depending on whether allele A or B is inactivated in the mutantpositive population. If allele A is inactivated, the predicted neutrophil XCIP (A%:B%) = [(y/100) x 100] + [(z/100) x % A_T] : [(z/100) x % B_T]. If allele B is inactivated, the predicted neutrophil XCIP (A%:B%) = [(z/100) x % A_T] : [(y/100) x 100] + [(z/100) x % B_T].

Using this calculation, the two possible neutrophil XCIPs were predicted for each patient, allowing for the inactivation of either the A or the B-allele in the mutant-positive population. The two predicted neutrophil XCIPs were then compared to the actual neutrophil XCIP observed in each patient, and the one which was closer to the observed value was identified to allow the likely inactivated X-chromosome of the mutant-positive clone to be determined.

This analysis has already been published for 20 of the patients reported in this thesis (Gale *et al,* 2007) and demonstrated that the difference between the observed neutrophil XCIP and the closer of the two predicted ones ranged between 0% and 17% – i.e. in every case the difference was within the limit of technical variation (20%) for this technique, indicating that the size of the mutant-positive population could indeed account for the difference between the T-cell and neutrophil XCIP. This analysis was then performed for the two V617F-positive female patients included in this thesis who had not previously been reported and revealed that difference between the observed and the predicted neutrophil XCIP was 14% in one patient and 27% in the other (Table 3.10).

Table 3.10JAK2 V617F-positive patients with an interpretable XCIP: the effect ofmutant level on neutrophil clonality status

	Clonality	Neuti XC	rophil CIP	T-c XC	ell IP	%	%	Predi	cted Net	utrophil	ĸci₽§	% Difference
Patient	Status*	۵%	B%	۵%	B%	V617F†	V617F+ cells‡	if A c	lone	if B c	lone	(observed – predicted)
		ζ,	D70	Ż	D 70			Α%	В%	Α%	В%	
1	С	100	0	69	31	7%	14%	73	27	59	41	27
2	Р	73	27	72	28	26%	52%	87	13	35	65	14

- * C indicates clonal, P indicates polyclonal
- † In neutrophil DNA
- ‡ Assumes heterozygous mutant status in all cells
- § Calculated using 'y' to represent the percentage of mutant-positive cells and 'z' to represent the percentage of WT cells, based on a T-cell XCIP of A_T %: B_T %, and the following equations: if the A allele is inactivated in the clone the predicted neutrophil XCIP (A%:B%) = [(y/100) x 100] + [(z/100) x % A_T] : [(z/100) x % B_T]; if the B allele is inactivated in the clone the predicted neutrophil XCIP (A%:B%) = [(z/100) x % A_T] : [(z/100) x % A_T] : [(y/100) x 100] + [(z/100) x % B_T].

3.3.5 MPL exon 10 mutation screening

PCR for *MPL* exon 10 was performed in 133 patients. PCR products were obtained for 131 of these and analysed by dHPLC. The 10% W515L-positive control was distinguishable from the wild-type control using a column temperature of both 62.5°C and 65.8°C (Figure 3.6). Interpretable WAVE chromatograms were obtained for 126 patients. Clearly abnormal chromatograms were observed in three (2%) of these patients– two females and one male. All three cases were *JAK2* WT and represented 4% of *JAK2* WT patients. Direct sequencing of these samples demonstrated mutations in *MPL* exon 10 in all three cases. One case of *MPL* W515K and two of *MPL* W515L were detected and in all three cases the height of mutant peak was smaller than that of the WT one on the sequencing chromatogram (Figure 3.7). For two further patients the WAVE chromatogram was equivocal but in these cases there was no evidence of mutations in *MPL* exon 10 on sequencing.







Figure 3.7 dHPLC chromatogram traces and corresponding direct sequencing chromatogram traces of samples with *MPL* W515 mutations

Codon 515 is boxed on the sequencing traces; vertical arrows indication loci of point mutations

3.4 Discussion

The data in this chapter address the clinical, laboratory and molecular characteristics of a cohort of 133 ET patients studied in our institution, and their relationship to outcome. Overall, the cohort of patients reported in this thesis was broadly similar to those reported by most published studies in terms of presenting characteristics and clinical course. The median age at

presentation, at 48 years, was slightly lower than the average reported in recent studies which is 54 years (Table 3.4), possibly as a result of referral bias to UCH which provides tertiary level specialist care for patients with haematological disorders. Consequently surrounding district hospitals may have preferentially referred younger patients who might be perceived to benefit most from expert management. Females represented 79% of the ET patients; this female preponderance is typical of most published series, although it is perhaps more marked than generally reported (59% of patients; Table 3.4) and it is likely that this reflected the active recruitment by the department of female patients specifically for XCIP analysis. The median presenting haemoglobin and white cell count in our cohort were similar to those reported by most other studies (13.7g/dL versus 14.0g/dL and 9.4 x10⁹/L versus 9.5x10⁹/L respectively), but the median platelet count was lower than that generally observed (805x10⁹/L versus 905x10⁹/L). The reason for this is unclear.

The incidence of thrombotic complications at or prior to presentation (25%) was similar to that reported by other groups: 22% in Bellucci *et al* (1986) and 19% in Passamonti *et al* (2004). The median duration since diagnosis for the group was 12.6 years, during which time 16% of patients suffered thrombotic complications. This is compatible with the 15-year cumulative thrombotic risk of 17% reported by Passamonti *et al* (2004) and the 18% rate of thrombosis observed by Tefferi *et al* (2001) during a median follow-up of 9.2 years. In the cohort studied here, patients who suffered a thrombosis were on average older at diagnosis than those who did not (53 versus 44 years old; Table 3.7) but both groups had similar presenting platelet counts, which is in keeping with the majority of published studies (Table 3.3).

The rates of leukaemic and myelofibrotic transformation during follow-up (1% and 5% respectively) were also similar to published data. Passamonti *et al* (2004) reported the rates of leukaemic and myelofibrotic transformation were both 5% at 15 years post-diagnosis, whilst equivalent figures at 10 years post-diagnosis were reported by Palandri *et al* (2009a) as 1% and 4% respectively, and by Wolanskyj *et al* (2006) as 3% and 5% respectively.

Sixty-two of the 105 female patients in the study had an interpretable XCIP, and of these 24 (39%) were clonal (Table 3.8), a slightly lower proportion than that reported in most published studies (range 43% to 69%, overall mean 62%; Table 1.4). It is notable, however, that with one exception (Teofili *et al,* 2002) all the studies which reported a clonal XCIP rate greater than 55% did not control for the possibility of constitutional skewing by analysing T-cell XCIPs. Consequently, some of the patients they classified as having acquired myeloid skewing might have had constitutional skewing, meaning the true figure may have been lower.

The finding that patients with clonal myelopoiesis had significantly higher presenting platelet counts than those with polyclonal myelopoiesis (median 950×10^9 /L versus 780×10^9 /L; P=0.05) has been previously reported (El-Kassar *et al*, 1997), although it was not observed by Chiusolo *et al* (2001) or Shih *et al* (2002). Of the 44 patients with interpretable XCIPs and clinical follow-up data studied in this thesis, 21 have previously been reported by Harrison *et al* (1999a), who did not find any association between presenting platelet count and XCIP (median 722x10⁹/L in clonal patients versus 778x10⁹/L in polyclonal patients, P=0.92). This discrepancy presumably reflects the different characteristics of the 23 patients reported here who were not included in the 1999 study.

A near-significant trend to younger age in clonal patients compared to polyclonal ones (35.5 versus 43.0 years, P=0.06) was noted in the cohort reported in this thesis. It is possible, however, that this trend may simply have reflected the exclusion from the analysis of 21 women with clonal XCIPs who were older than 65 on the grounds of possible age-related skewing. If these 21 patients are grouped together with the 24 women with clonal XCIPs who were 65 or younger, the median age of this combined group was no different to that of the polyclonal group (56 versus 43 years, P=0.25). As discussed above, Harrison *et al* (1999a) have previously reported on nearly half of the evaluable patients studied in this thesis and did not identify any difference in age between the clonal and polyclonal patients (48 versus 40 years, P=0.55). The other three studies which examined this issue produced conflicting data: Shih *et*

al (2002) reported that patients with a clonal XCIP were significantly younger at presentation than those with a polyclonal XCIP, whereas el-Kassar *et al* (1997) observed the reverse, and Chiusolo *et al* (2001) did not identify any association between XCIP and age at diagnosis. There does not appear to be, therefore, a clear relationship between XCIP and age at presentation.

In the cohort examined in this thesis, the overall rate of thrombotic events was higher in patients with a clonal myeloid XCIP than those whose XCIP was polyclonal, but this was not statistically significant (53% versus 28%, P=0.11). Harrison *et al* (1999a) had previously reported on many of the patients included in this thesis and found that clonal myelopoiesis was associated with a significantly higher thrombotic rate than polyclonal myelopoiesis (60% versus 15%, P=0.04). Whilst the results from these two analyses appear contradictory, the difference between them was of borderline statistical significance (P=0.05, Fisher's exact test). This slight difference may have been due to the relatively high thrombotic rate in the polyclonal patient group reported in this thesis compared to that reported in any of the other comparable studies (Table 3.11), which may have occurred by chance. It may also reflect minor methodological variations between the studies. For instance, angina was regarded in this thesis as a thrombotic episode, in keeping with current standard practice (Barbui *et al*, 2004; Harrison *et al*, 2005), but in some of the previous studies it was not.

Chiusolo *et al* (2001) observed a significantly higher thrombotic rate amongst clonal patients than polyclonal ones (41% versus 7%, P=0.04), and Shih *et al* (2002) reported a non-significant trend towards an increased thrombotic rate in patients with a clonal XCIP (33% versus 7%, P=0.07). By contrast, el-Kassar *et al* (1997) found no such association (32% versus 21%, P=0.70), although they grouped haemorrhagic and thrombotic complications together, making direct comparison with the other studies difficult. Individually these studies are too small to convincingly demonstrate an association between myeloid XCIP status and thrombotic risk. However, if the data from the four studies is combined (excluding the data from Harrison *et al*, 1999a), then there is more robust evidence that thrombotic risk is significantly higher in

patients with clonal myelopoiesis (38% for patients with a clonal XCIP versus 18% for patients with a polyclonal XCIP, P=0.008; Table 3.11).

Table 3.11	Summary of studies examining association between myeloid XCIP
	status and thrombotic rate

	Median	Ро	lyclonal XCIP		Clonal XCIP	Significant
Reference	follow- up, months	Total n	Thrombotic complications n (% of total)	Total <i>n</i>	Thrombotic complications n (% of total)	association between XCIP and thrombosis?
(El-Kassar <i>et</i> <i>al,</i> 1997)	45	14	3 (21%)*	19	6 (32%)*	No P=0.70
(Harrison <i>et</i> <i>al,</i> 1999a)	73	13	2 (15%)	10	6 (60%)	Yes P=0.04
(Chiusolo <i>et</i> <i>al,</i> 2001)	36	15	1 (7%)	17	7 (41%)	Yes P=0.04
(Shih <i>et al,</i> 2002)	45	15	1 (7%)	33	11 (33%)	No P=0.07
Reported in this thesis	151	29	8 (28%)	15	8 (53%)	No P=0.11
Overall †	84 [‡]	73	13 (18%)	84	32 (38%)	Yes P=0.008

Includes only the four studies which allow the criteria described in Section 1.3.5 to be applied and which included data on thrombotic complications. All P-values derived using Fisher's exact test

- * Includes both thrombotic and haemorrhagic complications (this study did not distinguish between them)
- ⁺ Data from Harrison *et al* (1999a) is excluded from the overall analysis since all the patients reported in this study are included in the cohort reported in this thesis
- # Weighted according to size of study

Of the 133 patients in the cohort, 55 (41%) were *JAK2* V617F-positive, with a median mutant level of 17%. Both these findings are in keeping with those reported by the majority of published studies, in which the overall prevalence of *JAK2* V617F is 53% and the median mutant level is 22% (Tables 1.7 and 3.12). At presentation, *JAK2* V617F-positive patients had

	Calle			ET		PV		MF	onley. d	enlev-d
Reference	analysed	Quantification method	۲	Mutant level %	r	Mutant level %*	c	Mutant level %	(ET vs. PV)	(ET vs. MF)
(Passamonti <i>et al</i> , 2006)	Neut	qRT-PCR	10	15 (2-35)	23	32 (2-90)	10	27 (12-93)	0.01	NR
(Moliterno <i>et al,</i> 2006)	Neut	AS-RTPCR	36	47 (10-63)	77	67 (35-100)	8	100 (48-100)	<0.001	NR
(Lippert <i>et al</i> , 2006)	Neut	qPCR	42	19 (3-50)	58	61 (8-98)			<0.001	
(Dupont <i>et al</i> , 2007)	Neut†	qRT-PCR	77	30 (2-85)	63	50 (10-100)			<0.001	
(Bellosillo <i>et al,</i> 2007)	Neut	AS RTPCR	43	10 ±7 [§]						
(Gale <i>et al,</i> 2007)	Neut	PCR/restriction digest	43	18 (5-100)						
(Larsen <i>et al,</i> 2007)	leuk	qPCR	21	7 (1-39)	06	23 (1-92)	13	67 (37-99)	0.001	<0.00001
(Pemmaraju <i>et al,</i> 2007) [‡]	Neut	AS-RTPCR	37	43 (11-64)						
(Kittur <i>et al,</i> 2007)	NR	AS qPCR	96	6 (1-100)						
(Theocharides <i>et al,</i> 2008)	Neut	R	15	23 ±15 [§]	33	63 ±25 [§]			<0.005	
(Vannucchi <i>et al,</i> 2008b)	Neut	qRT-PCR	382	26 (4-52)	297	48 (7-97)	168	46 (14-100)	<0.001	
(Girodon <i>et al,</i> 2008)	Neut	AS qPCR	127	18 ±12 [§]	94	54 ±24 [§]				

 Table 3.12
 JAK2 V617F mutant levels in the MPNs

(Hussein <i>et al,</i> 2009)	BM	Pyrosequencing	43	24 (5-40)	138	47 (NR)	78	44 (6-89)	<0.001	<0.01
(Pardanani <i>et al,</i> 2010)	BM	AS qPCR	118	7 (1-68)						
(Antonioli <i>et al,</i> 2010)	Neut	qRT-PCR	68	32 ±18 [§]	104	50 ±26 ^{\$}			<0.0001	
Overall			1158	22	1007	49	277	47		

Table 3.12 JAK2 V617F mutant levels in the MPNs – continued

Neut indicates Neutrophils, leuk indicates leukocytes, NR indicates not reported

- * Medians and ranges quoted unless otherwise stated
- ⁺ Also quantified V617F levels in total BM MNCs in ET & PV patients: results similar to those for Neut
- [‡] May include some of the patients reported by Moliterno *et al* (2006)
- [§] Mean ±SD

lower median platelet counts than *JAK2* WT ET patients (772x10⁹/L versus 950x10⁹/L, P=0.011), higher median haemoglobin concentrations (14.2g/dL versus 13.0g/dL, P=0.0002), and were older (53.0 versus 39.5 years, P<0.0001). These associations between *JAK2* mutant status and presenting characteristics have also been reported by most published studies (Table 3.4), and have led to the hypothesis that mutant-positive ET patients have a phenotype that lies somewhere between *JAK2* WT ET and PV (Campbell and Green 2006). Some studies have suggested that *JAK2* V617F-positive patients have higher presenting white cell counts than WT patients (Campbell *et al*, 2005; Kittur *et al*, 2007; Pardanani *et al*, 2010), but this was not observed in the cohort reported here (9.0 versus 10.0, P=0.52), nor was there any difference in the prevalence of splenomegaly.

In the cohort studied here, those patients who were JAK2 V617F-positive did not have a greater thrombotic rate than WT ones, whether at diagnosis (31% versus 19%, P=0.26), during follow-up (22% versus 11%, P=0.19), or overall (45% versus 26%, P=0.10). The published data are conflicting and suggest that any effects of JAK2 V617F on thrombotic risk are modest (Table 3.5) and this cohort may, therefore, have been too small to detect them. Nonetheless, it is noteworthy that patients who suffered a thrombosis had a higher mean mutant level than those who did not (11% versus 5%, P=0.02). In absolute terms, however, this difference (6%) is small and may not indicate a true biological difference between the two groups. It may, instead, reflect the presence of a few individuals with high mutant levels in the group of patients who had a thrombosis. According to a number of recent studies, patients with mutant levels greater than 50%, seen in 2%-6% of ET patients, have a significantly increased thrombotic risk compared to all other ET patients (De Stefano et al, 2009; Vannucchi et al, 2007), and two groups have observed the same phenomenon in patients with mutant levels above 25% (Antonioli et al, 2008; Carobbio et al, 2009). Of the 104 patients reported in this thesis for whom clinical follow-up data was available, the proportion of patients with a mutant level greater than 25% was significantly higher in the 37 patients who had a thrombosis than in

the 67 who did not (16% versus 5%, P<0.0001). It appears, therefore, that the thrombotic risk may be related to the presence of high mutant levels rather than simply the presence or absence of *JAK2* V617F, and this may account for the lower thrombotic rate in ET patients compared to PV patients, who more frequently harbour high mutant levels. One study has even suggested that at equivalent mutant levels the thrombotic risk is similar in the two disorders (Carobbio *et al,* 2009).

The mortality rate was significantly higher in the V617F-positive group, but since the majority of deaths were not related to haematological causes, this is likely to have been coincidence. No published studies have reported any similar association between *JAK2* mutational status and mortality, though Vannucchi *et al* (2007) did find that the 2% of ET patient who had a V617F mutant level greater than 50% did have an increased rate of myelofibrotic transformation.

MPL W515 mutations were detected in only three of the patients reported in this thesis, accounting for 2% of all patients and 4% of *JAK2* WT patients, figures similar to those of other studies (Table 1.8). None of them suffered thrombotic complications either prior to or following presentation, but the small patient numbers precluded meaningful interpretation of these findings. Of the three *MPL* W515-positive patients, one had a clonal neutrophil XCIP, one was male, and one was uninterpretable on the grounds of constitutive skewing.

No association between *JAK2* V617F and XCIP clonality was observed. This is surprising since, if V617F-positive cells are clonal, a higher rate of XCIP clonality might be expected in V617F-positive patients than in WT patients. In fact the opposite was true, although the difference was not significant: the proportion of patients with clonal myelopoiesis was 27% in the V617F-positive group compared to 45% in the *JAK2* WT one, P=0.19. These findings are in keeping with those of Levine *et al* (2006) who compared XCIP clonality and V617F mutant levels in PV, ET and MF. Whereas a correlation between clonality and mutant level was demonstrated in

PV, no such correlation was detectable in ET and MF. The apparent absence of correlation between mutant level and clonality in ET may relate the lower mutant levels usually reported in ET compared to PV (Table 3.12). The smaller mutant-positive population in ET would, therefore, be expected to exert a more modest effect on the overall myeloid clonality than in PV. Since the XCIP approach to clonality analysis has a limit of technical variation of 20%, it may not be sufficiently sensitive to detect such subtle effects.

An earlier study from our group has examined the relationship between the size of the mutant-positive population and T-cell XCIP in determining the neutrophil XCIP (Gale et al, 2007). Using the assumptions that mutant-positive cells are heterozygous for V617F and clonal, and therefore inactivate the same X-chromosome, this study calculated two possible resulting neutrophil XCIP values, depending on whether the clonal population was assumed to inactivate the 'A' or 'B' HUMAR alleles. In all 20 patients studied, one of the two calculated neutrophil XCIPs was within the limit of technical variation for the assay (20%) of the observed neutrophil XCIP. In the studies reported in this thesis, this analysis was performed for a further two patients: in one the closer of the two calculated neutrophil XCIPs was 14% different than the observed value, in the other it was 27% (Table 3.10). In the second individual, the observed neutrophil XCIP was 100%:0%, whereas the closer of the calculated values was 73%:27%, based on a T-cell XCIP of 69%:31% and the assumption that the mutant-positive population, which represented 14% of myeloid cells, inactivated the A-allele. In this patient the V617Fpositive population could not account for the difference between the T-cell and neutrophil XCIPs, implying that the clonal population was larger than mutant-positive one. One possible explanation for this is that the V617F-positive clonal population is derived from a larger, JAK2 WT one, in keeping with a pre-JAK2 event in this individual.

Overall, in 21 of the 22 patients, the neutrophil XCIP could be predicted from their mutant level and T-cell XCIP, within the technical limits of the assay. It therefore appears that in the majority of ET patients, the size of the mutant-positive population can account for the

difference between the T-cell and neutrophil XCIP, even if the mutant-positive population is too small to achieve a neutrophil XCIP which fulfils the criteria for clonality. The size of the mutant-positive population required to achieve a clonal neutrophil XCIP depends on the T-cell XCIP and the X-chromosome which is inactivated in the mutant-positive population. For instance, in a patient whose T-cell XCIP is 60%:40%, if the mutant-positive population inactivates the A allele, a mutant level of 25% is needed to generate a clonal neutrophil XCIP (80%:20%) based on the calculation described in Section 3.3.4. However, if it inactivates the B allele a level of 33% is required for a clonal neutrophil XCIP (20%:80%). In the cohort reported in this thesis, five patients with interpretable XCIPs had mutant levels of at least 25%, of whom four had evidence of clonal myelopoiesis. The fifth patient (Table 3.10, patient 2) in this group had a mutant level of 26%, yet her T-cell and neutrophil XCIPs were very similar (72%:28% and 73%:27% respectively).

It is notable that the majority of patients with a clonal XCIP studied in this thesis were *JAK2* WT (18 out of 24 patients, 75%). This would indicate that these 18 individuals harboured a myeloid clone, large enough to affect the overall neutrophil XCIP, which was due to a somatic mutation other than *JAK2* V617F. One of these harboured a *MPL* W515K mutation, but in the other 17 no mutations were detected.

In summary, the majority of the clinical and laboratory characteristics of the cohort of patients studied in this thesis were similar to those reported in most published studies, as were the observed differences between *JAK2* V617F-positive and *JAK2* WT patients with regard to haemoglobin level, platelet count and age. XCIP status appeared not to predict the incidence of vascular complications, though when the data was combined with that from other studies a significant association was observed. Whilst *JAK2* status was not associated with thrombotic risk, the *JAK2* mutant level was. There was no association between the presence of *JAK2* V617F and clonal myelopoiesis as determined by XCIP analysis. This observation may have been due

to the XCIP technique's lack of sensitivity to small clones, and conversely to the presence of V617F-negative patients with clonal myelopoiesis who are likely to harbour other mutation.

4 *JAK2* V617F-positive and *JAK2* WT thrombopoiesis in ET

4.1 Introduction

Although *JAK2* V617F is present in a large proportion of patients with the three major MPNs, the mutant level has consistently been reported to be lower in ET than in PV or MF (Table 3.12). A number of studies have suggested that, in ET, V617F -positive neutrophils represent only a minority of the total neutrophil population and that this proportion remains stable over many years (Antonioli *et al*, 2010; Besses *et al*, 2010; Gale *et al*, 2007; Girodon *et al*, 2008; Theocharides *et al*, 2008). This implies that *JAK2* V617F-positive and *JAK2* WT granulopoiesis stably co-exist in ET patients for prolonged periods, which differs from erythropoiesis in PV, where low EPO levels lead to suppression of *JAK2* WT erythroid cells and selection of EPO-hypersensitive V617F-positive erythroid cells (Dupont *et al*, 2007).

The stable co-existence of mutant-positive and *JAK2* WT granulopoiesis in ET is not typical of a neoplastic disorder and may simply be a feature of analysing neutrophils, rather than the more biologically-relevant platelets. The nature of this relationship between V617F-positive and *JAK2* thrombopoiesis is, therefore, unclear and will be investigated in this chapter.

4.1.1 The long-term stability of the V617F-positive population in ET

Three of the seminal studies which first reported on the occurrence of *JAK2* V617F in MPNs observed that transfection of Ba/F3 cells with the mutant *JAK2* gene resulted in these cells becoming hypersensitive to EPO and IL-3 (James *et al,* 2005; Kralovics *et al,* 2005; Levine *et al,*

2005). Kralovics *et al* (2005) went on to demonstrate that, in the presence of serum, V617Ftransfected cells showed increased survival and proliferation compared to *JAK2* WTtransfected ones. In view of this, the V617F-positive myeloid cells in patients with ET might be expected to outgrow the *JAK2* WT ones and eventually to represent the majority of the myeloid population, as they do in PV and Philadelphia-positive cells do in CML (Dubé *et al*, 1984aa; Dubé *et al*, 1984b). There is convincing evidence, however, that this is not the case.

A number of studies have examined the V617F mutant level in the three major MPNs (Table 3.12). Although the mutant level varies widely in individual ET patients, the overall median level for all the published studies is only 22% (range of medians 6%-47%), and significantly lower than that observed in either PV or MF (overall medians 49% and 47% respectively). In ET the percentage of myeloid cells carrying the mutation is likely to be double the mutant level, based on the data from Scott *et al* (2006). They genotyped haemopoietic colonies from 17 V617F-positive ET patients and found that in every case a mix of *JAK2* WT-homozygous and V617F-heterozygous colonies was present, but no patient harboured V617F-homozygous colonies. Although Dupont *et al* (2007) and Pardanani *et al* (07 2007) subsequently reported occasional V617F-homozygous haemopoietic colonies in small numbers of ET patients, it is likely that in most ET patients, the vast majority of V617F-positive cells are heterozygous for the mutation. Based on the overall median mutant level in ET of 22%, this implies that on average fewer than half of myeloid cells carry the mutation.

One possible explanation for the relatively low mutant level in ET is that the condition represents an early phase in the V617F-positive MPN spectrum and that PV and MF, in which the mutant levels are significantly higher, are later stages of the same disorder. It is possible that over time the mutant population outgrows the WT one leading to a change from ET to PV or MF. In support of this hypothesis, there is evidence from murine models that increasing the V617F-to-WT ratio changes the MPN phenotype. Tiedt *et al* (2008) generated a transgenic murine model in which expression of V617F in haemopoietic cells was controllable by Cre-lox

recombination and showed that when WT expression exceeded mutant expression, a thrombocytosis resulted whereas when mutant levels were greater than WT levels, polycythaemia was observed.

Mutant levels in ET, however, appear not to increase with time, according to a number of published studies. Gale et al (2007) sequentially analysed mutant levels in 19 V617F-positive ET patients over a median follow-up of 47 months. In 18 of these the mutant level remained stable during follow-up, within the limits of technical variation, including five patients who had not received cytoreductive therapy, suggesting that the stability of the mutant level was not due to treatment. Four subsequent studies, whilst primarily examining the effect of hydroxycarbamide therapy on mutant levels, have included untreated patients, with similar conclusions. Theocharides et al (2008) observed that in six untreated ET patients, mutant levels increased by a mean of 9% over a median follow-up of 26 months, which was considered to be within the limits of technical variation. Similarly, Girodon et al (2008) found no significant change in mutant levels in nine untreated ET patients during a median follow-up of 16 months, and Antonioli et al (2010) reported that in 24 untreated ET patients, followed for a median of 24 months, the median mutant level rose from 27% to 31%, which again was not significant. Finally, Besses et al (2010) observed a small but non-significant rise in mutant levels in 24 untreated ET patients over a median follow-up of 48 months, from a median of 26% to 29%. In summary, there is no evidence that mutant levels in ET do rise over time, even in the absence of cytoreductive therapy, or that the mutant-positive myeloid population outgrows the JAK2 WT one. Instead a steady state appears to exist, with no further advantage for the JAK2mutated cells, although the reason for this remains unclear.

4.1.2 JAK2 V617F mutant levels in neutrophils and platelets

A second possible explanation for the relatively low mutant levels in ET is that this is a feature of analysing neutrophils rather than platelets or MKs. The majority of studies which have examined the issue have quantified mutant levels in neutrophils since these are abundant in the peripheral blood and are therefore the easiest myeloid cells to analyse. In ET, however, the most striking abnormality is in megakaryopoiesis and thrombopoiesis and therefore the mutant levels in MKs and platelets are perhaps more relevant than those in neutrophils. Because peripheral blood platelets contain no detectable DNA, quantification of their *JAK2* mutant level requires RNA analysis, and this has been done relatively infrequently.

Two studies did find that, in ET, mutant levels were higher in platelet RNA than in neutrophil DNA, suggesting that there may be differences in *JAK2* mutant levels in different myeloid lineages (Bellosillo *et al*, 2007; Moliterno *et al*, 2006). It is likely, however, that some patients in both studies had received treatment with cytoreductive agents, which have been reported to reduce mutant levels (Girodon *et al*, 2008; Kiladjian *et al*, 2008; Larsen *et al*, 2009). Consequently it remains uncertain whether V617F-positive platelets represent the majority of the platelet population in mutant-positive ET and the low mutant levels reported by the studies in Table 3.12 are an artefact of studying neutrophils, or whether mutant levels in platelets and neutrophils are similar.

Determining whether, in ET, V617F-positive and *JAK2* WT thrombopoiesis genuinely co-exist in a stable, long-term equilibrium has profound implications for our understanding of ET, since neoplastic conditions are usually characterised by an expansion of the abnormal population at the expense of the normal one.

4.1.3 Aims

The studies reported in this chapter set out to quantify *JAK2* V617F levels in neutrophils and platelets from a cohort of 10 patients with untreated ET in order to establish whether there was a difference in mutant levels between the two lineages, and to investigate the relative sizes of the V617F-positive and the *JAK2* WT platelet populations.

4.2 Patients, materials and methods

4.2.1 JAK2 V617F mutant levels quantification in neutrophil and platelet RNA

Peripheral blood samples were collected from 10 ET patients prior to the initiation of cytoreductive therapy which might affect mutant levels. Each patient's platelet count at the time of collection was known. Neutrophils and platelets were purified as described in Section 2.1.3, RNA extracted as described in Section 2.1.4, and cDNA was generated using the SuperScript[™] III First-Strand Synthesis SuperMix for qRT-PCR kit as described in Section 2.2.5. BIOTAQ[™] PCR was then performed as described in Section 2.2.3.1 to amplify a 169bp fragment covering the mutation site in exon 14, using a 1 in 4 dilution of this cDNA as template. The primers used were *JAK2*ex14/(MM)F and an exon 15 fluorescently-labelled reverse primer (*JAK2*ex15/RF), both at a final concentration of 0.25μ M. Only 30 cycles of PCR were performed in order to minimise the formation of heteroduplex products, since they would not be cut by *Af*/III and would therefore mimic V617F-positive product. Denaturation, annealing and extension times were 60s in each cycle, with an annealing temperature of 61°C.

Following *AfI*III digestion (as described in Section 3.2.3.1) the uncut mutant allele was identified by fragment analysis as a peak at 169bp, whilst the WT allele was cut to 28+141bp,

though only the 141bp peak was visible as a peak since the 28bp fragment had no fluorescent label. As described in Section 3.2.4, the mutant level was calculated by expressing the area beneath the mutant peak as a percentage of the total area beneath both peaks. In every case, each sample was analysed three times and the mean value for all three runs calculated.

4.2.2 Calculation of the numbers of V617F-positive and WT platelets

The mutant levels in platelet RNA (%Mut) and the total platelet count (PLT_T) at the time of sampling were used to calculate the number of mutant-positive platelets (PLT_M), based on the assumption that all the latter platelets were heterozygous for the mutation (Scott *et al*, 2006), using the following equation:

$$PLT_M = PLT_T \times 2(\%Mut \div 100).$$

The difference between the total and the mutant-positive platelet count represents the WT platelet count.

4.3 Results

4.3.1 Mutant allele levels in neutrophil DNA and RNA and platelet RNA

The median V617F mutant level in samples from 10 ET patients obtained prior to cytoreductive therapy was 15% (range 11%-27%) in neutrophil gDNA and 21% (range 12%-31%) in neutrophil RNA (Figure 4.1). There was no significant difference in mutant levels between the two groups (P=0.07, paired t-test). In eight patients, the mutant level was higher in neutrophil RNA than in gDNA (range 2% to 17% higher), whereas in two patients the reverse was observed (2% and

10% higher in neutrophil gDNA than in RNA). In the patient whose mutant level was 10% higher in neutrophil gDNA than in RNA (27% versus 17%) there was no evidence of sample transposition, and the quantification assay was repeated with similar results. If this patient's values were excluded, the median mutant level was significantly higher in neutrophil RNA than in gDNA (21% versus 14%, P=0.01, paired t-test).



Figure 4.1 JAK2 V617F mutant levels in neutrophil DNA and RNA, and platelet RNA in 10 ET patients

The median mutant level in platelet RNA was 27% (range 20-39%), significantly higher than in neutrophil gDNA and neutrophil RNA (P=0.001 and P=0.002 respectively, paired t-test). The

mutant level was higher in platelet RNA than in neutrophil RNA in every patient (range 2% to 18% higher), and higher than in neutrophil gDNA in 9 of the 10 patients (range 6% lower to 22% higher). The one exception was the patient described in the paragraph above (mutant level in neutrophil gDNA 27% versus 21% in platelet RNA).

4.3.2 Calculation of WT and mutant platelet counts

Using the equation described in Section 4.2.3, the WT and mutant-positive platelet counts were calculated for the 10 ET patients investigated (Figure 4.2A). The median total, WT, and mutant-positive platelet counts were 745×10^9 /L (range $489 \cdot 1023 \times 10^9$ /L), 303×10^9 /L (range $218 \cdot 426 \times 10^9$ /L), and 410×10^9 /L (range $263 \cdot 798 \times 10^9$ /L) respectively. There was no difference between the median WT and mutant platelet counts (P=0.11, t-test; Figure 4.2B). In five patients the WT platelet count was higher than the mutant-positive platelet count, and in the other five it was lower.

In all 10 patients the WT platelet count was above the lower limit of normal for the laboratory $(150 \times 10^9/L)$ and in two patients it was above the upper limit of normal $(400 \times 10^9/L)$. There was no correlation between mutant level and total platelet count (r^2 =0.325, P=0.085; Figure 4.3A), between total platelet count and WT platelet count (r^2 =0.003, P=0.87, Figure 4.3B), or between mutant platelet count and WT platelet count (r^2 =0.30, P=0.10, Figure 4.3C).



В

Α



Figure 4.2 Calculation of *JAK2* wild-type and V617F-positive platelets in 10 ET patients

- A. The whole bar represents the total platelet count at the time of testing. The open bar indicates the WT platelet count, and the filled bar the V617F-positive platelet count. The figure above the bar is the percentage of JAK2 V617F-mutant positive platelets calculated from the mutant level in platelet RNA. The dashed horizontal lines indicate the normal range in our laboratory.
- B. Comparison between WT and mutant platelets for each patient. The dashed horizontal lines indicates the median value for each group.



Figure 4.3 Correlation between platelet counts and JAK2 V617F mutant level

4.4 Discussion

JAK2 mutant levels in V617F-positive ET patients are usually low, and on average fewer than half of their neutrophils carry the mutation. Moreover, mutant levels in these patients usually remain stable over many years. These observations are not keeping with the evidence that acquisition of *JAK2* V617F confers a proliferative advantage on affected cells but may simply be a consequence of analysing neutrophils, rather than the biologically more relevant platelets. The studies reported in this chapter, therefore, set out to investigate whether mutant levels in neutrophils and platelets differ in ET, and to examine the effect of the presence of a mutantpositive platelet population on the size of the *JAK2* WT one.

Mutant levels were quantified in neutrophil gDNA and RNA, and platelet RNA from 10 untreated ET patients. The results indicate that, whilst median V617F levels were similar in neutrophil gDNA and RNA (15% and 21% respectively, P=0.07), the median level in platelet RNA (27%) was higher than in both neutrophil gDNA and RNA (P=0.001 and P=0.002 respectively). Nonetheless, the difference between the median level in neutrophil RNA and platelet RNA of 6% is, in absolute terms, small – especially in view of the technical variation for the assay of 10% (Gale *et al*, 2007). Considering the patients individually, all 10 had mutant levels which were higher (by between 2% and 18%) in platelet RNA than in neutrophil RNA and, with one exception, than in neutrophil gDNA (by between 7% and 22%). The sample from the patient whose mutant level was higher in neutrophil gDNA than in both other samples was checked and re-analysed; whilst there was no evidence of sample transposition, this could not be verified absolutely. If this patient's results are excluded, mutant levels were significantly higher in neutrophil RNA than in neutrophil gDNA (21% versus 15%, P=0.01), but the other comparisons were unaffected.

These observations are in keeping with those of Moliterno *et al* (2006) who found that in 13 ET patients the median mutant level was significantly higher in platelet RNA than in neutrophil gDNA than in (42% versus 22%, P=0.001). The difference in median mutant level between the platelet RNA and neutrophil gDNA, 20%, was more marked than that observed in the cohort studied in this chapter, 12%, but since the number of patients receiving cytoreduction was not reported, it is difficult to compare the two studies. A larger study of 43 mutant-positive ET patients by Bellosillo *et al* (2007) also found a higher median mutant level in platelet RNA than in neutrophil RNA (13% versus 10%, P = 0.02). In this study approximately one-third of the ET patients were receiving cytoreductive therapy at the time of sampling, which might account

for the relatively low overall mutant levels compared to the majority of other studies (Table 3.12).

Whilst the data presented in this chapter does suggest that in ET the mutant level in platelets is higher than in neutrophils, the actual figure of 27% is still considerably lower than the median reported by most studies for neutrophils in PV or MF (Table 3.12). If V617F-positive cells are usually heterozygous for the mutation in ET, as Scott *et al* (2006) suggest, this implies that on average approximately half of the platelets are mutant-positive and half are *JAK2* WT. These data therefore appear to confirm the notion that in ET the V617F-positive populations (both neutrophils and platelets) do not rapidly outgrow their WT counterparts, but seem to coexist, typically representing 40-60% of the neutrophils and platelets in the peripheral blood.

To further explore the relationship between *JAK2* WT and mutant-positive platelets in ET, the number of WT and mutant-positive platelets at diagnosis were calculated for each of the 10 ET patients, based on platelet counts and simultaneous mutant levels quantified from samples taken prior to cytoreductive therapy. The overall median WT platelet count was similar to the median mutant platelet count $(303 \times 10^9/L \text{ versus } 410 \times 10^9/L$, P=0.11; Figure 4.2B), and in no patient was the WT platelet count below the lower limit of normal. In two patients the WT platelet count $(401 \times 10^9/L \text{ and } 426 \times 10^9/L)$ was marginally above the upper limit of normal for this laboratory $(400 \times 10^9/L)$ (Figure 4.2A). Moreover, there was no correlation between mutant level and platelet count (Figure 4.3A), nor any evidence that the WT platelet count fell as the total platelet count or mutant-positive platelet count increased (Figures 4.3B and 4.3C).

The fact that production of *JAK2* WT platelets was not suppressed in the presence of a thrombocytosis caused by mutant-positive platelets suggests that the regulation of thrombopoiesis generally (i.e. of both *JAK2* WT and V617F-positive platelets) is defective in ET. One possible mechanism for this relates to the role of c-Mpl in the normal homeostatic feedback pathway for thrombopoiesis (see Chapter 1). Thrombopoiesis is principally regulated

by plasma levels of the c-Mpl ligand, TPO, which, in the absence of an inflammatory stimulus, vary inversely with platelet count in haematologically normal individuals. Clearance of TPO is mediated by its internalisation and degradation by platelets and megakaryocytes via c-Mpl. Impaired c-Mpl expression in ET has been reported by several groups (Harrison *et al*, 1999b; Kralovics *et al*, 2003; Li *et al*, 2000) and this may account for the elevated TPO levels often seen in the disorder (Table 1.2). Moreover, *JAK2* V617F may play a role in the impaired expression of c-Mpl. Moliterno *et al* (2006) observed an inverse correlation between mutant levels and platelet c-Mpl expression in ET, suggesting that the abnormal *JAK2* molecule may interfere with expression of the receptor and therefore lead to impaired TPO clearance. They also noted that c-Mpl platelet expression was similarly impaired in *JAK2* WT ET patients, and concluded that in these cases a different molecular mechanism was responsible for the abnormal receptor expression.

It is, therefore, possible that impaired c-Mpl expression results in disruption of the normal thrombopoietic feedback system and is responsible for the lack of suppression of *JAK2* WT platelets in the presence of a V617F-positive population and thrombocytosis. As a consequence, despite the presence of a thrombocytosis, there is sufficient circulating TPO to stimulate the remaining WT megakaryocytes to produce WT platelets in normal numbers. Over time, even if the absolute number of V617F-positive MKs and platelets were to increase, their defective c-Mpl expression would render them unable to clear TPO in the normal way, and WT thrombopoiesis would therefore be unaffected.

Mutant levels in PV are markedly higher than in ET, and this may in part relate to the regulation of erythropoiesis which is mediated by EPO. EPO levels are controlled by a quite different mechanism than that of TPO: elevated haemoglobin levels result in suppression of renal EPO production, and thereby downregulate bone marrow erythroid activity. In PV, EPO levels are usually suppressed (Johansson *et al*, 2002; Mossuz *et al*, 2004), and correspondingly little WT erythropoiesis would be expected. *JAK2* V617F-positive erythroid progenitors are,

however, hypersensitive to EPO (Dupont *et al,* 2007) and would therefore be able to proliferate despite the low EPO levels. Defective TPO clearance might account for the persistence of WT thrombopoiesis at normal levels in V617F-positive ET; it does not, however, explain why mutant levels in neutrophil DNA, which is what most studies have measured, from ET patients are relatively low and remain so over prolonged follow-up.

An alternative hypothesis to explain the persistence of relatively low mutant levels in ET in both platelets and neutrophils is that the degree of expansion of the V617F-positive population depends on the molecular background on which the mutation arises, since this could affect the relative proliferative advantage which acquisition of the mutation confers. If *JAK2* V617F occurred in an individual whose myeloid progenitors already possessed an unusually high proliferative rate – whether due to hereditary or acquired factors – the V617Fpositive progenitor might possess a lesser growth advantage compared to *JAK2* WT progenitors than if it arose in a completely haematologically normal individual. The subsequent expansion of its progeny would be less marked and, since the *JAK2* WT myelopoiesis is more proliferative than normal it would not necessarily be suppressed by the V617F-positive myelopoiesis.

The possible nature of such a pre-existing 'hyper-proliferative' tendency remains largely speculative, but there is evidence for a pre-*JAK2* V617F event, at least a proportion of ET cases. Kralovics *et al* (2006) identified six ET patients whose neutrophil mutant level was less than 25% (i.e. fewer than 50% of granulocytes carried the mutation), who had evidence for monoclonal haemopoiesis by XCIP analysis or using a cytogenetic marker (del20q). This, they concluded, indicated that in a proportion of MPD patients *JAK2* V617F had arisen on a background of clonal haemopoiesis due to a prior somatic mutation. In keeping with this, one of the patients reported in Chapter 3 of this thesis was found to have a clonal population which was larger than the V617F-positive one.

Further evidence for a possible pre-*JAK2* event came from data from Campbell *et al* (2006a) and Theocharides *et al* (2007) who observed that in the majority of V617F-positive ET cases which transformed to AML, the leukaemic blasts were WT. They found no evidence for mitotic recombinations, deletions or point-mutations, which could account for the loss of V617F from a V617F-positive progenitor, and concluded that in some cases at least, it was more likely that the blasts had arisen from a pre-*JAK2* clonal population.

Alternatively, the speculative pre-*JAK2* abnormality might be inherited. Pardanani *et al* (2008) found that at three different *JAK2* SNPs, there was a strong association between one allele and PV, and the other allele and ET. The three SNPs were all in linkage disequilibrium with specific *JAK2* haplotypes, and it was suggested that the phenotype-determining allele may lie within these haplotypes. There is also evidence that the variation in haematological parameters in normal individuals may be determined by hereditary factors. In a study of nearly 14,000 individuals, Soranzo *et al* (2009) identified 22 SNPs which were associated with full blood count variability, including four which were associated with variation in platelet counts. Acquisition of *JAK2* V617F in an individual who harboured an inherited tendency to have a high-normal platelet count might be enough to result in a thrombocytosis, and simultaneously deprive the mutant-positive cells of the marked proliferative advantage required to dominate haemopoiesis.

In conclusion, the data presented here suggest that, whilst *JAK2* V617F levels are higher in platelets than neutrophils in ET, this difference is, in absolute terms, small. Further, it appears that in V617F-positive ET, WT thrombopoiesis is not suppressed by the presence of a V617F-positive population, and the mutant-positive progenitors do not rapidly outgrow their WT counterparts, contrary to what may be expected in a neoplastic disorder. This raises the question of whether ET is necessarily a clonal neoplasm, which is considered further in the following chapter.

5 Testing for more than one *JAK2* V617F-positive population in patients with mutant-positive ET

5.1 Introduction

In 2008, the WHO reclassified what had hitherto been called myeloproliferative disorders as myeloproliferative neoplasms (Swerdlow *et al*, 2008). This change in terminology served to emphasize the belief that these were neoplastic diseases which arose as a consequence of clonal expansion of haemopoietic progenitors (Tefferi and Vardiman 2008). Central to this view of the nature of MPNs was the perception that the acquisition of a number of somatic mutations, most commonly affecting *JAK2* exons 12 and 14 (V617F) and *MPL* exon 10, is critical to the pathogenesis of these neoplasms.

Parallels between the Philadelphia-negative MPNs and CML, and their respective constitutively-activated tyrosine kinases were soon drawn and the role of *JAK2* V617F in the Philadelphia-negative MPNs was equated to that of BCR-ABL in CML. ET and PV were likened to CML in chronic phase, MF to accelerated-phase CML, and the leukaemic transformation of all 3 Philadelphia-negative MPNs to the blast crisis phase of CML (Campbell and Green 2006). Conceptualising the Philadelphia-negative MPNs in this way was intellectually appealing since it allowed a rationalised classification of all 4 major MPNs. Moreover, it implied that the advances in the understanding of CML which followed the discovery of BCR-ABL might be rapidly emulated in the V617F-positive MPNs, facilitating the development of novel, targeted therapies. However, classifying ET as a clonal neoplasm akin to chronic-phase CML ignores marked differences in the natural history of the two disorders, and comparing *JAK2* V617F to

BCR-ABL as the pathogenic mutation overlooks several fundamental differences between the two molecular abnormalities.

5.1.1 Is V617F-positive ET a clonal neoplasm initiated by JAK2 V617F?

Four observations suggest that V617F-positive ET is not analagous to chronic-phase CML, and that *JAK2* V617F does not play the same pathogenic role in ET that BCR-ABL does in CML. First, ET does not clinically behave as a neoplastic disease. The mortality associated with ET is low and is usually due to a thrombotic event. In the absence of leukaemogenic medications the rate of malignant transformation is low – approximately 1-3% over 10 years – and the median survival is between 15 and 25 years (Passamonti *et al*, 2004; Passamonti *et al*, 2008; Wolanskyj *et al*, 2006). By contrast in untreated CML there is a 25% per annum risk of transformation to acute leukaemia (Champlin and Golde 1985) and a median survival of less than 5 years (Cervantes and Rozman 1982).

Second, in the small proportion of ET patients who do develop acute leukaemia, the malignant clone is often V617F-negative even when the preceding ET was V617F-positive (Campbell *et al*, 2006a; Theocharides *et al*, 2007), whereas in CML loss of Philadelphia-positivity is not observed following leukaemic transformation – indeed BCR-ABL expression usually increases (Elmaagacli *et al*, 2000; Gaiger *et al*, 1995). Campbell *et al* (2006a) studied four V617F-positive MPN patients who developed AML (two ET and two PV), and observed that in three cases the blast cells were *JAK2* WT, although the mechanism for this was unclear. If the blasts were derived from a V617F-positive progenitor, mitotic recombination might explain the loss of the V617F mutation. This was, however, considered unlikely for two reasons. Firstly, in one of the three patients reported, the pre-AML V617F-positive granulocytes carried cytogenetic abnormalities – del20q and t(1;9) – which were not detectable in the blast cells, suggesting
that the blast cells were not derived from the V617F-positive ones. Secondly, in two patients, the blast cells were shown by microsatellite analysis to be heterozygous for two chromosome 9p SNPs, located either side of the *JAK2* gene, suggesting that the blasts cells had not undergone mitotic recombination. Consequently, the authors speculated on the existence of a pre-*JAK2* clonal population from which both the V617F-positive MPN cells and the *JAK2* WT blasts had arisen. Theocharides *et al* (2007) also reported that the majority of V617F-positive MPN patients who developed AML had *JAK2* WT blasts, found in nine out of 14 interpretable cases. Again, microsatellite analysis did not suggest the occurrence of mitotic recombination, but interestingly in one patient the V617F-positive MPN granulocytes and the *JAK2* WT blasts shared a common cytogenetic abnormality (del11q), which was interpreted by the authors as in keeping with a pre-*JAK2* transformed population. By implication, therefore, *JAK2* V617F was not the initiating event in MPN, at least in these individuals.

Third, in V617F-positive ET patients, the mutant-positive population is usually small and remains stable over many years. The median mutant level observed in neutrophil DNA in our patient cohort (17%) was similar to that reported by most published studies (overall median 22%; Table 3.12), and implies that on average approximately 30-40% of myeloid cells carried the mutation (see Chapter 4). In the majority of ET patients, the V617F level remains stable for many years (Gale *et al*, 2007; Theocharides *et al*, 2008) and, as demonstrated in the previous chapter, *JAK2* WT thrombopoiesis is not suppressed by the presence of a mutant-positive population. This is very different to the situation in CML where the cytogenetically normal progenitors are suppressed, and the Philadelphia-positive progenitors expand rapidly to constitute more than 99% of dividing myeloid precursors (Dubé *et al*, 1984a; Dubé *et al*, 1984b).

Finally, most studies examining XCIPs in ET have reported that a proportion, on average 40%, of patients have polyclonal, or at least oligoclonal, myelopoiesis (Table 1.4). Even in *JAK2* V617F-positive ET patients, haemopoiesis is frequently poly- or oligoclonal and there is no

correlation between V617F-positivity and clonality (Antonioli *et al*, 2005; Levine *et al*, 2006). In our cohort, 72% of V617F-positive ET patients with an interpretable XCIP had a polyclonal pattern, compared to 55% of *JAK2* WT patients. Taken together, these four observations suggest that V617F-positive ET is not necessarily a clonal neoplasm in the mould of CML but is instead a more complex, heterogeneous disorder.

5.1.2 JAK2 V617F and clonality in ET

Of the observations presented above, the lack of association between the presence of *JAK2* V617F and myeloid clonality is perhaps the most surprising, since detection of a cell population carrying an acquired mutation is usually assumed to indicate the presence of a clonal disorder. One possible explanation for this observation, discussed in Chapter 3, is that in ET the mutant population is often too small to affect the overall myeloid clonality status. Nonetheless, an association between high mutant levels and myeloid clonality might still be expected, yet Levine *et al* (2006) demonstrated that in ET no such association exists, whereas in PV it does. It is also notable that in one patient discussed in Chapter 3 (Table 3.10, patient 2), T-cell and neutrophil XCIPs were nearly identical (72%:28% and 73%:27% respectively), despite a mutant level of 26%.

An alternative explanation is that, in some ET patients the V617F-positive population is not monoclonal and therefore does not necessarily result in myeloid clonality even in the presence of high mutant levels. This idea is not in keeping with conventional views of the role of somatic mutations in haematological disorders and implies that the V617F-positive population is derived from more than progenitor, each of which has acquired the mutation independently. Individual MPN patients have occasionally been reported to harbour more than one MPNassociated mutation. Pardanani *et al* (2006) studied 1182 MPN patients and identified six

individuals (0.5% of the cohort), including two with ET, who had both *JAK2* V617F and *MPL* W515 mutations. Of these six patients, two had *MPL* W515L and W515K mutations as well as *JAK2* V617F. Beer *et al* (2009; Jones *et al*, 2009; Olcaydu *et al*, 2009) described 13 MPN patients with two different genetic lesions, including five ET patients, four with *JAK2* V617F and *MPL* W515L, one with *JAK2* V617F and del20q. They used colony analysis to investigate the clonal relationship of the different mutant-positive populations and found that in seven of the 13 patients, including all five ET patients, some colonies carried one mutation, others carried the second mutation, but none was positive for both. They inferred from this that there were two or more distinct clones. They also noted that the frequency of patients with biclonal disease – seven out of approximately 1000 patients screened (0.7%) – was much greater than would be expected by chance and speculated about the existence of either an inherited or an environmental factor which predisposed to acquisition of these genetic abnormalities.

There is increasing evidence that some individuals have a hereditary predisposition to develop *JAK2* V617F-positive MPNs (discussed in detail in Chapter 6). Moreover, the data presented in Chapter 4 suggests that in most mutant-positive ET patients, *JAK2* WT myelo- and thrombopoiesis is not suppressed by the mutant-positive population, indicating the long-term persistence of WT myeloid progenitors. If these *JAK2* WT cells are particularly susceptible to acquisition of *JAK2* V617F, multiple mutant-positive clones might result. It is therefore feasible that the mechanism which predisposed these individuals to acquire *JAK2* V617F in the first place may impart a tendency to do so repeatedly.

The suggestion that in ET there may be several co-existing independent V617F-positive clones, rather than one dominant one, is contentious since it implies that the disorder is oligoclonal. The obstacle to examining this hypothesis is finding a means to identify multiple occurrences of the same mutation. To overcome this, two different approaches have been adopted. One technique uses an informative A/G SNP, rs2230724 located in exon 19 of the *JAK2* gene, to identify the *JAK2* allele (maternal or paternal) on which the mutation has arisen; the presence

of the mutation on both alleles in a single individual would be indicative of more than one mutational event. The other technique uses XCIP analysis to determine whether mutantpositive haemopoietic colonies from a single individual share a common clonal origin.

5.1.3 Aims

The experiments reported in this chapter aimed to investigate the hypothesis that in V617Fpositive ET, multiple mutant-positive independent clones exist.

5.2 Patients, materials and methods

5.2.1 Patient selection

The possibility that more than one V617F-positive population exists in mutant-positive ET patients was initially investigated using a *JAK2* SNP, rs2230724, and a PCR-restriction enzyme digestion-PCR approach to identify the allele on which the mutation had arisen. This involved generating a PCR product spanning both the *JAK2* V617F mutation site (exon 14) and the SNP locus (exon 19). Since exons 14 and 19 are too far apart to allow a PCR product spanning both the mutation and SNP loci to be amplified from gDNA (8.0Kb), cDNA was used (798bp product). For analysis using this technique, therefore, patients had to be V617F-positive and heterozygous for the exon 19 SNP, and have cDNA available.

5.2.2 Determining JAK2 status

5.2.2.1 Screening for JAK2 V617F using gDNA

PCR for *JAK2* exon 14 and restriction enzyme digestion with *Afl*III was used to screen DNA for the V617F mutation as described in Section 3.2.3.

5.2.2.2 Screening for JAK2 V617F using cDNA

To screen cDNA for the V617F mutation (see Section 5.2.4), a similar PCR and restriction enzyme digest technique to that described in Section 3.2.3 was used, except that a reverse primer which annealed to exon 15 was used (*JAK2*ex15/R; Figure 5.1A). A 169bp fragment spanning exon 14 was amplified using BIOTAQTM PCR with primers *JAK2*ex14/(MM)F and *JAK2*ex15/R. Thirty-five cycles of PCR were performed using an annealing temperature 61°C. PCR products were then digested with *AfI*III using 7µL PCR product in a reaction mixture containing 5 units *AfI*III enzyme, 1x manufacturer's buffer 3 and 100µg/mL BSA, made up to a 10µL total reaction volume with ddH₂0 and incubated for 6 hours at 37°C. *AfI*III cut the WT allele to 28 and 141bp whilst the mutant allele remained uncut (169bp). Digestion products were separated by agarose electrophoresis (3% gel) and visualised by UV transillumination.

5.2.3 JAK2 exon 19 SNP (rs2230724) genotyping

5.2.3.1 Exon 19 SNP genotyping using gDNA

Patient T-cell gDNA was genotyped for the JAK2 exon 19 A/G SNP by PCR and restriction enzyme digestion. T-cell, rather than neutrophil, gDNA was used to avoid interference by possible acquired loss of homozygosity of chromosome 9 in myeloid cells. A 214bp fragment covering JAK2 exon 19 and part of the flanking introns was amplified by PCR using BIOTAQTM

Α



Figure 5.1 Schematic representation of the technique used to determine the *JAK2* exon 19 SNP (rs2230724) genotype of V617F positive-mutant alleles

- A. Genomic structure of JAK2 gene showing primers used.
- B. PCR-restriction enzyme digestion-PCR technique used to determine the JAK2 exon 19 SNP genotype of JAK2 V617F mutant-positive alleles using cDNA

polymerase as described in Section 2.2.3.1, with primers *JAK2*ex19/F and *JAK2*ex19/R (Figure 5.1A), an annealing temperature of 60°C and 35 cycles of PCR, using 50ng of T-cell gDNA as template per reaction. Following this, a digestion mixture was prepared containing 5 units *Bst*NI enzyme, 1x manufacturer's buffer 2 and 100µg/mL BSA, made up to 5µL with ddH₂0, and to this 5µL PCR product was added and incubated for 2 hours at 60°C. Digestion products were separated by agarose electrophoresis (3% gel) then visualised by UV transillumination. *Bst*NI cut the G-allele to 93 and 121bp but did not cut the A-allele (214bp).

5.2.3.2 Exon 19 SNP genotyping using cDNA

To genotype cDNA for the exon 19 SNP (see Section 5.2.4), a 235bp fragment was amplified using BIOTAQ[™] PCR as described in Section 2.2.3.1, with exonic primers *JAK2*ex18/F and *JAK2*ex20/R. Thirty-five cycles of PCR were performed using an annealing temperature 62°C. PCR products were then digested with *Bst*NI using 7µL PCR product in a reaction mixture containing 5 units *Bst*NI enzyme, 1x manufacturer's buffer 2 and 100µg/mL BSA, made up to a 10µL total reaction volume with ddH₂0 and incubated for 2 hours at 60°C. *Bst*NI cut the Gallele to 108 and 127bp whilst the A-allele remained uncut (235bp).

5.2.4 *JAK2* exon 19 SNP (rs2230724) genotyping of V617F-positive alleles by PCRrestriction enzyme digestion-PCR

JAK2 V617F-positive patients who were heterozygous for SNP rs2230724, and for whom cDNA was available, were investigated for the presence of more than one V617F-positive population using a PCR-restriction enzyme-PCR technique (Figure 5.1B).

cDNA was generated from neutrophil RNA by reverse transcription using the SuperScript[™] III First-Strand Synthesis SuperMix for qRT-PCR kit as described in Section 2.2.5. A 798bp

fragment covering JAK2 exons 14-20 was then amplified from neutrophil cDNA using BIOTAQ[™] polymerase, as described in Section 2.2.3.1, with primers JAK2ex14/(MM)F and JAK2ex20/R, and 1µL cDNA diluted 1 in 4 in ddH₂O as template. Thirty-eight cycles of PCR, each of 60 seconds at 95°C, 60 seconds at 62°C, 90 seconds at 72°C were performed, followed by a final 5 minute extension at 72°C. Products were visualised on 2% agarose gel containing ethidium bromide.

Products were digested with *Bsa*AI to cut the WT but not the mutant allele. *AfI*III could not be used for this digestion due to the presence of an *AfI*III site common to both WT and mutant alleles which would destroy the ability to re-amplify the mutant products. For the digestion, a reaction mixture was made up containing 5 units *Bsa*AI enzyme and 1x manufacturer's buffer 3, made up to 5 μ L with ddH₂0, and to this 5 μ L PCR product was added. The mixture was incubated for 6 hours at 37°C.

This digest was then used as template for a second round of PCR. Undiluted digests were too concentrated to yield interpretable second-round PCR products and a range of dilutions of the digests in ddH₂0 was therefore made (1 in 50, 1 in 250, 1 in 1,000 and 1 in 10,000). For the second round of PCR, 1 μ L of each dilution was used as template. As in the first round PCR, a 798bp fragment spanning exons 14-20 was amplified using BIOTAQTM polymerase with primers *JAK2*ex14/(MM)F and *JAK2*ex20/R. Thirty-two cycles of PCR were performed, each of 60 seconds at 95°C, 60 seconds at 62°C, 90 seconds at 72°C, followed by a final 5 minute extension at 72°C.

The final stage of the process was to determine the *JAK2* V617F status and exon 19 SNP genotype of the resulting 798bp PCR product, by PCR and restriction enzyme digestion as described in Sections 5.2.2.2 and 5.2.3.2 respectively. For both PCRs, a 1 in 100 dilution of the second-round PCR product was used as template.

5.2.5 JAK2 exon 19 SNP genotyping of V617F-positive alleles by TA cloning

In the process described above, incomplete *Bsa*AI digestion of the WT first-round PCR products, even if minor, would result in some WT allele amplification in the second-round PCR. Consequently it is possible that the *JAK2* exon 19 SNP genotyping performed in the final step of the process might be from a mix of WT and mutant-positive alleles, and would therefore be misleading.

To overcome this, the exon 19 SNP genotype of V617F-positive alleles was studied by TA cloning using the Invitrogen TOPO TA Cloning® Kit (Figure 5.2). For the five patients studied, cDNA was generated from neutrophil RNA as described in Section 5.2.4; cDNA was also generated from platelet RNA for two of the five patients, and from RNA obtained from CD34positive cells for one patient. Using 1 in 4 dilutions of the cDNA as template, a 798bp fragment spanning JAK2 exons 14-20 was amplified by BIOTAQ[™] PCR, as described in Section 5.2.4. The entire 20µL PCR product obtained from each reaction was run on a 1.5% low-melting point agarose gel, visualised by UV transillumination and cut out from the gel. Gel-purification of the PCR products was necessary to remove the primers which interfered with subsequent reactions. The gel fragment containing the PCR product was melted by warming to 60°C. Depending on the strength of the original band as assessed visually, between 1 and 4μ L of the molten gel was used in the ligation step of the TA cloning process which was performed according to manufacturer's instructions (see Section 2.2.10). Bacterial clones were grown on LB-agar-carbenicillin plates coated with X-gal and transformed clones containing the JAK2 exon 14-20 PCR product were identified by their white or light-blue colour. These were picked and expanded by incubation in 50µL LB medium containing carbenicillin at 37°C overnight.



Figure 5.2 Schematic representation of the TA cloning technique used to determine the *JAK2* exon 19 SNP (rs2230724) genotype of *JAK2* V617F mutant-positive alleles using cDNA

The *JAK2* V617F status of individual clones was determined by PCR and restriction enzyme digestion (Section 5.2.2.2) using 1µL bacterial suspension in the 20µL PCR mix as template. Prior to the first cycle of PCR, the reaction mix was incubated for 2 minutes at 95°C to lyse the bacteria. V617F-positive clones were then genotyped for the *JAK2* exon 19 SNP by PCR and restriction enzyme digestion (Section 5.2.3.2), again using 1µL bacterial suspension in the 20µL PCR mix as template and a 2 minute 95°C bacterial lysis step prior to the first PCR cycle.

5.2.6 Using a JAK2 intron 14 SNP (rs12343687) to genotype V617F-positive alleles

Whilst this work was underway, data on several previously unrecognised *JAK2* SNPs was published (Jones *et al*, 2009; Kilpivaara *et al*, 2009; Olcaydu *et al*, 2009), including one in intron

14 which was sufficiently close to the V617F locus to allow a 482bp PCR product spanning both sites to be amplified from gDNA. The patients who had been studied using the exon 19 SNP were therefore investigated using the intron 14 SNP to establish whether RNA and gDNA-based approaches gave concordant results. The process described in Sections 5.2.4 and 5.2.5 was therefore repeated but here the *JAK2* intron 14 SNP, rs12343687 (C/T), was studied instead of the *JAK2* exon 19 SNP.

5.2.6.1 Genotyping patients for intron 14 rs12343687 SNP status

In order to identify which of the patients studied in Sections 5.2.4 and 5.2.5 were heterozygous for the intron 14 SNP, a 164bp fragment spanning this SNP was amplified using primers *JAK2*ex14/F2 and *JAK2*ex14/(MM)R (Figure 5.3A). The reverse primer introduced a mismatch immediately downstream of the SNP to create a recognition site so that *Bsa*AI would cut products with the C allele whilst leaving the T-allele products uncut. T-cell gDNA was used as template and 35 cycles of BIOTAQTM PCR were performed as described in Section 2.2.3.1, with an annealing temperature of 59°C. Products were run on a 2% agarose gel and visualised by UV transillumination. Digestion with *Bsa*AI was then performed: a reaction mix was made up containing 4 units *Bsa*AI enzyme and 1x manufacturer's buffer 3, made up to a 4µL ddH₂0, and to this 4µL PCR product was added and the mixture incubated for 4 hours at 37°C. Products were run on a 3% agarose gel and visualised by UV transillumination. *Bsa*AI cut the C allele to 31 and 133bp whilst the T allele remained uncut (164bp).



Figure 5.3 Schematic representation of *JAK2* exons 14 and 15 showing primers used to genotype V617F-positive alleles using intron 14 SNP (rs12343687)

- A. Genomic structure of JAK2 exons 14 to 15 showing primers used.
- B. PCR-restriction enzyme digestion-PCR technique used to determine the *JAK2* intron 14 SNP genotype of *JAK2* V617F mutant-positive alleles using gDNA

5.2.6.2 JAK2 intron 14 SNP genotyping of V617F-positive alleles by PCR-restriction enzyme digest-PCR

Neutrophil gDNA from patients who were heterozygous for rs12343687 was then studied using a similar PCR-restriction enzyme digest-PCR technique to that described in Section 5.2.4 except that a 482bp fragment was amplified covering the *JAK2* exon 14 mutation site and the rs12343687 SNP in intron 14 (Figure 5.3B). A first round PCR was performed using GoTaq[®] DNA polymerase, primers *JAK2*ex14/(MM)F and *JAK2*ex14/(MM)R, 35 cycles of PCR at an annealing temperature of 59°C. PCR products were digested using *Af*/III as described in Section 3.2.3 which cut the WT allele so that only mutant-positive PCR products were re-amplified in a second round of PCR. Conditions for the second round PCR were identical to those used in the first-round PCR but with 1µL of the digest diluted 1 in 20 as template. The *JAK2* V617F status of the second-round PCR products was then determined using the PCR-restriction digestion technique described in Section 5.2.2.1, followed by determination of their rs12343687 status as described in Section 5.2.6.1.

5.2.6.3 JAK2 intron 14 genotyping of V617F-positive alleles by TA cloning

TA cloning was performed in three patients, using the same approach to that described in Section 5.2.5 except that competent bacteria were transformed with the 482bp PCR product which spanned the *JAK2* exon 14 mutation site and rs12343687 in intron 14. This was amplified from neutrophil gDNA using primers *JAK2*ex14/(MM)F and *JAK2*ex14/(MM)R as described in Section 5.2.6.2. V617F-positive clones were identified by PCR and restriction enzyme digestion as described in Section 5.2.2.1 and the rs12343687 allele present in these clones was determined as described in Section 5.2.6.1. For both PCRs, 1µL bacterial suspension was added to the 20µL PCR mix as template and prior to the first cycle of PCR, the reaction mix was incubated for 2 minutes at 95°C to lyse the bacteria.

5.2.7 X-chromosome inactivation analysis of erythroid burst-forming units (BFU-Es)

To establish whether V617F-positive haemopoietic progenitors shared a common clonal origin, BFU-Es were cultured from mononuclear cells (MNCs) from patients with mutant-positive ET. *JAK2* V617F-positive BFU-Es were then identified and analysed to determine their inactivated X-chromosome.

5.2.7.1 Culturing BFU-Es from peripheral blood

To minimise contamination all processing except centrifugation was performed in a laminar air flow cabinet. Approximately 30mL peripheral blood was collected in EDTA and centrifuged for 15 minutes at 200g at room temperature and the platelet-rich plasma was removed. The buffy coat of the remaining sample was then harvested and MNCs purified by Ficoll-Hypaque density centrifugation as described in Section 2.1.3. They were resuspended in 15mL RPMI containing 10% FCS (RF10%) and spun for 5 minutes at 800g at room temperature. The supernatant was removed, the pellet resuspended in 5mL RF10% and the resulting MNC suspension was counted using a Sysmex KX21N whole blood cell analyser.

MACS HSC-CFU media complete with Epo (Miltenyi Biotec, Bisley, UK) was thawed at 37°C and divided into 2.5mL aliquots. To each aliquot, an appropriate volume of the MNC suspension was added to generate the following dilutions: 0.5×10^5 cells/mL, 1.0×10^5 cells/mL, 2.0×10^5 cells/mL and 4.0×10^5 cells/mL. At least four aliquots at each concentration were made up. Each aliquot was vortexed and gently layered into a single well of a 6 well tissue-culture plate using a 5mL syringe and kwill, ensuring no air bubbles were introduced into the media. Approximately 3mL sterile PBS was added to the spaces between the wells, to prevent dehydration of the media, and each plate was incubated at 37°C in a humidified atmosphere containing 4% CO₂.

After 14 days, each well was inspected under a dissecting microscope and BFU-Es identified. Single, discrete BFU-Es were harvested into 200µL PBS using a sterile fine-tip pastette and centrifuged at 2000g for 5 minutes at room temperature. The supernatant was removed and to each pellet 43µL TNE lysis buffer (10mM Tris-HCl, 400mM NaCl, 2mM EDTA) was added. Samples were incubated on ice for 2 minutes, and 5µL 10% SDS and 1µL 20mg/mL proteinase K added. Samples were then vortexed, incubated at 56°C for one hour and placed at -20°C for at least 12 hours. After thawing, DNA was extracted by the addition of 1µL glycogen, 5µL 3M sodium acetate and 150µL absolute ethanol, followed by centrifugation at 14,000g for 15 minutes at room temperature. The supernatant was removed, 200µL 75% ethanol added, and the sample spun at 14,000g for 5 minutes at room temperature. The supernatant was again removed, the DNA pellet allowed to air dry and resuspended in 15µL ddH₂0.

5.2.7.2 Identification of V617F-positive colonies

JAK2 V617F-positive colonies were identified by amplification refractory mutation system (ARMS) PCR technique using the methodology of Jones *et al* (2005; Figure 5.4). This single-step technique allowed more rapid analysis of large numbers of colonies than the PCR and restriction enzyme digestion described in Section 5.2.2. Since mutant-positive colonies were expected to be heterozygous for V617F, with a mutant level of 50%, a high-sensitivity technique was not required. For this PCR, four primers were used in order to generate three possible products – a 463bp common (control) product, a 229bp WT product, and a 279bp mutant product. Two common intronic primers were used, both at 1µM: a forward primer (*JAK2*ex14/ARMS/FO) located in intron 13, upstream of the mutation site, and a reverse primer (*JAK2*ex14/ARMS/RO) located in intron 14, downstream of the mutation site. These two primers generated the 463bp common product regardless of the presence of V617F. In addition, a WT-specific forward primer (*JAK2*ex14/ARMS/FW) was used at 0.5µM, which only annealed to the WT *JAK2* sequence and together with *JAK2*ex14/ARMS/RO generated the

229bp WT product. Finally, a V617F-specific reverse primer (JAK2ex14/ARMS/RM) was used at 0.5μM, which, in the presence of V617F and JAK2ex14/ARMS/FO, generated the 279bp mutant product. GoTaq[®] PCR was performed as described in Section 2.2.3.2 except for the use of the 4 primers at the specified concentrations, with 1μL colony DNA as template. Thirty-two cycles of PCR were performed, with an annealing temperature of 60°C, and products electrophoresed and visualised as before.

Α.



Figure 5.4 JAK2 exon 14 ARMS PCR to detect V617F (based on Jones et al, 2005)

- A. Schema for PCR showing *JAK2* exons 14 and 15, the four primers used, and the three possible products
- B. Agarose gel (2%) showing the three possible products in four V617F-positive ET patients with known mutant levels (indicated at the bottom of the gel), and one JAK2 WT control ('WT'). 'Neg' indicates negative control and 'ladder' indicates Bioline Hyperladder 4.

5.2.7.3 Inactivated X-chromosome analysis of V617F-positive colonies

The inactivated X-chromosome of V617F-positive colonies was determined using a modified HUMARA technique (Section 3.2.5). A digestion mixture containing 12.5 units *Hpa*II, 2.5 units *Rsa*I and 1x manufacturer's buffer 1 was prepared and made up to 2.5 μ L with ddH₂O. To this mix 2.5 μ L colony DNA was added and the mixture incubated overnight at 37°C. A PCR product spanning the VNTR sequence in the first exon of the *HUMAR* gene was then amplified. A reaction mix containing 1x manufacturer's flexibuffer, 0.2mM each dNTP, 1ng/ μ L each HUMARA primer and 1 unit GoTaq[®] Hot Start polymerase was prepared, made up to 15 μ L with ddH₂O, and added to the 5 μ L digestion mix. Twenty-five cycles of PCR were performed, using an annealing temperature of 68°C, and a final extension of 72°C for 15 minutes. Cycling conditions were as described in Section 2.2.3.2.

PCR products were then analysed by fragment analysis, as described in Section 2.2.7, in order to determine the inactivated *HUMAR* allele. Since all cells in each colony were descended from a single cell, all would be expected to inactivate the same *HUMAR* allele. Consequently, only a single primary peak was expected on fragment analysis. The presence of more than one primary peak implied the presence of either two separate colonies or incomplete enzyme digestion, and such results were repeated and excluded if persistent.

5.3 Results

5.3.1 JAK2 exon 19 SNP (rs2230724) genotyping in haematologically normal controls and ET patients

A cohort of 114 haematologically normal controls, 111 ET patients and 26 PV patients was screened to determine the rs2230724 genotype (AA, AG or GG). There was no difference in the

distribution of SNP genotypes in ET patients, regardless of mutational status, or in PV patients compared to normal controls (Table 5.1). However, when compared to the expected genotype frequency based on HapMap European population data, there was a significant excess of Aalleles in ET patients overall, V617F-positive ET patients, and PV patients (P<0.001, P= 0.03 and P=0.04 respectively, Chi-square).

5.3.2 JAK2 exon 19 SNP genotyping of V617F-positive alleles using SNP rs2230724

Twenty-three V617F-positive ET patients were heterozygous for rs2230724, and of these 11 had cDNA available. Samples from these 11 patients were therefore screened for more than one V617F-positive population, initially using the PCR-restriction enzyme digestion-PCR technique. PCR products were visualised by agarose gel electrophoresis. Inspection of the gel revealed that adequate first-round and second-round PCR products were amplified in 10 of the 11 patient samples studied; in the remaining patient (Table 5.2, patient E10) both PCR products were too weak to allow accurate genotyping for V617F and the exon 19 SNP.

In the 10 samples where good PCR products were obtained, re-amplification of only mutantpositive and not WT alleles in the second-round PCR was confirmed by exon 14 PCR and *Afl*III digestion (Figure 5.5A). SNP genotyping of the second round PCR products from nine of the 10 patients showed a mixture of A and G alleles (Figure 5.5B) with only one patient (patient E1, lane 3) showing only A alleles. This patient had a relative mutant level of 100%, consistent with the presence of chromosome 9pLOH. The relative A:G allele proportions, as determined by visual estimation of the two bands (Figure 5.5B), varied in the other nine patients (Table 5.2). In six patients, the mutation arose principally on the A-allele (Table 5.2, patients E2,E3,E6, E8, E9 and E10), in three it arose principally on the G-allele (patients E4, E7 and E11), and in one patient the distribution between the two alleles was approximately equal (patient E5). This

suggested that *JAK2* mutations were present on both chromosomes in most of the patients studied.

To ensure that these results were not due to incomplete *Bsa*AI digestion of the WT PCR product, which would mimic the mutant product, full-length PCR products covering the V617F mutation and the SNP (exons 14-20) were TA cloned using neutrophil cDNA from five of these patients. V617F-positive clones were identified and the exon 19 SNP present in these clones determined. Between 29 and 33 V617F mutant-positive clones were genotyped from each sample. In each case, both allele A and allele G clones were detected (Table 5.2, patients E7-E11). In three cases, the majority of clones had allele A: 82%, 76% and 77% of clones respectively in patients E8-E10. In one case allele A and G clones were approximately equal (patient E7, 41% A clones), and in the remaining case there were more allele G clones (patient E11, 23% A clones). In all these individuals, the results from the TA cloning approach mirrored those from the PCR-restriction enzyme digest-PCR approach (Table 5.2).

Platelet samples from two patients gave comparable results to the neutrophil samples: 82% and 78% allele A clones respectively in neutrophils and platelets from patient E8; 77% and 72% from patient E9. Neutrophil samples taken from the same patient 23 months apart (patient E9) were analysed and gave comparable results with 77% and 87% allele A clones respectively. RNA from a purified bone marrow CD34+ sample was also available from this patient and yielded similar results to the neutrophil sample, 78% allele A clones.

Of the 26 V617F-positive PV patients studied, ten were heterozygous at the exon 19 SNP and cDNA was available for five of these. Neutrophil cDNA samples from these five patients were screened using the PCR-restriction enzyme digestion-PCR technique for the presence of more than one mutational event. In all five cases, the V617F-positive alleles were associated with a single exon 19 SNP allele (allele A in every case; Figure 5.6).

Incidence of JAK2 exon 19 SNP rs2230724 genotypes in normal controls, ET and PV patients Table 5.1

			IAL	K2 exon 19 5	SNP genoty	эс		Observed versus	Observed versus
Diagnosis	и	Ō	bserved n (%	(9	_	Expected n*		normal control	expected*
		AA	AG	99	AA	AG	GG	(P-value) †	(P-value)†
Normal controls	114	33 (29%)	57 (50%)	24 (21%)	23	56	34	1	0.17
ET All patients:	111	39 (35%)	55 (50%)	17 (15%)	22	55	34	0.43	<0.001
V617F positive:	43	16 (37%)	23 (53%)	4 (9%)	6	21	13	0.20	0.03
JAK2 WT:	68	23 (34%)	32 (47%)	13 (19%)	14	34	21	0.79	0.13
PV‡	26	13 (50%)	10 (38%)	3 (12%)	Ŋ	13	8	0.11	0.04

* expected frequencies of AA = 20%, AG = 49%, GG = 30%, based on HapMap allele frequency of SNP rs2230724 (A:G = 0.448:0.552) in a European population (http://www.ncbi.nlm.nih.gov/projects/SNP/snp_ref.cgi?rs=2230724)

t Chi-square P-values shown

all 26 PV patients were JAK2 V617F-positive

JAK2 exon 19 and intron 14 SNP analysis of PCR products from 11 patients with JAK2 V617F mutant-positive ET Table 5.2

		JAK2 exon 19 S	SNP (rs2230724)	JAK2 intron 14 S	NP (rs12343687)
Patient	% Mutant JAK2 (Neutrophil DNA)	SNP allele A:G ratio in V617F PCR products* (Neutrophil RNA)	% of SNP allele A V617F+ clones† (No. clones examined)	SNP allele C:T ratio in V617F PCR products* (Neutrophil DNA)	% of SNP allele C V617F+ clones in neutrophil DNA (No. clones examined)
E1	100	А	I	Not informative	ı
E2	13	A>G	I	Not informative	I
E3	40	A>G	I	C>T	87% (16)
E4	25	A <g< td=""><td>I</td><td>Not informative</td><td>I</td></g<>	I	Not informative	I
E5	24	A=G	I	C=T	I
93	42	A>G	I	C=T	I
E7	26	A <g< td=""><td>41% (31)</td><td>C<t< td=""><td>14% (22)</td></t<></td></g<>	41% (31)	C <t< td=""><td>14% (22)</td></t<>	14% (22)
E8a (Oct 2000)	20	A>G	82% (33)	Not informative	
E8b (Dec 2004)	33	A>G	Platelets 78% (27)		
E9a (Jul 2000)	27	A>>G	77% (33)		
			Platelets 72% (29)		
E9b (Jun 2002)	20	A>>G	87% (31)	C=T	100% (14)
			CD34+ 78% (32)		
E10	14	ЬР	76% (29)	Not informative	ı
E11	11	A <g< td=""><td>23% (31)</td><td>C<t< td=""><td>I</td></t<></td></g<>	23% (31)	C <t< td=""><td>I</td></t<>	I
visual estimates from					

עוצעאו פאווווא אויוו אַכו

*

t neutrophil RNA unless otherwise indicated

PP indicates poor product unsuitable for analysis; dash (-) indicates not done; not informative indicates patient's gDNA was homozygous at this SNP

A. JAK2 V617F mutant status



B. Exon 19 rs2230724 SNP status



Figure 5.5 *JAK2* exon 19 SNP genotype analysis of V617F mutant alleles in 11 ET patients

- A. Gel showing *AfI*III restriction enzyme digest of 14/(MM)F + 15/R PCR products amplified from the 798bp second-round PCR product. WT products were cut (28+141bp), V617F mutant-positive products remained undigested (169bp).
- B. Gel showing *Bst*NI restriction enzyme digest of 18/F + 20/R PCR products amplified from the 798bp second-round PCR product. Exon 19 SNP allele G products were cut (108+127bp), but allele A products remained undigested (235bp).

Un=no restriction enzyme added. WTc = wild-type control. Gc=allele G control. Breaks within a single panel indicate spliced lanes from a single run



Figure 5.6 JAK2 exon 19 SNP genotype analysis of V617F mutant alleles in 5 PV patients

- A. Gel showing AfIII restriction enzyme digest of 14/(MM)F + 15/R PCR products amplified from the 798bp second-round PCR product. WT products were cut (28+141bp), V617F mutant-positive products remained undigested (169bp).
- B. Gel showing *Bst*NI restriction enzyme digest of 18/F + 20/R PCR products amplified from the 798bp second-round PCR product. Exon 19 SNP allele G products were cut (108+127bp), but allele A products remained undigested (235bp).

Un=no restriction enzyme added. WTc = wild-type control. Gc=allele G control.

5.3.3 JAK2 intron 14 SNP genotyping of V617F-positive alleles using SNP rs12343687

DNA samples from the 11 ET patients investigated in Section 5.3.2 were genotyped for the intron 14 SNP (rs12343687). Six were heterozygous for the intron 14 SNP and could therefore be assessed for the presence of more than one V617F-positive population by PCR-restriction enzyme digestion-PCR or TA cloning. Using the former technique, in all six cases the second-round PCR amplified only V617F-positive PCR products (Figure 5.7A) and these mutant-positive products contained a mix of C and T alleles at SNP rs12343687 (Figure 5.7B). In one patient (E3), the mutation arose principally on the C allele, in two patients (E7 and E11) it arose principally on the T allele, and in three patients (E5, E6 and E9) the distribution between the two alleles was approximately equal.

For three patients, the 482bp PCR product spanning the mutation site and the intron 14 SNP was TA cloned. In two of these (Table 5.2, patients E3 and E7), the results from these confirmed the results from the PCR-digest-PCR technique – i.e. mutant-positive clones were a mix of C and T alleles at the intron 14 SNP. For patient E3, both the PCR-restriction enzyme digest-PCR and the TA cloning approach indicated that *JAK2* V617F had arisen principally on the C allele, and in patient E7 both techniques indicated that the mutation arose principally on the T allele. However, in one (patient E9) only one intron 14 SNP allele (C) was associated with V617F-positive clones even though the PCR-digest-PCR technique suggested that the mutation arose on both the C and T alleles (Figure 5.7B).



Figure 5.7 JAK2 intron 14 SNP (rs12343687) genotype analysis of V617F mutant alleles in 6 ET patients

- A. Gel showing Af/III restriction enzyme digest of 14/(MM)F + 14/R PCR products amplified from the 482bp second-round PCR product. WT products were cut (28+129bp), V617F mutant-positive products remained undigested (157bp).
- B. Gel showing *Bsa*AI restriction enzyme digest of 14/F2 + 14/(MM)R products amplified from the 782bp second-round PCR product. Intron 14 SNP allele C products were cut (31+133bp), but allele T products remained undigested (164bp).

Un=no restriction enzyme added. WTc = wild-type control. Cc=allele C control. Breaks within a single panel indicate spliced lanes from a single run

These results were then compared with those obtained using the exon 19 SNP (rs2230724) for the six patients for whom data for both SNPs was available. In two patients (E7 and E11) in whom the mutation arose principally on the T allele at the intron 14 SNP, it arose principally on the G allele at the exon 19 SNP. Conversely, in the single patient (E3) in whom the mutation arose principally on the C allele of the intron 14 SNP, it arose principally on the A allele at the exon 19 SNP, and in one patient (E5) the mutation was equally distributed between the two alleles at both SNPs. In two further patients (E6 and E9), however, the mutation was distributed equally between both intron 14 SNP alleles but arose preferentially on the A allele at the exon 19 SNP.

5.3.4 Genotyping of peripheral blood erythroid burst-forming units (BFU-Es)

Freshly-collected peripheral blood from female V617F-positive ET patients was required for this assay. Of the eighteen appropriate patients under active follow-up, six were available to provide samples and erythroid colonies from these six individuals were cultured and analysed (Table 5.3). The mutant level in these patients varied between 5% and 35%, implying that approximately 10% to 70% of the myeloid cells were mutant-positive, assuming that all mutant-positive cells were heterozygous for the mutation (Scott *et al*, 2006). Two of the six patients (E8 and E12) had polyclonal neutrophil XCIPs, two had clonal ones (E13 and E14), and two had uninterpretable ones – one due to age (E6) and one due to constitutional skewing (E7).

For each of the six patients studied, between 53 and 192 colonies were harvested and underwent *JAK2* V617F mutation screening. There was a significant correlation between the V617F mutant level in neutrophil DNA and proportion of mutant-positive colonies for the six patient samples (P=0.04, r²=0.71). Patients E6 and E7 had the highest mutant level, in keeping with 70% and 46% of their cells being mutant-positive respectively, and the greatest proportion of their colonies were V617F-positive, 98% and 35% respectively (Table 5.3). Patient E13 had the lowest mutant level, equivalent to 10% mutant-positive cells, and the lowest proportion of mutant-positive colonies, 4%. The other three patients had intermediate

values for both the mutant levels and the proportion of erythroid colonies which were mutantpositive.

Patient	% <i>JAK2</i> V617F (neuts)	% V617F+ cells*	XCIP CD3+	XCIP neuts	Total colonies harvested n	V617F+ colonies n (% of total)	Colonies with HUMARA result n	A-allele inactivated n (% of total)	B-allele inactivated n (% of total)
E6	35%	70%	46% : 54%	20% : 80%	53	52 (98%)	17	0 (0%)	17 (100%)
E7	23%	46%	83% : 17%	82% : 18%	184	64 (35%)	37	37 (100%)	0 (0%)
E8	13%	26%	38% : 62%	45% : 55%	126	16 (13%)	15	0 (0%)	15 (100%)
E12	16%	32%	31% : 69%	31% : 69%	58	14 (24%)	13	5 (38%)	8 (62%)
E13	5%	10%	64% : 36%	82% : 18%	96	4 (4%)	3	3 (100%)	0 (0%)
E14	10%	20%	40% : 60%	18% : 82%	192	48 (25%)	23	0 (0%)	23 (100%)

Table 5.3 Summary of XCIP analysis of V617F-positive BFU-Es from 6 ET patients

* Assumes V617F-positive cells are heterozygous for the mutation

The inactivated *HUMAR* allele for every V617F-positive colony was analysed. In a proportion of colonies it could not be established due to the presence of more than one peak on fragment analysis, indicating contamination by another colony. For this reason the number or colonies with a HUMARA result was generally smaller than the number of V617F-positive colonies identified (Table 5.3). In five of the six patients, all the mutant-positive colonies from a single patient inactivated the same *HUMAR* allele – the A allele in two patients (E7 and E13), and the B allele in three patients (E6, E8 and E14). In one patient (E12), however, five (38%) of the 13 V617F-positive colonies inactivated the A allele, and eight (62%) inactivated the B allele. The neutrophil and T-cell XCIPs in this patient were both 31%:69%.

Three of these six patients (E6-E8) had been investigated using TA cloning and found to have more than one V617F-positive population (Section 5.3.2). All three of these patients appeared to have the same inactivated X-chromosome in V617F-positive erythroid colonies.

5.3.5 Calculation of neutrophil XCIP based on T-cell and BFU-E XCIPs and mutant levels

The calculation described in Section 3.3.4 uses the T-cell XCIP and V617F mutant levels to generate two different neutrophil XCIPs, depending on which X-chromosome is inactivated in the V617F-positive population. The data presented in Table 5.3 allows the inactivated X-chromosome of the mutant-positive populations to be deduced from the XCIP of the BFU-Es, and the calculation can therefore be refined as follows:

Predicted neutrophil XCIP (A%:B%) = $[(y/100 \times A_{C}) + (z/100 \times \%A_{T})] : [(z/100 \times \%B_{T}) + (y/100 \times B_{C})]$

where y is the percentage mutant-positive cells, z is the percentage WT cells, the T-cell XCIP is A_T %: B_T %, and the ratio of inactivated A to B alleles in the V617F-positive colonies is A_c : B_c .

This calculation generates only a single predicted neutrophil XCIP, which was compared to the observed value (Table 5.4). For the six cases studied the median difference between the predicted and the observed neutrophil XCIP values was 12% (range 2-17%).

5.4 Discussion

The experiments reported in this chapter have investigated the possibility that there is more than one *JAK2* V617F-positive population in mutant-positive ET patients. Two different *JAK2* SNPs were exploited in order to identify the allele on which the V617F mutation had arisen, initially using a PCR-restriction enzyme digestion-PCR technique and then confirmed by TA cloning. Erythroid colonies were subsequently cultured from PB MNCs from V617F-positive female ET patients and mutant-positive colonies analysed to determine the inactivated X chromosome in order to investigate whether they shared a common clonal origin.

The results indicate that, in this small cohort of mutant-positive ET patients, the majority had evidence for more than one V617F-positive myeloid population. In 10 of the 11 patients studied the V617F mutation had arisen on both *JAK2* alleles. This implies that, in these individuals, the mutation was acquired on at least two different occasions. Because there are only two possible alleles on which the mutation could arise, it is impossible to determine the exact number of times the mutation has been acquired, except that it must have been more than once. Poisson statistics suggest that there is likely to be more than two independent mutational events in most cases: if there were just two events in a single individual, the probability of that person acquiring each event on different alleles is 0.5, and therefore on average only about half of the 11 individuals would be expected to harbour V617F on different alleles. In fact 10 of the 11 cases studied had biallelic acquisition of V617F, and the probability of this occurring by chance if each individual acquired the mutation just twice is small – equal to 11/2¹¹ or 0.005. This suggests that the majority of the patients studied had more than two mutational events. By contrast, in all five PV patients there was no evidence for more than one mutational event.

In six ET patients, the mutation arose principally on the A-allele (Table 5.2, patients E2,E3,E6, E8, E9 & E10), in three it arose principally on the G-allele (patients E4, E7 & E11), and in one patient the distribution between the two alleles was approximately equal (patient E5). In the two patients in whom platelet or CD34+ cell cDNA was analysed (patients E8 and E9), the A:G ratio was comparable in these cells to that obtained in neutrophils. Likewise, analysis of samples from the one patient (E9) at separate time points yielded equivalent results. This suggests that the A:G ratio is similar regardless of cell lineage studied, which is compatible with the idea of the mutational events occurring in a multipotent progenitor, and that it remains stable over time.

The findings from the experiments using the *JAK2* intron 14 SNP (rs12343687) supported these conclusions. In the six patients who were heterozygous for this SNP, the PCR-restriction

enzyme digest-PCR technique showed that mutant-positive alleles were associated with a mix of C and T alleles. Data from the TA cloning approach was in keeping with the acquisition of *JAK2* V617F on both alleles in two of the three patients studied (E3 & E7; Table 5.2). In one patient, however, (E9; Table 5.2) the mutant-positive clones carried only A-alleles at the intron 14 SNP, even though the PCR-restriction enzyme digest-PCR technique suggested a mix of mutant-positive alleles. The reason for this discrepancy is unclear, but may be due to chance since only 14 mutant-positive clones were analysed.

Data from haplotype analysis (see Chapter 6) suggests that the C allele at rs12343687 and the A allele at rs2230724 are both part of the 46/1 haplotype. In keeping with this, the patient (E3) who predominantly acquired V617F on the C allele at rs12343687 also predominantly acquired the mutation on the A allele at rs2230724. Conversely, the two patients (E7 and E11) who predominantly acquired V617F on the T allele at rs12343687 also predominantly acquired the mutation on the G allele at rs2230724. On patient (E5) appeared to acquire the mutation equally on both alleles for both SNPs investigated. In two patients (E6 and E9) the two SNPs gave discordant results – in both cases the SNP rs12343687 data suggested that the mutation was acquired equally on both alleles whereas the SNP 2230724 data indicated preferential acquisition of the mutation on the A allele.

Overall, these results strongly suggest that ET patients may acquire *JAK2* V617F more than once. This is not the first evidence for the existence of multiple mutant-positive myeloid populations in MPN patients. Pardanani *et al* (2008) and Beer *et al* (2009) both reported that approximately 0.5-1% of MPN patients harboured more than one MPN-associated mutation (see Section 5.1.2). However, the data reported in this chapter goes further and suggests that the majority (10 of 11 patients studied, 91%) of ET patients have more than one V617Fpositive population.

Data from Olcaydu *et al* (2009) supports the assertion that MPN patients may have more than one V617F-positive population, although they observed a markedly lower prevalence of this phenomenon. In their seminal study reporting the existence of a haplotype associated with acquisition of *JAK2* V617F, they analysed a PCR product spanning both the mutation site and the intron 14 SNP by allele-specific PCR and observed that V617F preferentially arose on the 46/1 haplotype allele (93 of 109 patients, 85%). In three cases (3%), however, the mutation appeared to have arisen on both alleles. The authors concluded that there was evidence for more than one V617F events in a minority of MPN patients. They speculated that the true prevalence was likely to be greater than 3% since patients in whom all the mutations had arisen on the same allele would appear to have just a single mutational event using this technique.

This does not, however, explain the marked difference between the proportion of patients with biallelic *JAK2* V617F mutations reported by Olcaydu *et a*l (2009) and that observed in our cohort – 3% and 91% respectively. It may be in part due to differences in the patient cohort. Of the five PV patients reported in this thesis, none had evidence for more than one V617F-positive event, and it may be that the frequency of this phenomenon is lower in PV than in ET. Consequently, the presence of PV patients in the Olcaydu *et al* (2009) series might have reduced the apparent frequency of biallelic mutations. Since the composition of their MPN cohort is not reported, this cannot be verified.

Another suggested explanation for this difference is that the apparent proportion of patients with biallelic mutations in our cohort was exaggerated by PCR artefact due to use of relatively high PCR cycle numbers, resulting in template switching during PCR extension and the generation of chimeric DNA molecules (Beer *et al*, 2010b). They argued that the true prevalence of more than one V617F event in ET is 5-10%. Although this is a possibility, if the findings presented in this chapter were due to random PCR error, it is improbable that two different samples, either from two different lineages or two separate time points, from the

same patient would consistently yield similar A:G ratios whilst samples from different patients produced different ones. Moreover, in all five PV patients studied *JAK2* V617F was present only on one allele (the A-allele), despite using identical methodology.

Regardless of the precise figure, it is apparent that a proportion of ET patients have more than one V617F-positive myeloid population. This does not, however, prove that the disorder is truly polyclonal, since all the mutant-positive populations may be derived from a pool of clonally-related progenitors which were themselves derived from a single transformed precursor, implying the existence of a pre-V617F event. In this case, in female patients all the V617F-positive cells would share a common inactivated X-chromosome. Alternatively, the mutant-positive cells may be derived from 'normal' polyclonal progenitors, implying that acquisition of *JAK2* V617F was the initial pathogenic event, and the disorder is truly oligo- or polyclonal. In this case the V617F-positive cells would not necessarily share a common inactivated X-chromosome. For this reason, establishing the clonal relationship of the V617Fpositive cells was particularly important in providing an insight into the pathogenesis of the disease.

XCIP analysis of V617F-positive BFU-Es was performed in six patients. In one patient 38% of colonies inactivated one X-chromosome whilst 62% of colonies inactivated the other – a ratio which was similar to the patient's constitutional XCIP (31%:69%). It therefore appears that in this patient the mutation arose independently in more than one clonally unrelated progenitor. Results from the other five patients suggested that in these cases all the V617F colonies inactivated a single X-chromosome, in keeping with the progenitors which acquired *JAK2* V617F being phylogenetically related and therefore the progeny of a single transformed precursor. It is also possible, however, they were not clonally related and simply shared a common inactivated X-chromosome by chance. The chance of two unrelated progenitors sharing the same inactivated X-chromosome will depend on the patient's constitutional (i.e. T-cell) XCIP – for instance in patient E7 (Table 5.3), whose T-cell XCIP was 82%:18%, the chance

of the mutation arising in two unrelated progenitors which both inactivate the A-allele is approximately (82%)² i.e. 67%. Given the relatively small number of colonies with XCIP results (between 3 and 37), it is possible that in some cases the colonies shared the same inactivated X-chromosome by chance.

If the inactivated X-chromosome observed in the V617F-positive colonies is taken to reflect the overall XCIP of the mutant-positive population, it can be used to predict the neutrophil XCIP of these six patients, based on the T-cell XCIP and the size of the mutant-positive population, as described in Section 5.3.5 (Table 5.4). In every case, the difference between the expected and the observed neutrophil XCIP was within the technical limits of the HUMARA (20%). This is in keeping with one of the conclusions drawn in Chapter 3 – that the neutrophil XCIP is a function of the T-cell XCIP, the mutant level, and the XCIP of the mutant-positive population – but refines it by accommodating the possibility that the V617F-positive cells may not necessarily all inactivate the same X-chromosome, as in patient E12.

Whilst it is not possible to draw firm conclusions from such a small cohort, the data suggests that at least some patients may harbour more than one V617F-positive population which are not clonally related and have arisen from polyclonal progenitors, although in the majority the mutant-positive populations are phylogenetically related and derived from a single transformed precursor. Studies in larger numbers of patients are required to confirm these observations.

These finding are compatible with those of Beer *et al* (2009) who determined the XCIP of the mutant-positive colonies in three female patients that carried two distinct mutant populations, either V617F and *MPL* W515, or V617F and a *JAK2* exon 12 mutation. They observed that in two, including one ET patient, the colonies carrying different mutations inactivated different X-alleles, proving that in these two patients the two clones had arisen independently and were not from a shared founder clone. Mutant-positive colonies from the third patient all expressed

Prediction of neutrophil XCIP in 6 patients, based on their CD3-positive XCIP, mutant levels in neutrophil DNA and the XCIP of Table 5.4

V617F-positive colonies

%difference (observed- predicted)			9	6	17	2	14	14
Predicted neutrophil XCIP [‡]	A% B%		14 86	91 9	28 72	33 67	68 32	32 68
XCIP in V617F- positive BFU-Es [†]	A% B%	'A _c ' 'B _c '	0 100	100 0	0 100	38 62	100 0	0 100
Observed neutrophil XCIP	A% B%		20 80	82 18	45 55	31 69	82 18	18 82
CD3+ XCIP	A% B%	A_{T}^{\prime} B_{T}^{\prime}	46 54	83 17	38 62	31 69	64 36	40 60
% WT cells		,λ,	30%	54%	74%	68%	%06	80%
% V617F+ cells*		,Z,	%02	46%	26%	32%	10%	20%
% JAK2 V617F (neuts)			35%	23%	13%	16%	5%	10%
Patient			E6	E7	E8	E12	E13	E14

* Assumes V617F-positive cells are heterozygous for the mutation

⁺ Based on inactivated X-chromosome ratios of V617F-positive colonies (Table 5.3)

^{\pm} Using the calculation: predicted neutrophil XCIP (A%:B%) = [(y/100 x A_c) + (z/100 x %A_T)] : [(z/100 x %B_T) + (y/100 x B_c)],

where y is the percentage mutant-positive cells, z is the percentage WT cells, the T-cell XCIP is A_T%: B_T%, and the ratio of inactivated A to B alleles in the V617F-positive colonies is A_c:B_c

the same X-chromosome. This would suggest that the sequence of mutational events varies between patients.

In keeping with this, studies which have attempted to determine the timing of acquisition of V617F in relation to other mutations have come to contradictory conclusions. Some published reports have suggested that JAK2 V617F is primarily a secondary molecular event in MPN. Levine et al (06 2006) reported that in 23% of V617F-positive ET patients, the clonal myeloid population based on XCIP analysis was larger than the V617F-positive one. They argued that this implied the presence of a pre-JAK2 clone, from which the JAK2 V617F-population was derived. Kralovics et al (2006) came to the same conclusion, also by comparing XCIP and V617F data, as did Delhommeau et al (2009) but using a different method. Delhommeau et al (2009) employed haematopoietic colony analysis to determine the sequence of mutational events in five patients with JAK2 V617F-positive TET2-mutated MPN, including one case of ET. In all five cases, they detected TET2-mutated haemopoietic colonies which were JAK2 WT, but they did not identify any V617F-positive, TET2 WT colonies. They interpreted this as evidence that in these patients acquisition of TET2 mutations preceded acquisition of JAK2 V617F. Similarly Beer et al (2010a) identified an individual with post-ET AML who had both MPL-mutated and V617F-positive populations, each of which was shown to be derived from a common TET2mutated clone.

There is, however, evidence that *JAK2* V617F may precede other molecular abnormalities. Colony analysis by Schaub *et al* (2009) suggested that in three of nine MPN patients with both *JAK2* V617F and del20q, including 2 ET patients, V617F preceded del20q, whereas in two patients the reverse was true. In the remaining four cases there was evidence for multiple del20q and 9pLOH events, and the sequence was uncertain. The same group subsequently investigated eight patients with V617F-positive, *TET2*-mutant MPN and found that in four patients the *TET2* mutation appeared to arise first, in two patients V617F arose first, and in the remaining two patients there were two distinct populations each carrying a single mutation

(Schaub *et al*, 2010). Saint Martin *et al* (2009) also identified a single PV patient who appeared to have acquired *JAK2* V617F prior to *TET2* mutations.

Taken together, these reports emphasise the heterogeneity of the molecular pathogenesis of V617F-positive ET, and that the role of the mutation in the disorder is not straightforward. This concept is in keeping with the main finding from the experiments presented in this chapter – that a large proportion of ET patients have more than one V617F-positive population, and that in some patients these populations are clonally distinct whereas in others they are likely to be phylogenetically related. It thus appears that whilst some patients have a monoclonal mutant-positive population, others have an oligoclonal population with more than one independent V617F-positive clone. This latter situation is reminiscent of paroxysmal nocturnal haemoglobinuria (PNH). Multiple clones, each with a different mutation in the phosphatidylinositol glycan anchor (*PIGA*) gene, are seen in some PNH patients, and the extent to which the clones expand varies widely among patients (Parker 2007).

PNH is generally considered to be a benign disorder and shares a number of key characteristics with V617F-positive ET. In both conditions, the expansion of mutant-positive clones is limited and the peripheral blood is a relatively stable mosaic of normal and abnormal cells, the most common serious complications are thrombotic, and transformation to AML occurs more frequently than in haematologically normal individuals but is nonetheless uncommon (Parker 2007). Consequently some patients with *JAK2* V617F may have a benign disorder, and the presence of the mutation should not invariably be taken to indicate the presence of a malignant neoplasm.

These findings raise two further questions. If the mutation does arise repeatedly in V617Fpositive ET patients, why does the mutant level remain stable, often for many years; and what causes a specific mutation which is not observed in normal controls to occur repeatedly in some individuals?

Addressing the first question, it is not known why repeated acquisition of JAK2 V617F mutations in WT progenitors does not result in a progressive increase in the mutant level. One possibility is that the V617F-positive progenitors have a limited lifespan, resulting in an equilibrium in which the size of the V617F-positive population is determined by the rate at which old mutant-positive progenitors die and new ones arise. Alternatively, there may be some limit on the total size of the mutant-positive population, governed perhaps by the proliferative advantage conferred by JAK2 V617F compared to its JAK2 wild-type counterparts. If JAK2 V617F is a second event and arises in pre-transformed progenitors, those JAK2 WT progenitors which are also part of the pre-transformed clone may have a proliferative or survival advantage compared to truly normal progenitors. The relative advantage of acquiring JAK2 V617F might then be less marked than if it were acquired on a normal background, and expansion of V617F-positive cells would be correspondingly slower. In this way it is possible that the nature of the pre-JAK2 mutation determines the size of the stable V617F-positive population, and therefore the disease phenotype (ET or PV). The second question, why a specific mutation should occur repeatedly in some individuals, is even more intriguing but there is increasing evidence for an inherited predisposition to developing V617F-positive MPNs. The observation that PV patients and those with V617F-positive ET had a significant excess of A-alleles at the JAK2 exon 19 SNP (Table 5.1) suggested an association between these disorders and this SNP. In fact, this SNP is part of a wider MPN-associated haplotype, a subject which is explored in detail in the next chapter.

In conclusion, the findings presented in this chapter suggest that some, and possibly most, patients with *JAK2* V617F-positive ET have more than one mutant-positive population. In many cases these appear to be clonally related and may therefore have arisen from a larger clonal population, but in at least some cases they are unrelated and may indicate that *JAK2* V617F has been acquired by a number of polyclonal haematopoietic progenitors. This is compatible with other studies which have indicated that ET is a heterogeneous disease, and suggests that
the role of *JAK2* V617F in the pathogenesis of ET is complex and not directly analogous to that of BCR-ABL in CML. Finally, these results show that at least some V617F-positive ET patients have an oligoclonal disorder, in keeping with its benign clinical course, and more generally they indicate that the presence of an acquired mutation in a haematological disorder does not necessarily imply the presence of a malignant disease.

6 Investigations into the 46/1 haplotype

6.1 Introduction

Inherited causes of thrombocytosis which show classical Mendelian inheritance are well described but uncommon (Table 1.3). Moreover, they do not behave like sporadic ET, since individuals with these conditions typically develop a polyclonal thrombocytosis at an early age, do not manifest the increased risk of thrombotic and leukaemic complications seen in the sporadic variant, and very rarely harbour the mutations associated with sporadic ET (Skoda 2009; Teofili *et al*, 2007). There is, however, increasing evidence that at least a proportion of patients with sporadic ET have an inherited predisposition to develop the disorder. This chapter will examine this phenomenon and investigate possible underlying mechanisms.

6.1.1 Familial clustering of ET

Familial clustering of individuals with classical, adult-onset ET and other MPNs has been repeatedly described (Brubaker *et al,* 1984; Fickers and Speck 1974; Janssen *et al,* 1990) but was for many years regarded as a rare phenomenon of questionable significance. It was thought that, whilst this might have suggested the existence of a genetic or environmental factor which predisposed to the development of MPNs in a small number of cases, it may equally well have been due to chance (Perez-Encinas *et al,* 1994). A recent Swedish Registry study (Landgren *et al,* 2008), however, has convincingly demonstrated that this clustering is unlikely to be due to chance. This study determined the relative risk (RR) of developing an MPN for 24,577 first-degree relatives of 11,039 MPN patients, compared to 99,542 first-degree relatives of 43,550 haematologically normal matched controls. The overall RR was 5.6 times

greater in relatives of MPN patients compared to relatives of controls, although it varied depending on the MPN subtype of the proband. First degree relatives of ET patients had a RR of 6.8 compared to control relatives of developing an MPN, not necessarily ET, but this figure was 4.9 and 2.7 for relatives of PV and MF patients respectively. Relatives of ET patients were at greatest risk of developing ET (RR = 8.0) but also had an increased risk of developing PV (RR = 5.4), whilst relatives of PV patients had a similar risk of developing both PV and ET (RR = 6.0 and 5.4 respectively).

Familial clustering of MPN cases, therefore, appears to occur more frequently than would be expected by chance, but the mechanism for this is unknown. Moreover, the majority of MPN cases do not arise in familial clusters and it is possible that the clustered cases represent a different disease than the much commoner sporadic ones, with a different pathogenesis. A number of recent reports, however, suggest that this is not the case and that MPN patients within family clusters frequently harbour the same genetic mutations as those with the sporadic disorder.

Bellanné-Chantelot *et al* (2006) screened 46 'MPN families' – defined as having at least two members, mostly first-degree relatives, diagnosed with MPN – for the presence of *JAK2* V617F. They found that in 6 families the mutation was not detectable in any individual, in 22 families all affected individuals carried the mutation, and most significantly, in 18 families some affected individuals carried the mutation whilst others did not. This last finding suggested that *JAK2* V617F was not a prerequisite for development of MPN, even in patients with V617Fpositive relatives. Furthermore, in those family members with V617F-positive MPN, the mutation was detectable in myeloid cells but, with one exception, not in their lymphoid cells. None of their haematologically normal relatives harboured the mutation. This would suggest that the mutation was acquired rather than constitutional. Combined with the observation that in some families only a proportion of members with an MPN carried *JAK2* V617F, this might imply the existence of a heritable factor which predisposed to acquisition of several

mutations, including *JAK2* V617F, which in turn contributed to development of the MPN. Another group subsequently reported similar findings in their study of 10 Italian MPN families investigated with regards to *JAK2* V617F status (Rumi *et al*, 2006). They observed that in four families some, but not all, of the affected individuals carried *JAK2* V617F, and in the V617Fpositive individuals the mutation was confined to myeloid cells. Of the 11 informative female MPN patients studied using XCIP analysis, eight had clonal haemopoiesis, of whom seven were V617F-positive.

The similarity between cases of sporadic MPN and those occurring within family clusters was further highlighted by a later study which reported on 458 patients with apparently sporadic MPN (Rumi *et al*, 2007). Detailed investigation of family members revealed that 35 (8%) of these 458 patients in fact had a relative with an MPN. The natural history of the disorder was similar regardless of whether or not the patient had an identifiable relative with MPN. Of note, second-generation patients appeared to develop the disorder significantly earlier than firstgeneration ones and exhibited telomere shortening, indicative of disease anticipation. As noted in the previous studies, in some families only a proportion of MPN patients were *JAK2* V617F-positive.

In summary, there is increasing evidence for the existence of a mechanism which confers an inherited susceptibility to acquiring MPN-associated mutations such as *JAK2* V617F in the familial MPN clusters. Because sporadic MPN shares a number of clinical and molecular characteristics with MPN occurring in familial clusters, such a mechanism may be common to both variants. One possibility is that in the familial variant there is a germline mutation in an unknown gene which results in a tendency to acquire further genetic events, and that this mutation is acquired in the truly sporadic condition. Since mutations in *TET2* occur in up to 17% of sporadic MPN patients (Delhommeau *et al*, 2009; Tefferi *et al*, 2009), and often precede *JAK2* V617F in patients who harbour both mutations, and because *TET2* is believed to be a tumour suppressor gene, inherited mutations in this gene were proposed as a mechanism

for familial MPNs (Saint-Martin *et al*, 2009). However, two studies have failed to support the suggestion that germline mutations in *TET2* underlie the familial occurrence of MPN (Olcaydu *et al*, 2010; Saint-Martin *et al*, 2009).

6.1.2 The 46/1 haplotype

Currently, therefore, no single mutation has been found to convey an increased risk of acquiring MPN-associated mutations. Instead, there is evidence that a specific haplotype within the *JAK2* gene may do so, at least in sporadic MPN. Three initial studies reported an association between this haplotype and V617F-positive MPNs, though it was unclear whether the haplotype also predisposed to V617F-negative MPNs (Jones *et al*, 2009; Kilpivaara *et al*, 2009; Olcaydu *et al*, 2009). All three studies, published simultaneously, arrived at similar conclusions using different approaches.

Jones *et al* (2009) set out to investigate whether *JAK2* V617F arose in association with a particular *JAK2* haplotype by studying six *JAK2*-spanning SNPs (Figure 6.1): rs7864782 (A/G), rs10758669 (A/C), rs7046736 (A/C), rs1234241 (G/C), rs10974947 (A/G) and rs 2031904 (A/G). They selected 142 MPN patients who were heterozygous for these SNPs and had a mutant level greater than 50%, which was taken to indicate the presence of 9pLOH. They predicted that 9pLOH should result in preferential amplification of the SNP allele which was in *cis* with the mutant allele. By determining the allelic ratios at each SNP using pyrosequencing, they established the SNP allele sequence on which the mutation had arisen. In 109 (77%) of the 142 patients studied, the mutation had been acquired in *cis* with one particular haplotype, characterised by the SNP allele sequence ACAGGG at the six loci studied.





The 46/1-associated SNP alleles are shown beside the reference

They then analysed the Wellcome Trust Case Control Consortium (WTCCC) SNP data from 1,500 UK blood donors and concluded that, in these haematologically normal controls, a 280kb linkage disequilibrium (LD) block existed spanning JAK2 and two neighbouring genes, INSL4 and INSL6. Within this block, 92 JAK2 haplotypes were apparent and one of these, designated number 46, had the allele sequence ACAGGG at the six SNPs initially studied. Another haplotype, number 1, had five of these six alleles at the same loci. Jones et al (2009) focused on a total of 20 JAK2 SNPs, and at 19 of these, haplotypes 1 and 46 were mutually identical and were therefore referred to jointly as the 46/1 haplotype. This combined haplotype represented 24% of the WTCCC cohort. Using C/G SNP rs12340895, which was not one of the six SNPs originally investigated but was in complete LD with the 46/1 haplotype, they determined the haplotype frequency amongst a larger cohort of MPN patients and haematologically normal controls. The G allele, which tagged the 46/1 haplotype at rs12340895, was over-represented in PV and V617F-positive ET patients, and in one of the two groups of V617F-negative ET patients studied. Using allele-specific PCR, they then amplified a fragment spanning V617F and rs12343867, a C/T SNP located in JAK2 intron 14, which was also in LD with 46/1. Sixty-six patients who were heterozygous for both the mutation and the SNP were investigated and in 49 (74%) cases the mutation arose in *cis* with the C allele, which tagged the 46/1 haplotype. They concluded that the 46/1 haplotype predisposed to the development of V617F-associated MPNs, either by increasing the somatic mutation rate at this locus (the 'hypermutability' hypothesis) or by enhancing the selective advantage of the mutation when acquired in the presence of it (the 'fertile ground' hypothesis).

Olcaydu *et al* (2009) arrived at similar conclusions in their study which initially set out to examine the relationship between *JAK2* V617F, del13q and del20q in a patient with MF using haemopoietic colony analysis. They noted that two V617F-positive populations were apparent, each of which carried a different cytogenetic abnormality and appeared to have acquired *JAK2* V617F independently. Like Jones *et al* (2009), they used allele-specific PCR to amplify a product

spanning V617F and SNP rs12343867 in JAK2 intron 14, but here the purpose was to screen MPN patients for multiple V617F events. Analysis of 109 V617F-positive MPN patients who were heterozygous for this SNP revealed that in three cases (2.8% of the cohort) the mutation had arisen in *cis* with both SNP alleles in a single individual, in keeping with more than one V617F event. Moreover, in the 106 individuals in whom there was no evidence of multiple V617F events, the mutation had arisen in *cis* with the C allele in 93 of them (88%). Examination of a larger cohort of 333 MPN patients, of whom 213 were V617F-positive, revealed an association between the C allele at SNP rs12343867 and V617F-positivity. Similar analysis of a further seven JAK2 SNPs revealed that at three of them, rs3780367, rs10974944 and rs1159782, a specific allele (G,G and C respectively) was associated with V617F-positivity. The authors concluded that this defined a haplotype, termed 'GGCC', which predisposed to acquisition of JAK2 V617F. Their work suggested that this haplotype was confined to a 10kb region of JAK2, located within the larger LD block identified by Jones et al (2009). Since this 10kb haplotype did not contain the promoter regions of the gene, which would affect its expression, they suggested that a difference in mutability between the two haplotypes might explain their findings.

In the third study, Kilpivaara *et al* (2009) performed genome-wide SNP analysis on neutrophil DNA from 181 ET and PV patients to investigate the hypothesis that germline variation contributed to both predisposition to MPNs and phenotypic differences between the disorders. They observed that, compared to 1500 haematologically normal controls, V617Fpositive MPN patients – but not, *JAK2* WT ones – were significantly more likely to have a G rather than a C allele at SNP rs10974944, located in *JAK2* intron 12. This association was present regardless of whether neutrophil or buccal DNA from the MPN patients was studied, suggesting it was an inherited phenomenon and not due to somatic abnormalities in neutrophils, such as acquired UPD of chromosome 9. Using allele-specific PCR to amplify a DNA fragment spanning both the SNP and *JAK2* exon 14, they also noted that 38 of 45 (84%) V617F-

positive patients who were heterozygous for the SNP had acquired the mutation in *cis* with the G allele. They concluded that their data indicated the presence of a *JAK2* SNP allele which favoured the in-*cis* acquisition of *JAK2* V617F.

Taken together, the three seminal studies described above suggested that V617F-positive MPN patients were more likely to have a specific *JAK2* haplotype than haematologically normal controls, and that the mutation usually, but not invariably, arose in *cis* with this haplotype. Subsequent studies using a single tagged SNP have confirmed the association between the 46/1 haplotype and V617F-positive, but not *JAK2* WT, MPNs (Andrikovics *et al*, 2010; Pardanani *et al*, 2010; Trifa *et al*, 2010).

Establishing whether or not the association is confined to V617F-positive disorders is important since it will strongly indicate which of the two postulated mechanisms is more likely. If the association extends to JAK2 WT MPN, the argument for the 'fertile ground' hypothesis is greatly strengthened, since it is difficult to explain how a JAK2 haplotype could influence the mutability of other genes. Consequently, the recent finding that the 46/1 haplotype is associated with MPL W515 mutations is potentially of great significance and supports the 'fertile ground' hypothesis, since the MPL gene is located on chromosome 1p (Jones et al, 2010). This study pooled data from UK, German, Italian and Greek cohorts and reported that the frequency of 46/1 alleles in 176 MPL W515-mutated ET patients was significantly higher than in controls (0.37 versus 0.28, P=0.0003). However much of this difference was due to a relatively high 46/1 allele frequency in the small Greek and Italian cohorts (n = 24 and 27 respectively) of 0.46 and 0.43 respectively. By contrast, in the larger UK and German cohorts (n = 56 and 69 respectively), the 46/1 allele frequency was 0.34 and 0.36 respectively, and not significantly higher than in controls (P=0.164 and 0.267 respectively). Furthermore, a similar study by Patnaik et al (2010) found that the 46/1 allele frequency amongst 22 MPL W515positive MPN patients (0.25) was not significantly higher than in local or WTCCC controls. The association between the 46/1 haplotype and MPL W515 is, therefore, not strong. It is possible

that both the 'hypermutability' and 'fertile ground' mechanisms are involved, and this would explain the stronger association between the haplotype and V617F-positive MPN than with *JAK2* WT MPN.

6.1.3 Possible mechanisms underlying the association between the 46/1 haplotype and MPNs

In order to explain the 'fertile ground' mechanism, Jones *et al* (2009) speculated on the existence of a functional variant in LD with 46/1 which interacted with *JAK2* V617F so that acquisition of the mutation on the background of this haplotype was more likely to result in clinical disease than if it arose on the non-46/1 haplotype. They postulated that, if the 46/1 haplotype was preferentially expressed, then in-*cis* acquisition of *JAK2* V617F might result in increased expression of the mutant allele, potentially resulting in a proliferative advantage for the mutated cells. They, therefore, investigated expression of two *JAK2* SNPs which were part of the 46/1 haplotype – rs2230724 in exon 19 in 29 normal controls, and rs10429491 in exon 6 in 40 normal controls – and found that at both loci expression of the 46/1 allele was no different to that of the non-46/1 allele. Since only normal controls were examined, however, the possibility of aberrant expression of the 46/1 allele in individuals with MPN, possibly due to a functional interaction with *JAK2* V617F, could not be excluded.

Another possible mechanism for the 'fertile ground' hypothesis postulated by Jones *et al* (2009) is that the 46/1 haplotype is in LD with a functional variant which results in different baseline myeloid proliferative activity in 46/1-positive individuals compared to non-46/1 individuals. Acquisition of *JAK2* V617F might have different effects in these different contexts. They therefore, assessed myeloid colony formation in 56 normal controls with and without the 46/1 haplotype. Peripheral blood derived from individuals who were either homo- or

heterozygous for 46/1 grew significantly fewer granulocyte-macrophage colony-forming units than blood from individuals who were homozygous for the non-46/1 haplotype, although there was no difference in the number of BFU-Es. The authors interpreted this as evidence that the 46/1 haplotype affected myeloid proliferation and that this was more suggestive of the 'fertile ground' rather than 'hypermutability' hypothesis. MPN patients were not studied, and the authors did not speculate further on potential mechanisms for this functional interaction. Two studies have evaluated presentation blood counts in 46/1-positive and 46/1-negative MPN patients and found no difference in any parameter, suggesting that if any difference exists it is very subtle (Jones *et al*, 2010; Pardanani *et al*, 2010).

Although no association between the 46/1 haplotype and blood counts has yet been described, a genome-wide analysis of 13,943 patients did identify an association between another chromosome 9 SNP and variability in platelet counts (Soranzo *et al*, 2009). This SNP, rs505534, is located in an intergenic region of chromosome 9p23 approximately 8 megabases upstream from *JAK2*. If one of the rs505534 alleles is associated with a platelet count towards the upper limit of normal then acquisition of a mutation such as *JAK2* V617F or *MPL* W515 by a patient with a particular rs505534 genotype might be sufficient to generate the ET phenotype. In this case, an association might be expected between a particular rs505534 SNP allele and ET.

A further hypothesis to explain the association between V617F-positive MPNs and the 46/1 haplotype which has not been explored to date is that the latter affects *JAK2* methylation. In the presence of the 46/1 haplotype, at least two SNPs, rs12343867 and rs12340895, both located in *JAK2* intron 14, generate CpG dinucleotides which are lost in the absence of the 46/1 haplotype. Since cytosine bases within CpG dinucleotides may be methylated to form 5-methylcytosine, the presence of the 46/1 haplotype might thus alter the number of CpG dinucleotides available for methylation. Aberrant gene methylation has been implicated in the pathogenesis of many human malignancies, including numerous haematological cancers

(Esteller 2003; Feinberg and Vogelstein 1983). The best described mechanism involves the hypermethylation of promoter regions of tumour suppressor genes which normally prevent cells with DNA damage from proliferating. The consequent reduction in their expression allows genetically-mutated cells to accumulate (Esteller 2008). Even when promoter regions are not affected, aberrant methylation can predispose to hypermutability by altering the DNA conformation and inhibiting DNA mismatch repair (Muheim *et al,* 2003). Furthermore, the tendency of 5-methylcytosine to undergo spontaneous deamination to thymine (Coulondre *et al,* 1978) can also lead to hypermutability.

Mutations in *TET2*, a member of the *TET* family of putative tumour suppressor genes which are believed to be involved in epigenetic regulation, have been described in a small number of ET patients and it has been suggested that these may result in abnormal methylation (Mullighan 2009; Tahiliani *et al*, 2009). It is possible that mutations in *TET2*, or other genes regulating methylation, could interact with the 46/1 haplotype and lead to aberrant *JAK2* methylation thereby predisposing to MPN by increasing the *JAK2* mutation rate.

6.1.4 Aims

The work in this chapter aimed to study the incidence of the 46/1 haplotype in the cohort of patients described earlier in this thesis, and to investigate possible mechanisms for the 'fertile ground' and 'hypermutability' hypotheses, namely preferential expression of the 46/1 haplotype, an association between an rs505534 allele and MPN, and aberrant *JAK2* methylation in the presence of the 46/1 haplotype.

6.2 Patients, materials and methods

6.2.1 Screening JAK2 V617F-positive patients for the 46/1 haplotype using 5 JAK2 SNPs

Initially, DNA from the 11 well-characterised ET patients described in sections 5.2.2 and 5.2.3 was screened for the presence of the 46/1 haplotype by PCR and restriction enzyme digestion. Five SNPs which were reported to be associated with the 46/1 haplotype were selected (Figure 6.2). Primers were designed to amplify a fragment spanning the SNP and then allow discrimination between the two possible alleles by restriction enzyme digestion. For two SNPs (rs3780367 and rs10974944), restriction enzymes were available which could distinguish between the two SNP alleles directly but for the other three SNPs mismatched primers were required to generate a cutting site. Using the reagents and conditions described in Table 6.1, GoTaq[®] PCR (see Section 2.2.3.2) was performed using T-cell gDNA as template for each of the five SNPs to amplify a PCR product which was then digested using the specified restriction enzyme. A digestion mix was prepared containing 5 units of the appropriate enzyme, 1x appropriate buffer and where specified $100\mu g/mL$ BSA. The mix was made up to $3\mu L$ with ddH₂O and to this 7µL PCR product was added. The resulting mix was incubated using the conditions shown in Table 6.1. Following 3% agarose gel electrophoresis, digestion products were visualised by UV transillumination and the alleles identified by their size (Table 6.1). All primer sequences are listed in Appendix 1.

6.2.2 Screening for the presence of the 46/1 haplotype using SNP rs12343867

To screen the whole ET patient cohort and a group of haematologically normal controls for the presence of the 46/1 haplotype, each individual was genotyped for a single tagged SNP,

PCR and restriction digestion conditions used to determine the genotype of five SNPs within the JAK2 gene Table 6.1

SNP name (locus on <i>JAK2</i> gene)	Possible alleles (relative frequency in control population)*	Allele in 46/1 haplotype	Primers (T _a , expected size)	Restriction enzyme (incubation conditions, buffer)	Expected post-digestion allele sizes (bp)
rs3780367 (intron 10)	G/T (0.26/0.74)	σ	JAKZex10/F JAKZex10/R (59°C, 172bp)	Msel (37°C for 4 hours, buffer 2, BSA)	G: 24 + 148 T: 21 + 24 + 127
rs10974944 (intron 12)	C/G (0.75/0.25)	U	JAK2ex12/F2 JAK2ex12/R2 (59°C, 205bp)	<i>Bcl</i> l (50°C for 4 hours, buffer 3)	G: 31 + 174 C: 31 + 37 + 137
rs12343867 (intron 14)	C/T (0.29/0.71)	U	JAK2ex14/F2 JAK2ex14/(IMM)R (59°C, 164bp)	BsaAl (37°C for 4 hours; buffer 3)	C: 31 + 133 T: 164 (uncut)
rs12340895 (intron 14)	C/G (0.74/0.26)	U	JAKZex14/(MM)F3 JAKZex14/R2 (58°C, 178bp)	BstUl (60°C for 2 hours; buffer 4)	G: 22 + 156 C: 178 (uncut)
rs1159782 (intron 15)	T/C (0.74/0.26)	U	JAKZex15/(MM)F JAKZex15/R2 (56°C, 169bp)	<i>Sca</i> l (37°C or 4 hours; buffer 4)	C: 24 + 145 T: 169 (uncut)

* From HAPMAP European Cohort





For each SNP, the 46/1-associated allele is given first

rs12343867, which is in complete LD with the 46/1 haplotype. Using primers *JAK2*ex14/F2 and *JAK2*ex14/(MM)R which spanned this SNP (Figure 6.2), a 164bp PCR product was amplified from T-cell gDNA by GoTaq[®] PCR with 35 cycles of PCR and an annealing temperature of 59°C. A digestion mix was then prepared containing 5 units *Bsa*AI enzyme and 1x manufacturer's buffer 3, made up to 3µL with ddH₂O, and to this 7µL PCR product was added. The resulting mix was incubated for 4 hours at 37°C. Following 3% agarose gel electrophoresis, digestion products were visualised by UV transillumination. *Bsa*AI cut the C-allele to 133 and 31bp but did not cut the T-allele (164bp).

6.2.3 Expression analysis of the 46/1 and non-46/1 alleles

To investigate the hypothesis that preferential expression of one of the 46/1 alleles might account for the association between JAK2 V617F and the 46/1 haplotype, the relative expression of the two alleles (A/G) of rs2230724, a 46/1-associated SNP located in exon 19 of JAK2 was analysed using cDNA. A PCR-restriction enzyme digestion approach was adopted to distinguish between the two alleles, and the relative amounts of each one quantified by fragment analysis.

6.2.3.1 Validation of the technique using neutrophil gDNA

To ensure the technique was quantitative, analysis was initially performed using neutrophil gDNA from 22 normal controls who were heterozygous for rs2230724 where a 1:1 allelic ratio was expected. A 214bp fragment covering *JAK2* exon 19 and part of the flanking introns was amplified by PCR using BIOTAQTM polymerase as described in Section 2.2.3.1, with primers *JAK2*ex19/F and fluorescently-labelled *JAK2*ex19/RF, an annealing temperature of 60°C, 25 cycles of PCR, and approximately 50ng of neutrophil gDNA as template per reaction (Figure 6.3A). A digestion mix was prepared containing 5 units *Bst*NI, 1x manufacturer's buffer 2,

100µg/mL BSA, made up to 3μ L with ddH₂O and 5μ L PCR product added. This was incubated for 2 hours at 60°C then separated by fragment analysis using the CEQ 8000 Genetic Analysis system. *Bst*NI cut the G allele to 121 and 93bp but the A allele remained uncut (214bp). The relative amount of each allele was calculated from the area under the appropriate peak and expressed as a proportion of the total area of both peaks.



Figure 6.3 Schematic representation of PCR of *JAK2* exon 19 followed by *Bst*NI digestion to determine expression of A and G alleles at SNP rs2230724

- A. Using gDNA as template
- B. Using cDNA as template

6.2.3.2 Expression analysis in neutrophil cDNA

To be suitable for analysis, subjects had to be heterozygous for the SNP and have neutrophil RNA available. Neutrophil cDNA was generated using RNA from normal controls, *JAK2* V617Fpositive ET patients, and *JAK2* WT ET patients using the Invitrogen Superscript III kit as described in Section 2.2.5. A 235bp fragment spanning *JAK2* exons 18-20 was amplified by BIOTAQ[™] PCR with primers *JAK2*ex18/F and fluorescently-labelled *JAK2*ex20/RF (Figure 6.3B).

Twenty-five cycles of PCR were performed, with an annealing temperature of 62° C and using 2µL of the cDNA diluted 1 in 4 in ddH₂O as template per reaction. The PCR product was incubated with *Bst*NI as above which cut the G allele to 108 and 127bp but left the A allele uncut (235bp), analysed on the CEQ 8000 system, and the relative amounts of the A and G alleles determined by calculating the area under the appropriate peaks.

6.2.4 Analysis of SNP rs505534

The cohort of ET patients was genotyped for rs505534, a C/T SNP located on chromosome 9p, 8Mb upstream of the *JAK2* gene, to investigate the possibility that it might be a 46/1associated variant. A 222bp fragment spanning the SNP was amplified using GoTaq[®] PCR (see Section 2.2.3.2) with T-cell gDNA as template, a mismatched forward primer rs505534/(MM)F, reverse primer rs505534/R, an annealing temperature of 57°C and 35 cycles of PCR. Products were digested with *Bsm*AI. A reaction mix was prepared containing 5 units *Bsm*AI and 1x manufacturer's buffer 4, made up to 2µL with ddH₂O, and to this 8µL PCR product was added and incubated for 2 hours at 55°C, then run on a 3% agarose gel. *Bsm*AI cut the C allele to 116, 75 and 29bp, and the T allele to 145 and 75bp.

6.2.5 Methylation analysis

To investigate the possibility that the 46/1 haplotype may affect *JAK2* methylation, bisulphite sequencing followed by pyrosequencing was used to investigate the methylation status of two *JAK2* intron 14 loci: the rs12343867 C/T SNP, and the cytosine nucleotide immediately upstream of the rs12340895 C/G SNP, both of which potentially form a CpG dinucleotide in the presence of the 46/1 haplotype (Figures 6.4A,B). This approach allowed methylated and non-

methylated cytosine bases to be distinguished since treatment of DNA with bisulphite converts unmethylated cytosine to uracil, which is converted to thymine by subsequent PCR, but does not affect 5-methylcytosine. The ratio of cytosine to thymine at the locus being investigated, which can be determined by pyrosequencing, reflects the degree of methylation in the original DNA.



Figure 6.4 Effect of bisulphite treatment on possible CpG dinucleotides that involve SNPs rs12343867 and rs12340895 (*JAK2* intron 14)

Assumes that C nucleotides immediately prior to G nucleotides are methylated (C^{M}), and 100% conversion of unmethylated C nucleotides to T nucleotides. Vertical arrows indicate the nucleotide for which the methylation status was determined by pyrosequencing, and the possible CpG dinucleotide is indicated by a dashed box

- A. DNA sequence around C/T SNP rs12343867 (underlined)
- B. DNA sequence around C/G SNP rs12340895 (underlined)
- C Schematic representation showing primers used for methylation assays

6.2.5.1 Bisulphite treatment

Bisulphite treatment was performed using the Zymo EZ DNA methylation-gold kit (Cambridge Bioscience Ltd, Cambridge UK) according to manufacturers' instructions. Neutrophil gDNA, 500ng in 20µL ddH₂O, was added to 130µL CT conversion agent, mixed and incubated at 98°C for 10 minutes, 64°C for 2.5 hours, then stored at 4°C for up to 20 hours. The reaction mixture was added to 600µL M-binding buffer in a Zymo-Spin[™] IC column, mixed and spun at 14,000g for 30 seconds at room temperature. The run-through liquid was discarded and 100µL M-wash buffer was added to the column, which was spun again for 30 seconds at 14,000g. The runthrough was discarded and 200µL M-desulphonation buffer was added to the IC column, incubated at room temperature for 20 minutes, then centrifuged for 30 seconds at 14,000g at room temperature. Two washes were performed each consisting of the addition of 200µL Mwash buffer to the IC column and a 30 second room-temperature spin at 14,000g. The IC column was then transferred to a 1.5mL microcentrifuge tube and 16µL M-elution buffer added. The bisulphite-treated DNA was eluted by centrifugation at 14,000g for 30 seconds and stored at -20°C.

6.2.5.2 Qualitative analysis of SNP rs12343867 methylation status

The methylation status of C/T SNP rs12343867 within *JAK2* intron 14 was determined nonquantitatively by direct sequencing of T-cell DNA from 2 haematologically normal controls, one known to be CC, the other TT at this locus. A 198bp product spanning rs12343867 was amplified by GoTaq[®] PCR using 1.5µL bisulphite-treated DNA as template, with forward primer *JAK2*ex14/(meth)F1 and reverse primer *JAK2*ex14/(meth)R1 (Figure 6.4C). Other reagents and conditions were as described in Section 2.2.3.2. Thirty-three cycles of PCR were performed, each 30 seconds at 95°C, 30 seconds at 56°C, 45 seconds at 72°C, and a final extension of 5 minutes at 72°C. PCR products were visualised by UV transillumination following 2% agarose gel electrophoresis, then purified using the peqGold micro-spin cycle-pure kit and sequenced

by the UCL Wolfson Institute for Biomedical Research DNA sequencing service using *JAK2*ex/14(meth)F1 as the sequencing primer.

6.2.5.3 Quantitative analysis of SNP rs12343867 methylation status

Pyrosequencing was used to quantify the relative amounts of the two alleles (C or T) present at the SNP following bisulphite treatment, from which the relative degree of methylation could be calculated. If the sample analysed was known to be homozygous CC at the SNP, and these cytosine bases were completely methylated, then the post-bisulphite treatment C:T ratio is predicted to be 100%:0%. If the actual percentage C (%C) determined by pyrosequencing was less than 100%, this would imply incomplete methylation of the cytosine at this SNP. Similarly, if the sample was heterozygous CT, the predicted %C is 50%, and an observed %C of less than this would indicate incomplete methylation.

Generation of a calibration curve

In order to establish whether the assay was quantitative throughout its range, a calibration curve was first established using standards generated by mixing T-cell DNA from two normal controls, one known to have rs12343867 genotype CC, the other TT. Results from the previous stage indicated the C allele was completely methylated in control samples (see Section 6.3.5.1.). The control DNA samples were therefore mixed to give an expected %C of 0%, 2%, 5%, 10%, 25%, 50%, 75%, 90%, 95%, 98% and 100% and then bisulphite treated using the Zymo EZ DNA methylation-gold kit as described above. A 198bp PCR product spanning the SNP was then amplified in a 50µL GoTaq[®] PCR using 2.5µL bisulphite-treated DNA with forward primer *JAK2*ex14/(meth)F1 and a biotinylated reverse primer *JAK2*ex14/(meth)R1, both at 0.08mM. Fifty cycles of PCR were performed, each of 95°C for 30 seconds, 56°C for 30 seconds and 72°C for 45 seconds followed by a 5 minute 72°C extension stage. Samples were analysed using the PyroMark[™]Q96 MD pyrosequencing system (Qiagen, Crawley, UK) according to manufacturers' recommendations by the UCL Wolfson Institute for Biomedical Research DNA sequencing service, using JAK2ex14/(meth)seq1 as the sequencing primer. For each of the standard mixes, the %C determined by pyrosequencing was then plotted against the expected %C.

Analysis of neutrophil gDNA

To be suitable for analysis, subjects were either heterozygous or homozygous for the C allele at SNP rs12343867. Neutrophil gDNA from a cohort of normal controls and ET patients was bisulphite treated, and the 198bp PCR product spanning SNP rs12343867 amplified by PCR and analysed by pyrosequencing as described in the preceding paragraph. Samples were analysed in duplicate.

6.2.5.4 Quantitative analysis of methylation status near to SNP rs12340895

The degree of methylation of the cytosine immediately upstream of a second C/G SNP, rs12340895, was analysed in the same group of normal controls and ET patients (Figure 6.4B). A 192bp PCR product spanning SNP rs12340895 was amplified from bisulphite-treated neutrophil gDNA with conditions as described in Section 6.2.5.3 except that here forward primer JAK2ex14/(meth)F2 and biotinylated reverse primer JAK2ex14/(meth)R2 were used (Figure 6.4C). Samples were then analysed by pyrosequencing using primer JAK2ex14/(meth)seq2 as the sequencing primer.

6.3 Results

6.3.1 Screening JAK2 V617F-positive patients for the 46/1 haplotype

The 11 ET patients studied in detail in Chapter 5 were initially selected for analysis. For each patient, the genotype at each of five *JAK2* SNPs was determined (Table 6.2). Of the 11 patients, five were homozygous at all five SNPs: TT, CC, TT, CC and TT at rs3780367, rs10974944,

rs12343867, rs12340895 and rs1159782 respectively. From these patients a haplotype could be deduced, with the allele sequence TCTCT, i.e. these five patients were homozygous for the non-46/1 haplotype. The remaining six patients were heterozygous at all 5 SNPs: GT, GC, CT, GC and CT. It therefore appeared that these six patients were heterozygous for the TCTCT haplotype determined above, and for a second haplotype, GGCGC, i.e. the 46/1 haplotype. No patient was homozygous for the 46/1 haplotype and none was homozygous at some SNPs and heterozygous at others.

SNP	rs3780367	rs10974944	rs12343867	rs12340895	rs1159782	rs2230724
JAK2 locus	Intron 10	Intron 12	Intron 14	Intron 14	Intron 15	Exon 19*
46/1 allele	G	G	С	G	С	
Patient E1	TT	CC	TT	CC	TT	AG
Patient E2	TT	CC	TT	CC	TT	AG
Patient E3	GT	GC	СТ	GC	СТ	AG
Patient E4	TT	CC	TT	CC	TT	AG
Patient E5	GT	GC	СТ	GC	СТ	AG
Patient E6	GT	GC	СТ	GC	СТ	AG
Patient E7	GT	GC	СТ	GC	СТ	AG
Patient E8	TT	CC	TT	CC	TT	AG
Patient E9	GT	GC	СТ	GC	СТ	AG
Patient E10	TT	CC	TT	CC	тт	AG
Patient E11	GT	GC	СТ	GC	СТ	AG

Table 6.2JAK2 SNP genotypes of 11 JAK2 V617F-positive ET patients

* Determined in section 5.2.1

6.3.2 Genotyping normal controls and ET patients for the presence of the 46/1 haplotype using SNP rs12343867

SNP rs12343867, which tagged the 46/1 haplotype, was genotyped in 101 haematologically normal controls, 122 ET patients and 24 PV patients (Table 6.3). Within each patient category, the total number of 46/1 (i.e. C) alleles and of non-46/1 (T) alleles was calculated, assuming

that patients who were CC homozygous at rs12343867 contributed two C alleles, those who were heterozygous contributed one C allele, and TT homozygotes contributed none, and viceversa to calculate the total number of T alleles. The frequency of C alleles (as a proportion of total alleles) in each patient group was then compared to those observed in the control cohort, and to equivalent data published by Jones *et al* (2009) from the WTCCC analysis of 1,500 UK blood donors.

Table 6.3	SNP rs12343867 genotypes of the normal controls, ET and PV patients
	studied

			rs12343867 genotype* n (percent of patients)			
		n	СС	СТ	тт	
Normal controls		101	8 (8%)	42 (42%)	51 (50%)	
ET patients						
Overall		122†	15 (12%)	50 (40%)	57 (46%)	
By JAK2 status	V617F+	46†	6 (13%)	23 (50%)	17 (37%)	
	V617F-	76	9 (12%)	27 (36%)	40 (53%)	
By exon 19 SNP status‡	AA	39	11 (28%)	19 (49%)	9 (23%)	
	AG	55	4 (7%)	26 (47%)	25 (45%)	
	GG	17	0 (0%)	0 (0%)	17 (100%)	
PV patients		24	2 (8%)	16 (67%)	6 (25%)	

* The C allele is part of the 46/1 haplotype

⁺ The rs12343867 status was not known for 3 JAK2 V617F-positive ET patients

[‡] The exon 19 SNP status was not known for 14 ET patients.

Complete 9pLOH in myeloid cells would convert a patient who was heterozygous for the *JAK2* SNPs in their germline DNA to homozygosity in myeloid cells and confound the calculation above. To prevent this, T-cell gDNA was used for normal controls and ET patients. Since T-cell gDNA was not available for PV patients, only those with V617F levels of less than 80% were selected – i.e. without evidence of complete 9pLOH.

The frequency of the C allele at SNP rs12343867 was 0.33 in ET patients overall, 0.38 in V617Fpositive ET patients, 0.29 in *JAK2* WT ET patients, 0.42 in PV patients and 0.29 in local controls. In no patient group was the frequency of the C allele significantly different from local controls (P=0.41, P=0.14, P=0.91 and P=0.09 respectively, Chi square test; Table 6.4). If PV patients were combined with V617F-positive ET patients, the C allele frequency in this group was just significantly higher than in local controls (0.39 versus 0.29, P=0.05, Chi-square test; OR 1.61, 95%CI 1.02-2.54). Compared to the WTCCC cohort of normal controls, the rs12343867 C allele frequency was significantly higher in ET patients overall, in V617F-positive ET patients, and in PV patients (P=0.003, P=0.003 and P=0.008 respectively, Chi square test) but not in *JAK2* WT ET patients (P=0.14, Chi square test).

The exon 19 SNP studied in Section 5, rs2230724, has been reported to be part of the 46/1 haplotype (Jones *et al*, 2009). In the cohort studied here, an association was noted between the presence of an A allele at rs2230724 and the C allele at rs12343867 (P=0.02, Chi-square test; Table 6.4), indicating that the A allele is in LD with the 46/1 haplotype.

Of the three *MPL* W515-positive patients, two were heterozygous for the 46/1 haplotype and one was homozygous for the non-46/1 haplotype.

Frequency of rs12343867 alleles in each group studied and comparison with local and WTCCC controls Table 6.4

		,		/ colled	Cor	nparison	versus	Con	parison v	/ersus
Group	c	alleles	alleles	total alleles	P-value	OR	95% CI	P-value	OR	95% CI
Local controls	101	58	144	0.29				0.15	1.28	0.93 - 1.75
WTCCC controls*	1500	720	2280	0.24	0.15	1.28	0.93 - 1.76			
ET patients										
All patients	122	80	164	0.33	0.41	1.21	0.81 - 1.81	0.003	1.55	1.17 - 2.04
V617F-positive patients	46	35	57	0.38	0.14	1.53	0.91 - 2.56	0.003	1.94	1.27 - 3.00
JAK2 WT patients	76	45	107	0.29	0.91	1.04	0.66 - 1.66	0.140	1.33	0.93 - 1.91
AA genotype at exon 19 SNP	39	41	37	0.53	0.0003	2.75	1.6 - 4.72	<0.0001	3.50	2.23 - 5.52
AA or AG genotype at exon 19 SNP	94	75	113	0.40	0.02	1.65	1.08 - 2.51	<0.0001	2.10	1.55 - 2.85
PV patients	24	20	28	0.42	0.09	1.77	0.93 - 3.40	0.008	2.26	1.27 - 4.04
V617F-positive ET and PV patients combined	70	55	85	0.39	0.05	1.61	1.02 - 2.54	<0.0001	2.05	1.45 - 2.91

All P-values were derived using the chi-square test

* Taken from Jones *et al* (2009)

6.3.3 Expression analysis of the 46/1 and non-46/1 alleles

In order to investigate the hypothesis that differential expression of the 46/1 haplotype might account for its association with V617F-positive MPN, the relative expression of the 46/1associated and non-46/1 alleles was analysed at the *JAK2* exon 19 SNP, rs2230724. The A allele at this locus is in LD with the 46/1 haplotype (see previous section). In patients who were heterozygous for the *JAK2* exon 19 SNP, rs2230724, the relative expression of the two possible alleles (A/G) was determined by PCR and restriction enzyme digestion with *Bst*NI, using a fluorescently-labelled reverse primer to allow quantification by fragment analysis. Validation that the technique was quantitative was performed using neutrophil gDNA from 22 controls, where a 50%:50% ratio of A to G alleles was expected. The mean level of the A allele, which is part of the 46/1 haplotype, in these subjects was 48.5% of the total, with an SD of 3.2%, giving a technical range of 42.1 - 54.9% (mean ±2SD).

This technique was then used to determine relative expression of the two SNP alleles in individuals who were heterozygous at the SNP and had neutrophil cDNA available. Eight normal controls, 17 *JAK2* V617F-positive and 8 *JAK2* WT ET patients were analysed. There was no difference in the relative expression of the A allele expression between the groups, with a mean level of 54%, 52% and 52% in normal controls, V617F-positive ET and *JAK2* WT ET respectively (Figure 6.5B).



Β.



Figure 6.5 Quantification of the relative levels of SNP rs2230724 alleles A/G

- A. Levels of allele A in neutrophil genomic DNA from 22 normal controls Dashed lines indicate mean ± 2 standard deviations (42.1 - 54.9%)
- B. Levels of allele A expressed in neutrophil cDNA from normal controls, JAK2 V617F-positive ET patients, and JAK2 wild-type patients.

Figures immediately above the graph give two-tailed t-test P-values, for the indicated comparisons

6.3.4 Analysis of SNP rs505534

The C/T SNP, rs505534, located on chromosome 9p, has been reported to be associated with variability in platelet count (Soranzo *et al*, 2009). The genotype frequency of this SNP was therefore examined to determine whether it was in LD with the 46/1 haplotype and thereby provide a mechanism for the association between 46/1 and MPNs. According to HapMap data, the relative frequency of the C and T alleles in Northern European controls is 0.975 and 0.025 respectively, giving an expected CC, CT and TT genotype frequency of 95%, 5% and 0% respectively. Of the 122 ET patients analysed, the rs505534 genotype was available for 114. The genotype was CC in 109 (96%) patients, CT in 5 (4%) and no patient was genotype TT. There was, therefore, no association between rs505534 genotype and ET, regardless of *JAK2* V617F, rs2230724 or rs12343867 genotype (Table 6.5).

Group	n	Total alleles	C- alleles	T- alleles	Frequency of C allele (% of total)	P-value (Chi-square)
Expected	114	228	222*	6*	98	
All ET patients	114	228	223	6	98	1.00
ET patients by JAK2 status						
V617F-positive patients	45	90	88	2	98	1.00
JAK2 WT patients	67	134	131	3	98	1.00
ET patients by SNP rs2230724	4 genotyp	e				
AA genotype	36	72	70	2	97	1.00
ET patients by SNP rs123438	67 genoty	ре				
CC genotype	13	26	24	2	92	0.19
CT genotype	45	90	87	3	97	0.71
TT genotype	53	106	106	0	100	0.18

Table 6.5SNP rs505534 genotypes of 114 ET patients

*Based on HapMap Northern European control cohort (frequency of C allele = 0.975, T allele = 0.025)

6.3.5 Methylation analysis of normal controls and ET patients

6.3.5.1 Qualitative analysis of SNP rs12343867 methylation status

At two *JAK2* SNPs, rs12343867 and rs 12340895, the 46/1 haplotype introduces two CpG dinucleotides which are not present the non-46/1 haplotype. In order to investigate the hypothesis that the 46/1 haplotype may affect *JAK2* methylation, the degree of methylation at these two 46/1-associated *JAK2* SNPs was determined. Neutrophil gDNA was bisulphite treated to convert unmethylated cytosine bases to thymine but leave methylated cytosine bases unchanged. Direct sequencing of the bisulphite-treated DNA from 2 normal controls showed that the individual with SNP rs12343867 genotype CC had a single 'C' peak at the appropriate nucleotide, in keeping with complete methylation at that locus (Figure 6.6B). All the other cytosine bases within the amplicon had been converted to thymine. As expected, the individual with rs12343867 genotype TT had a single 'T' peak at the appropriate nucleotide (Figure 6.6C).

6.3.5.2 Quantitative analysis of SNP rs12343867 methylation status

Quantitative analysis of the degree of methylation at this SNP was then performed by pyrosequencing. To establish linearity of the assay throughout the range, 11 DNA mixes of CC and TT gDNA were analysed, with an expected %C which varied between 0% and 100%, assuming that all C alleles were completely methylated. A good correlation was observed between the expected %C in the calibration samples and the observed %C at the rs12343867 locus (P<0.0001, r^2 =0.999; Figure 6.7). The Y-intercept was 2.5%, indicating that at the lower end of the calibration range, the assay overestimated the %C by 2.5%. At the upper end of the calibration range, the assay underestimated the %C by 4%, i.e. when the sample contained 100% C, the observed %C was 96%, due either to technical error or to incomplete methylation of C alleles.



rs12343867 (C/T)

- A. DNA sequence prior to bisulphite treatment. SNP rs12343867 is indicated by a hollow arrow
- B. Sequence chromatogram following bisulphite treatment in a normal control with rs12343867 genotype CC. Solid arrows indicate unmethylated cytosine bases which have been converted to thymine bases
- C Sequence chromatogram following bisulphite treatment in a normal control with rs12343867 genotype TT



Figure 6.7 Calibration curve of pyrosequencing analysis of cytosine methylation at SNP rs12343867 using artificially created mixes

For SNP rs12343867 results were obtained for 12 normal controls (all CT heterozygotes) and 64 ET patients (examples shown in Figure 6.8). The assay was performed in duplicate, with good concordance for the two results for each sample (P<0.0001, r^2 =0.996; Figure 6.9). The mean difference between values in the two runs was 0.1% (range 0-8%, SD=1.3%), indicating a technical variation of 5.3% (mean±2SD).



Figure 6.8 Pyrosequencing of C/T SNP rs12343867 following bisulphite treatment of neutrophil DNA

SNP rs12343867 is indicated by the shaded grey box. The relative percentage of each allele is calculated using the peak heights

- A Pyrosequencing trace from an ET patient with SNP genotype CC
- B Pyrosequencing trace from an ET patient with SNP genotype CT

The maximum %C at the SNP locus was dependent on the individual's SNP genotype – i.e. for CC individuals, it was 100% and for CT individuals it was 50%. The results are therefore analysed according to SNP genotype (Figure 6.10). For SNP heterozygous (CT) individuals the median %C was 48% in controls (range 46-52%), 47% in *JAK2* WT ET patients (range 44-64%) and 48% in V617F-positive ET patients (range 44-75%). No patient group was different from the controls. Of note, seventeen SNP heterozygous individuals had %C levels greater than the theoretical maximum of 50%, of whom nine had %C more than 2 standard deviations above



Figure 6.9 Concordance between duplicate pyrosequencing analyses of the percentage methylation at SNP rs1234386 in normal controls and ET patients

Open circles indicate SNP heterozygous individuals, closed circles indicate SNP homozygous individuals

the group median, comprising two *JAK2* WT patients and seven V617F-positive patients. Five of the seven V617F-positive patients had mutant levels greater than the median mutant level for the group (19%).

For SNP homozygous (CC) individuals, the median %C was 96% in *JAK2* WT ET patients (range 95-97%) and 97% in V617F-positive ET patients (range 96-97%). Reference to the calibration curve suggests that within technical limits these values were equivalent to 100% methylation

(Figure 6.7). The difference in median %C between *JAK2* WT and V617F-positive groups was not significant (P=0.19, t-test).



Figure 6.10 Pyrosequencing analysis of methylation at SNP rs12343867 in normal controls and ET patients

Solid horizontal lines indicates median (value shown to the left); dashed lines indicate median ±2 standard deviations

For the V617F-positive patients, patients with mutant levels below group median (19%) are represented by filled squares, those with mutant levels above group median are represented by open squares

6.3.5.3 Quantitative analysis of methylation status near to SNP rs12340895

The methylation of the cytosine base which was immediately upstream of C/G SNP rs12340895 was then studied (Figure 6.4B). At this locus, a CpG dinucleotide was only possible when a G allele was present at SNP rs12340895. For individuals who were GG homozygous at the SNP, CpG dinucleotides were generated on both alleles and their maximum expected %C was, therefore, 100%. For CG SNP heterozygotes, the maximum expected %C was 50%. For SNP heterozygous individuals the median observed %C was 23% in controls (range 20-27%), 21% in *JAK2* WT ET patients (range 18-27%) and 24% in V617F-positive ET patients (range 18-52%; Figure 6.11). The difference between the latter two groups was statistically significant (P=0.009, t-test), due to five patients in the V617F-positive group with %C greater than two standard deviations above the group median. No difference between any of the other groups was observed. For SNP homozygous individuals, the median %C was 96% in both *JAK2* WT ET patients (range 23-97%) and V617F-positive ET patients (range 96-97%) in keeping with complete methylation, within technical limits. One JAK2 WT ET patient who was classified as GG homozygous at this SNP had an observed %C of 23%, suggesting a sampling error and that this patient was SNP heterozygous.

6.3.5.4 Comparison of methylation data for SNP rs12343867 and SNP rs12340895

There was a strong correlation between the observed %C at the two SNPs studied (r^2 =0.887, P<0.001, Figure 6.12). The four SNP-heterozygous individuals with the highest %C at SNP rs12343867 also had the highest %C at SNP rs12340895.



Figure 6.11 Pyrosequencing analysis of methylation at SNP rs12340895 in normal controls and ET patients

Solid horizontal lines indicates median (value shown to the left); dashed lines indicate median ±2 standard deviations

For the V617F-positive patients, patients with mutant levels below group median (19%) are represented by filled squares, those with mutant levels above group median are represented by open squares


Figure 6.12 Comparison of methylation analysis results at the two SNPs studied

Dashed lines indicate median %C values at the two SNPs

Triangles indicate SNP heterozygotes, circles indicate SNP homozygotes For the V617F-positive patients, those with mutant levels below the group median (19%) are represented by filled symbols and those with mutant levels above the group median are represented by open symbols

6.4 Discussion

The surprising finding that a specific *JAK2* haplotype predisposes to the development of an MPN has now been corroborated by numerous independent studies (Andrikovics *et al*, 2010; Guglielmelli *et al*, 2010; Jones *et al*, 2009; Kilpivaara *et al*, 2009; Olcaydu *et al*, 2009; Trifa *et al*, 2010; Patnaik *et al*, 2010). The mechanism, however, remains obscure. The experiments

reported in this chapter were designed to investigate whether there was an association between the 46/1 haplotype in the cohort reported in this thesis and to examine potential mechanisms.

Eleven ET patients had been studied in detail in Chapter 5 and these individuals were studied initially by genotyping five *JAK2* SNPs reported to be involved in the 46/1 haplotype (Jones *et al*, 2009; Kilpivaara *et al*, 2009; Olcaydu *et al*, 2009). Five patients were homozygous at all five SNPs, although it is notable that they were all homozygous for the non-46/1 alleles (TCTCT), and the other six patients were heterozygous for all five SNPs – i.e. they had one copy of the 46/1 haplotype. The fact that all 11 patients were either homozygous or heterozygous for all five SNPs – and none was homozygous at some loci, but heterozygous at others – is consistent with the presence of a haplotype spanning the region studied. A sixth SNP, rs2230724, located in *JAK2* exon 19 and which was studied in Chapter 5, appeared not to be in complete LD with the haplotype, since five patients were heterozygous for rs2230724 but homozygous for the other five SNPs. However, the presence of an A allele at rs2230724 was strongly associated with a C allele at rs12343867 (P=0.0003; Table 6.4), indicating the A allele at this locus is part of the 46/1 haplotype, in keeping with data from Jones *et al* (2009).

A cohort of normal controls, ET and PV patients was then screened for the presence of the 46/1 haplotype by genotyping a tagged intron 14 SNP, rs12343867. The frequency of the C allele, which tagged the 46/1 haplotype, was significantly greater in PV and V617F-positive ET patients combined than in local controls (0.39 versus 0.29, P=0.05; Table 6.4). In all other patient groups, the C allele frequency was similar to local controls. Compared with WTCCC controls, however, it was significantly higher in all patient groups except *JAK2* WT patients (P=0.008 for PV patients, P=0.003 for ET patients overall, P=0.003 for V617F-positive ET patients, P<0.0001 for PV and V617F-positive ET patients combined, P=0.14 for *JAK2* WT ET patients). These results suggest that in this cohort there was evidence for an association between the 46/1 haplotype and the development of PV and V617F-positive ET, although it

was subtle and was only apparent if the two groups were combined or individual groups were compared with a large WTCCC control group rather than a smaller cohort of local controls. No association was observed in *JAK2* WT ET patients.

These findings are broadly in keeping with the six published studies which have examined the relationship between 46/1 and MPNs (Jones et al, 2009; Kilpivaara et al, 2009; Olcaydu et al, 2009; Andrikovics et al, 2010; Pardanani et al, 2010; Trifa et al, 2010). Olcaydu et al (2009) and Kilpivaara et al (2009) analysed the frequency of particular genotypes (e.g. CC versus TT) whereas the other studies assessed the allele frequency (i.e. C and T) in affected individuals compared to controls. Table 6.6, therefore, shows the data from all six studies re-analysed by allele frequency, with the results from the cohort reported in this chapter included for comparison. All the studies which considered PV, V617F-positive ET, or both PV and V617Fpositive ET together reported that the 46/1 allele frequency was significantly higher in the patient groups compared to local controls (Jones et al, 2009; Kilpivaara et al, 2009; Olcaydu et al, 2009; Andrikovics et al, 2010; Pardanani et al, 2010; Trifa et al, 2010). Combining these studies, the overall 46/1 allele frequency was 0.51 in PV patients, 0.46 in V617F-positive ET patients, and 0.50 in both groups combined, compared to 0.26 in controls (P<0.0001 for all three comparisons). The data from these studies provides strong evidence for an association between the 46/1 haplotype and both PV and V617F-positive ET. This association was weaker in the cohort reported in this thesis than in published studies, particularly comparing patient groups to local controls, most likely due to smaller patient numbers in the cohort reported here.

It is not clear, however, whether the 46/1 haplotype is associated with the development of *JAK2* WT ET, particularly if comparison is made to local rather than WTCCC control groups. Although Olcaydu *et al* (2009) did report a difference compared to local controls, the P-value of 0.05 was borderline, and Jones *et al* (2009) observed a difference in only one of the two cohorts they studied (P=0.009 for the UK cohort, P=0.09 for the Greek cohort). Three further

Published studies examining the association between the 46/1 haplotype and ET and PV Table 6.6

JAK2 WT ET	*д	0.009	0.09 0.002 [§]	0.005 [§]	0.05	0.09 <0.0001 [§]	0.43 0.04 [§]	0.27 0.11 [§]	<0.0001	0.91 0.14 [§]
	46/1 allele frequency	0.38	0.33	0.34	0.38	0.38	0.33	0.32	0.36	0.29
	и	47	136	74	120	108	53	42	580	76
nd V617F-positive ET combined	*q	<0.0001	<0.0001	<0.0001 [§]	<0.0001	0.005	<0.0001	<0.0001	<0.0001	0.05 <0.0001 [§]
	46/1 allele frequency	0.53	0.43	0.50	0.57	0.44	0.49	0.50	0.50	0.39
PV a	и	262	143	245	213	118	231	107	1319	70
V617F-positive ET	*d	<0.0001	<0.0001	ı	ı	0.005	<0.0001	ı	<0.0001	0.14 0.003 [§]
	46/1 allele frequency	0.51	0.43	ı	ı	0.44	0.47	ı	0.46	0.38
	и	78	143	ı	ı	118	78	ı	417	46
PV	P*	<0.0001	I	I	I	I	<0.0001	I	<0.0001	0.09 0.008 [§]
	46/1 allele frequency	0.54	ı	ı	ı	I	0.48	ı	0.51	0.42
	u	184		ı	ı	ı	153	ı	337	24
ıtrols	46/1 allele frequency	0.24	0.25	0.25	0.28	0.28	0.29	0.26	0.26	0.29
Cor	и	188	108	1500^{\ddagger}	66	57	331	150	2433	101
SNP alleles	(46/1: non-46/1)	rs12343867 (C:T)	rs12343867 (C:T)	rs10974944 (G:C)	rs12343867 (C:T)	rs12343867 (C:T)	rs12343867 (C:T)	rs10974944 (G:C)		rs12343867 (C:T)
Reference		(Jones <i>et al,</i> 2009) UK cohort	(Jones <i>et al,</i> 2009) Greek cohort	(Kilpivaara <i>et al,</i> 2009)	(Olcaydu <i>et al,</i> 2009)	(Pardanani <i>et al,</i> 2010)	(Andrikovics <i>et al,</i> 2010)	(Trifa <i>et al,</i> 2010)†	Overall (weighted mean)	Reported in this thesis

* Chi-square P-value, compared to local controls unless stated otherwise

⁺ Includes 15 PMF patients (9 V617F-positive and 6 JAK2 WT); $^{\pm}$ WTCC cohort; $^{\$}$ Compared to WTCCC cohort

studies did not observe a difference compared to local controls (Andrikovics *et al*, 2010; Pardanani *et al*, 2010; Trifa *et al*, 2010). All five of these studies did report that the 46/1 allele frequency was significantly higher in *JAK2* WT ET patients than in the WTCCC group, but whether this was a valid control group for the non-UK studies is questionable. The overall 46/1 allele frequency amongst *JAK2* WT ET patients, 0.36, was significantly higher than that amongst all controls combined, 0.26 (P<0.0001). For the cohort reported in this thesis, no association between the 46/1 haplotype and *JAK2* WT ET was observed due to a lower 46/1 frequency in the *JAK2* WT ET patients compared to the published studies overall (0.29 versus 0.36) and a higher 46/1 frequency in the normal controls (0.29 versus 0.26). These differences may reflect variations in the ethnic composition of the groups or be due to chance.

Although the 46/1 haplotype is clearly an important risk factor for the development of some MPNs, its role should not be overstated. Amongst the patients reported in this thesis, 46% of ET patients and a 25% of PV patients were homozygous for the non-46/1 haplotype, suggesting that whilst the 46/1 haplotype may predispose to the development of MPNs, it is not a pre-requisite. In keeping with this, three of the patients reported in Chapter 5 in whom there was evidence that *JAK2* V617F had occurred more than once, the mutation arose predominantly in *cis* with the non-46/1 allele (G), suggesting that the haplotype predisposes to acquisition of the mutation on either allele. Similarly, in two of the three MPN patients reported by Olcaydu *et al* (2009) to have acquired *JAK2* V617F more than once, the mutation had arisen more frequently on the non-46/1 allele.

In order to explain the association between the 46/1 haplotype and MPNs, two potential hypotheses have been advanced. The hypermutability hypothesis suggests that this haplotype predisposes to the acquisition of *JAK2* V617F, and possibly other MPN-associated mutations. The fertile ground hypothesis proposes that in 46/1-positive individuals, mutations arise at the normal rate but a functional variant in LD with the haplotype means that 46/1-positive

individuals who acquire MPN-associated mutations are more likely to develop clinical disease than those who do not have this haplotype. In this chapter, both hypotheses have been considered. To investigate the fertile ground hypothesis, the possibility of preferential expression of the 46/1 allele was first investigated, since this might result in enhanced expression of V617F if it arose in *cis* with the 46/1 haplotype. The relative expression of the two possible alleles at an exonic JAK2 A/G SNP, rs2230724, was therefore studied. The A allele at this SNP is in LD with the 46/1 haplotype (see Section 6.3.2). Expression of the two rs2230724 alleles in neutrophil cDNA was similar in all three groups studied, 54% A-expression in normal controls, 52% A-expression in both V617F-positive and JAK2 WT ET patients (P>0.05 for all three comparisons, Figure 6.5B). All three values suggest the expression of the A and G alleles are balanced (i.e. 50%:50%) within the limits of technical variability for the technique. These results do not, therefore, support the notion that preferential expression of the 46/1 allele accounts for its association with V617F-positive MPNs, although the small size of the cohort studied means that subtle differences in expression cannot be excluded. These findings are in keeping with those of Jones et al (2009) who investigated the relative expression of rs2230724 alleles in 29 normal controls and of rs10429491 alleles, another 46/1-associated SNP located in exon 6 of JAK2, in 40 normal controls. They found no difference in expression of the 46/1 and the non-46/1 alleles.

Another possible mechanism to explain the fertile ground hypothesis is that 46/1 is in LD with rs505534, a chromosome 9 SNP located 8Mb upstream from *JAK2* that was associated with variability in platelet counts in a study of nearly 14,000 healthy German and British controls (Soranzo *et al*, 2009). If the *JAK2* V617F mutation arises in the presence of an rs505534 allele associated with higher platelet counts than average, then a clinical MPN might be more likely to result than if it arose on the other SNP allele. There was, however, no difference in allele frequency (C versus T) in any of the patient groups studied compared to the HapMap Northern

European control cohort and therefore no suggestion af an association between this SNP and ET or the 46/1 haplotype.

To investigate a possible mechanism to account for the hypermutability hypothesis, the possibility that the 46/1 haplotype might affect DNA methylation was considered, since changes in methylation may affect DNA conformation and influence susceptibility to mutations. In humans, DNA methylation affects cytosine bases that precede guanine bases (dinucleotide CpGs), and two of the reported 46/1-tagging SNPs (rs12343867 and rs12340895, both in intron 14) result in CpG sites when the 46/1-associated allele is present but not when the non-46/1 allele is present. To quantify the degree of methylation at these SNPs, pyrosequencing of bisulphite-treated neutrophil DNA from normal controls and ET patients, both *JAK2* WT and V617F-positive, was performed. At SNP rs12343867, the median post-bisulphite treatment %C was 47-48% in SNP heterozygous individuals, and 96% in SNP homozygous individuals (Figure 6.10), in keeping with complete methylation at this locus within the technical limits of the assay.

The degree of methylation was similar in normal controls and SNP heterozygous ET patients, regardless of *JAK2* status. It was notable, however, that a seven SNP heterozygous, mutant-positive patients had post-bisulphite %C values of greater than 50%, a figure which is not compatible with being CT heterozygous at this locus. Five of these patients had relatively high V617F mutant levels and the simplest explanation for the %C being greater than 50% in these patients was the presence of chromosome 9p LOH in some of their myeloid cells, resulting in amplification of the C allele to more than 50%. This phenomenon would not have been apparent when the patients' SNP genotypes was originally determined (Section 6.2.2) since this was done using T-cells, which are unlikely to be affected by 9pLOH. To further investigate this, it would be necessary to repeat the pyrosequencing analysis in tandem, using bisulphite-treated and non-bisulphite treated DNA from each individual, and then to compare the two

results from each individual. Overall, the C allele at SNP rs12343867 appeared to be completely methylated in both ET patients and normal controls.

At the second SNP studied, rs12340895, the median %C was 96% in SNP homozygous ET patients, but less than 25% in all three SNP heterozygous groups: 23% in controls, 21% in *JAK2* WT ET patients and 24% in V617F-positive ET patients (Figure 6.11). The %C was significantly different in these latter two groups (P=0.009), due to five mutant-positive SNP heterozygous patients with %C which was more than two standard deviations greater than the group median. These patients, three of whom had relatively high mutant levels, all had %C greater than 50% at SNP rs12343867 (Figure 6.12). The most plausible explanation is, again, the presence of 9pLOH. No other statistically significant difference was observed between any of the comparable groups at this SNP.

Amongst SNP heterozygous individuals the degree of methylation at SNP rs12340895 appeared to be markedly lower than that at rs12343867 in both controls and ET patients (overall median %C 22% versus 48%, P<0.0001). This was due, at least in part, to a software limitation which precluded accurate calculation of %C in patients who were heterozygous at SNP rs12340895 (all the controls and 49 of the 64 ET patients). The pyrosequencing analysis software could only interpret sequence data which varied a one locus, but the presence of a C allele at SNP rs12340895 affected the DNA sequence at two loci: at the SNP itself and at the preceding cytosine base, which in the presence of a C allele at the SNP was unmethylated and, therefore, converted to thymine during bisulphite treatment (Figure 6.4B). If analysis is, therefore, confined to ET patients who were homozygous for the 46/1 allele at the two SNPs, within technical limits both loci were completely methylated (median %C was 96% at SNP rs12343867 versus 97% at SNP 12340895, P=0.22). Overall, the data from these experiments suggests that the presence of the 46/1 haplotype may influence the degree of methylation of the *JAK2* gene, potentially affecting its conformation and susceptibility to acquiring mutations. It is notable that at least two of 46/1-tagging SNPs which might affect *JAK2* methylation, rs12343867 and

rs12340895, are located in intron 14, close to the V617F mutation site in exon 14. Clearly further study of other *JAK2* SNPs in larger cohorts of controls and patients is required.

In conclusion, the experiments reported in this chapter provide further evidence for a specific *JAK2* haplotype which is associated with the development of V617F-positive MPNs, though the association was weaker than that reported by other studies and did not extend to *JAK2* WT ET. Furthermore, nearly a half of ET patients and a quarter of PV patients were homozygous for the non-46/1 haplotype, indicating that the presence of the 46/1 haplotype is not required for the development of an MPN. There was no evidence that the predisposition to an MPN observed in 46/1-positive individuals is due to preferential expression of the 46/1 allele or an association with a nearby 'platelet variability' SNP. However, the two CpG dinucleotides introduced by the 46/1 haplotype within the *JAK2* gene were found to be completely methylated, and the possibility that this may affect the susceptibility to mutations merits further study.

7 Conclusions and future directions

7.1 Summary of findings

Although ET has been recognised as a clinical entity for nearly 80 years, the identification in 2005 of a recurrent acquired point mutation, *JAK2* V617F, in approximately 50% of cases resulted in a fundamental change in the way the disorder is perceived, leading to its reclassification by the WHO as a myeloproliferative neoplasm in 2008. This reclassification reflected a belief that ET is a clonal neoplastic disease which arises as a consequence of one or more acquired mutational events. However, it overlooked several characteristics of this disorder, in particular that ET does not typically behave as a malignant neoplasm and that a substantial proportion of cases are polyclonal according to XCIP data. The studies reported in this thesis, therefore, set out to investigate the molecular origins of ET, and in particular to consider the role of *JAK2* V617F in the pathogenesis of the disease and its relationship to the concept of clonality in ET. The aim was to gain a better understanding of this heterogeneous disease so that its clinical phenotype might be explicable in molecular terms.

The clinical characteristics of a cohort of 133 ET patients were evaluated and their association with haematological and molecular features examined (Chapter 3). In terms of demographic and clinical features, as well as haematological and molecular characteristics, these patients were similar to those described in most published studies. In keeping with most other series, the clinical course in the majority of patients studied was relatively benign, and the most common complications were thrombotic. Over a median time since diagnosis of 12.6 years, the thrombotic rate was 16% and the mortality was less than 5%. Although patients who had had a thrombosis were older at diagnosis than those who did not (median age 53 versus 44 years, P=0.03), presenting blood counts and a history of prior thrombosis did not predict thrombotic

risk. Of the 62 evaluable female patients, 24 (39%) had a clonal XCIP. These patients were younger at diagnosis and had higher presenting platelet counts than those with a polyclonal XCIP, and appeared to have a higher rate of thrombotic complications, though this did not reach statistical significance (53% versus 28%, P=0.11), probably due to the relatively small cohort studied. When these results were combined with data from other groups, however, a significant association between XCIP status and thrombotic rate was observed (38% in patients with a clonal XCIP versus 18% in patients with a polyclonal XCIP, P=0.008).

The *JAK2* V617F mutation was detectable in 55 (41%) of patients and, in keeping with the majority of published studies, these individuals were older at presentation than those with *JAK2* WT ET, had lower platelet counts and higher haemoglobin concentrations. The overall thrombotic risk in the two groups was not significantly different (45% versus 26%, P=0.10), but the median mutant level was higher in those who had had a thrombosis than in those who did not (11% versus 5%, P=0.02). This is consistent with data from other groups and suggests that the mutant level, rather than the presence of the mutation per se, may determine the thrombotic risk. It is also in keeping with the notion that the phenotype of V617F-positive ET lies somewhere between *JAK2* WT ET and PV in terms of presenting platelet counts and thrombotic risk. The presence of *JAK2* V617F is often taken to indicate the presence of a clonal myeloid population, yet the rate of myeloid clonality was similar amongst mutant-positive and *JAK2* WT patients (27% versus 45%, P=0.19). This may have been due to the inability of the XCIP technique to detect small clones, and conversely to the presence of *JAK2* WT patients with clonal myelopoiesis who are likely to carry other mutations.

The relative size of *JAK2* WT and V617F-positive platelet populations in ET patients was then investigated (Chapter 4). A number of prior studies had shown that, in ET, on average less than half the myeloid population carried the V617F mutation and that the size of the mutantpositive population appeared to remain stable over many years, both characteristics which might be considered unexpected for a neoplastic disease. To address the possibility that this

phenomenon was an artefact of studying neutrophils rather than the biologically more relevant platelets, mutant levels were quantified in platelet and neutrophil RNA from 10 V617F-positive ET patients, obtained prior to the initiation of cytoreduction. Whilst mutant levels were significantly higher in platelet RNA than in neutrophil RNA (27% versus 21%, P=0.002), the results indicated that on average only about half of the circulating platelets in ET were mutant-positive. None of the 10 patients studied had evidence that *JAK2* WT thrombopoiesis was suppressed in the presence of a thrombocytosis. Whilst this might have been due to defective regulation of thrombopoiesis in ET, possibly caused by impaired c-Mplmediated TPO clearance, this observation might also suggest that the *JAK2* WT platelets were abnormal in some way. Regardless of the mechanism, the implied persistence of *JAK2* WT progenitors, combined with recent evidence that some individuals harbour an inherited predisposition to acquire the mutation, raised the possibility that the *JAK2* V617F event might occur more than once in a single individual.

The possibility that more than one mutant-positive myeloid population might exist due to the acquisition of *JAK2* V617F on more than one occasion was, therefore, investigated (Chapter 5). The chromosome on which the mutation had arisen was determined in 11 V617F-positive ET patients who were heterozygous for a *JAK2* exon 19 SNP by examining which SNP allele was in *cis* with the mutation. In order to study only V617F-positive alleles, a PCR fragment which spanned both the mutation site and the SNP was amplified from neutrophil cDNA. Restriction enzyme digestion was then performed to cut *JAK2* WT products so that in a second-round PCR, only V617F-positive products were re-amplified. Analysis of the *JAK2* exon 19 SNP present in mutant-positive products revealed that in 10 of the 11 individuals studied, a mixture of SNP alleles was present. In each of these 10 patients, therefore, the mutation appeared to have arisen on both chromosomes, suggesting that the majority of ET patients studied had evidence for the independent acquisition of V617F on more than one occasion in different haematopoietic progenitors. Poisson statistics would suggest that the mutation had arisen

more than twice. These results were confirmed by subcloning the PCR product from five patients and determining the exon 19 SNP allele present in the mutant-positive clones. In all five patients the mutation appeared to have arisen more than once. The SNP allele with which the mutation was predominantly associated varied between individuals. Where samples obtained from a single patient at different time points or from different cells (platelets rather than neutrophils) were analysed, the SNP allele on which the mutation had predominantly arisen was the same in the two different samples. The concordance between different samples in a single individual suggests that the results genuinely indicated the presence of multiple V617F events and were not due to random PCR artefact, although this cannot be fully excluded. By contrast, in five PV patients studied there was no evidence for multiple mutantpositive populations.

The inactivated X-chromosome was then determined in V617F-positive erythroid colonies derived from six mutant-positive ET patients, including three previously shown to have more than one mutant-positive population using the approaches described above. In one patient, different mutant-positive colonies inactivated different X-chromosomes, indicating that in this case the V617F- positive populations were clonally distinct and that in this individual *JAK2* V617F had arisen more than once in phylogenetically unrelated progenitors. In the other five cases the mutant-positive colonies inactivated the same X-chromosome. This might suggest that in these patients – i.e. the majority – the mutant-positive populations were clonally related and that *JAK2* V617F arose more than once in different myeloid progenitors which were all derived from a single transformed progenitor. It is also possible, however, that the V617F-positive erythroid colonies from these five patients inactivated the same X-chromosome by chance. These results highlight the biological heterogeneity of ET, and suggest that in some cases at least, ET resembles a polyclonal non-malignant condition rather than a clonal neoplastic disorder.

The role of the 46/1 haplotype in the development of ET and the acquisition of *JAK2* V617F was also investigated (Chapter 6). A number of recently-published studies suggest that individuals who harbour this haplotype may be at higher risk of developing MPNs than those who do not. In keeping with this, an association was observed between the 46/1 haplotype and V617F-positive MPN amongst the patients reported in this thesis, although it was weaker than that reported by other groups. Moreover, nearly 50% of ET and 25% of PV patients were homozygous for the non-46/1 haplotype, indicating that this haplotype is not required for the development of an MPN. No association between the 46/1 haplotype and *JAK2* WT MPN was observed.

The mechanism underlying the association between the 46/1 haplotype and MPNs is unknown, but two hypotheses have been proposed. One suggests that the haplotype increases the mutational rate within the *JAK2* gene. The other argues that the mutation does not occur any more frequently than in non-46/1 individuals, but that when it does arise it is more likely to result in clinical disease, possibly due to the presence of a functional variant associated with the haplotype. A series of experiments was then performed to explore these two potential mechanisms. The possibility that the haplotype affected the *JAK2* expression or was in LD with a SNP which was associated with platelet variability was investigated, but the results did not support either hypothesis. The 46/1 haplotype was noted to introduce at least two CpG dinucleotides into the *JAK2* gene, which may be methylated and therefore alter the DNA conformation and render it more susceptible to mutations. The methylation at two 46/1associated SNPs was, therefore, investigated using bisulphite treatment followed by pyrosequencing. Both loci were observed to be completely methylated in the presence of the 46/1 haplotype, in both controls and ET patients, which might affect the mutability of the gene.

7.2 Future directions

Over the past decade, a number of important advances have been made in our understanding of the pathogenesis of ET, in particular the identification of the *JAK2* V617F and *MPL* W515 mutations and the recognition that the 46/1 haplotype may be a risk factor in the development of ET. Nonetheless, many challenges remain, both in comprehending the biology of the disease and in managing patients with it. Nearly fifty percent of patients with ET have no detectable molecular abnormality, and in these cases making the diagnosis can be problematic. Future work is, therefore, required to identify the causative mutations in these cases. It is notable that in 17 (71%) of the 24 ET patients with a clonal XCIP reported in this thesis, neither *JAK2* V617F nor *MPL* W515 mutations could be detected. Such patients would be suitable subjects for investigation by genome-wide approaches to mutation detection using high-throughput technology. Screening ET patients using high-resolution SNP microarray-profiling has not revealed any common regions of LOH or copy number alteration suitable for candidate gene mutational analysis (Kawamata *et al*, 2008; Stegelmann *et al*, 2010). High-throughput sequencing, for instance using sequencing by synthesis microarray technology, might, therefore, be the most appropriate method to address this issue.

The agents currently used to treat ET can cause a range of side-effects and consequently their use should be confined to patients at highest risk of complications. Prognostication in ET, however, remains imprecise and is still predominantly based on clinical factors, despite attempts to identify molecular markers of increased thrombotic risk. Myeloid XCIP status has previously been proposed as one such marker, but the association between XCIP status and thrombotic risk is relatively weak and only becomes clearly evident when multiple studies are combined. Moreover, this association relates predominantly to events prior to rather than after diagnosis and the assay is only performed in a research setting and cannot be used to analyse men and some women. Similarly, *JAK2* status does not appear to have a useful role in

risk stratification in ET, though the mutant level might do, at least in patients with very high mutant levels. It would worth investigating whether, in addition to the size of the V617Fpositive population, the phylogenetic relationship of its constituent cells affects prognosis. Data presented in Chapter 5 of this thesis suggests that most ET patients have more than one V617F-positive population, and in at least a minority of cases these populations are not clonally related. Larger patient cohorts might usefully be studied, therefore, to determine whether patients with more than one phylogenetically unrelated V617F-positive population have a different natural history than those whose mutant-positive populations appear to be derived from a common founder clone.

ET is incurable with current conventional therapies. Whilst the disease may not affect the life expectancy of older adults, this is probably not the case in younger patients, especially those with aggressive disease and in these cases at least, novel agents are required. Furthermore, the majority of existing treatments target the end result of the pathogenic process, the proliferating MK, and require lifelong, daily administration. If the initiating event could be identified and targeted, durable medication-free remissions or even cure may become possible. A number of new drugs which inhibit JAK2 are in development and whilst some are showing some promise, none is currently able to reproduce in MPNs the remarkable activity of BCR-ABL inhibitors seen in CML (Druker et al, 2006; Pardanani et al, 2011; Verstovsek et al, 2010). There is evidence for a so-far uncharacterised 'pre-JAK2' event in at least some V617Fpositive individuals (Campbell et al, 2006a; Theocharides et al, 2007), suggesting that V617F may not necessarily be the pathogenic abnormality in MPN, and this might explain the relative lack of efficacy of JAK2 inhibitors. In order to develop more effective targeted therapies, determining the pathogenic event is critical. The observation that JAK2 WT myelopoiesis and thrombopoiesis were not suppressed by the presence of a V617F-positive population (Chapter 4) raised the possibility that these JAK2 WT cells were themselves not normal, possibly due to a non-JAK2 abnormality. For this reason, the JAK2 WT cells present in mutant-positive

individuals warrant further investigation – for instance by growing up haemopoietic colonies and selecting the *JAK2* WT ones for analysis by high-throughput sequencing.

The evidence that most ET patients have more than one V617F-positive population (Chapter 5) also has implications for therapy. Any future attempts to eliminate the mutant-positive haemopoietic progenitor altogether and hence cure patients of the disease may prove futile if they have a pre-existing tendency to acquire *JAK2* V617F repeatedly. The role of the 46/1 haplotype in the development of MPNs warrants further research since a clearer understanding of how this inherited genotype results in predisposition to an MPN in individuals who harbour it may reveal potential therapeutic targets. For instance, if the 46/1 haplotype does result in increased *JAK2* methylation and thereby increases its susceptibility to acquiring mutations, then consolidating anti-*JAK2* therapy with hypomethylating agents might be appropriate. Methylation analysis of the entire *JAK2* gene in larger groups of normal controls and ET patients would, therefore, be useful to determine whether the presence of the 46/1 haplotype has a significant effect on overall gene methylation. Analysis of mutational rates in individuals according to 46/1 haplotype status might also be informative.

Finally, the question of whether ET is always a clonal, neoplastic disorder, as stated in the 2008 WHO classification, requires careful consideration, in view of the data presented in Chapters 4 and 5 of this thesis. Labelling ET as a neoplasm can generate unnecessary anxiety in patients if this is taken to indicate the presence of a malignant disease. Recognition that at least some ET patients have a disorder characterised by a mutant-positive population which is not monoclonal and remains stable in size over many years, should prompt further discussion as to whether ET is invariably a neoplastic condition.

7.3 Overall conclusions

In conclusion, the results presented in this thesis provide evidence for the biological heterogeneity of ET, and that accurate prognostic markers remain to be defined. Using XCIP analysis, a substantial proportion of ET patients appeared to have polyclonal myelopoiesis, including the majority of V617F-positive ones. Furthermore, in the majority of mutant-positive ET patients there was evidence which suggested that *JAK2* V617F had arisen more than once, and in at least one patient this had occurred in phylogenetically unrelated progenitors. In other patients, the multiple mutant-positive populations may have arisen from a clonal progenitor pool, in keeping with the existence of a pre-*JAK2* mutational event. Finally, these results provide further evidence for the presence of a specific *JAK2* haplotype which predisposes to the development of V617F-positive MPNs, but the mechanism is unclear. Together, these observations suggest that the role of *JAK2* V617F in the pathogenesis of ET is more complex than it being a single initiating event, that this role may vary between patients, and that the presence of the mutation does not necessarily indicate the presence of a monoclonal disorder.

Appendix 1: Primer sequences

Primer name	Primer sequence	Primer melting temperature (°C)
HUMAR/U	5'-TCCAGAATCGTTCCAGAGCGTGC-3'	67
HUMAR/D	5'-GCTGTGAAGGTTGCTGTTCCTCAT-3'	66
JAK2ex10/F	5'-TGGAACGTAGGTTGTATGTTGGC-3'	64
JAK2ex10/R	5'-CATGTAAGAAGGCAATGATAATTAATAGC-3'	64
JAK2ex12/F2	5'-GCTGTTCAAGGGTCAACTGTAG-3'	63
JAK2ex12/R2	5'-ACTGAGGATGTATTTACTTAAGTGTC-3'	62
JAK2ex14/ARMS/FO	5'-TCCTCAGAACGTTGATGGCAG-3'	62
<i>JAK2</i> ex14/ARMS/FWT	5'-GCATTTGGTTTTAAATTATGGAGTATATG-3'	63
<i>JAK2</i> ex14/ARMS/RMU	5'-GTTTTACTTACTCTCGTCTCCACAAAA-3'	64
JAK2ex14/ARMS/RO	5'-ATTGCTTTCCTTTTTCACAAGAT-3'	56
JAK2ex14/F2	5'-CAAAAGAGATTATGGCAGGTTCAAC-3'	63
JAK2ex14/(meth)F1	5'-TGTTTATGATAGTAAAAGAGATTATGG-3'	60
JAK2ex14/(meth)R1	5'-ATTTCTATAAACACCTAAATTTAACCAAAA-3'	60
JAK2ex14/(meth)seq1	5'-AATGGAATTGATAGAAATGATT-3'	52
JAK2ex14/(meth)F2	5'-AAATTTATGATTTGTTTTATTATGGTAGT-3'	58
JAK2ex14/(meth)R2	5'-CAATTTTCAAAAATATAAAAAAATATCATACA-3'	59
JAK2ex14/(meth)seq2	5'-TTTAGAGGTATGTTTTTATTTTAGT-3'	55
JAK2ex14/(MM)F	5'-CAAGCATTTGGTTTTAAATTATGGAGTA <u>C</u> GT-3'	66
JAK2ex14/(MM)F3	GAGGTATGCCTTTATTTTAGCGC-3'	62
JAK2ex14/(MM)R	5'-AGCCAAAAAATATATCTAGTATCATATCTAC-3'	64
JAK2ex14/R	5'-TAAATTATAGTTTACACTGACACCTAG-3'	61
JAK2ex14/R2	5'-CATATAACAGGTTCCAAACCACTG-3'	62

JAK2ex15/F	5'-CTTAGAACTCATGTGAAATGGC-3'	59
JAK2ex15/(MM)F	5'-ATTCATTTCATCACAAGCATAGAGTA-3'	61
JAK2ex15/R2	5'-CTGGAGCATTTAGTATGTTCAAG-3'	60
JAK2ex18/F	5'-TCAGGCCTTCTTTCAGAGCCA-3'	62
JAK2ex19/F	5'-GTATATCAGTTTAGTCCAGAGAATG-3'	61
JAK2ex19/R	5'-GACAATTTACCTTGCCAAGTTGC-3'	62
JAK2ex20/R	5'-TGTAGAGGGTCATACCGGCAC-3'	64
MPLex10/F	5'-TGGATGAGGGCGGGGCTCCG-3'	69
MPLex10/F3	5'-GTGGGCCGAAGTCTGACCCT-3'	65
MPLex10/R2	5'-CGCTCTGTGACCCCAGATCTC-3'	66
MPLW515L/F	5'-TGCTGCTGAGGTTGCAGTTTCCTGC-3'	70
<i>MPL</i> W515L/R	5'-GTGCAGGAAACTGCAACCTCAGCA-3'	68
rs505534/(MM)F	5'-GTTCATATATTTCTATCCAAAATTGTCT-3'	61
rs505534/R	5'- GCTAAGATTATACAACAATATTGTTAAG-3'	61

Appendix 2: Publications arising from experiments reported in this thesis

Lambert, J.R., Everington, T., Linch, D.C. & Gale, R.E. (2009) In essential thrombocythemia, multiple JAK2-V617F clones are present in most mutant-positive patients: a new disease paradigm. *Blood*, **114**, 3018-3023.

Lambert, J.R., Gale, R.E. & Linch, D.C. (2009) The production of JAK2 wild-type platelets is not downregulated in patients with JAK2 V617F mutant-positive essential thrombocythaemia. *Br J Haematol*, **145**, 128-130.

Austin, S.K. & Lambert, J.R. (2008) The JAK2 V617F mutation and thrombosis. *British Journal of Haematology*, **143**, 307-320.

References

- Adolfsson, J., Månsson, R., Buza-Vidas, N., Hultquist, A., Liuba, K., Jensen, C.T., Bryder, D., Yang, L., Borge, O.-J., Thoren, L.A.M., Anderson, K., Sitnicka, E., Sasaki, Y., Sigvardsson, M. & Jacobsen, S.E.W. (2005) Identification of Flt3+ Lympho-Myeloid Stem Cells Lacking Erythro-Megakaryocytic Potential: A Revised Road Map for Adult Blood Lineage Commitment. **121**, 295-306.
- Akada, H., Yan, D., Zou, H., Fiering, S., Hutchison, R.E. & Mohi, M.G. (2010) Conditional expression of heterozygous or homozygous Jak2V617F from its endogenous promoter induces a polycythemia vera-like disease. *Blood*, blood-2009-2004-215848.
- Alexander, W.S., Roberts, A.W., Maurer, A.B., Nicola, N.A., Dunn, A.R. & Metcalf, D. (1996) Studies of the c-Mpl Thrombopoietin Receptor through Gene Disruption and Activation. Stem Cells, 14, 124-132.
- Allen, A.J., Gale, R.E., Harrison, C.N., Machin, S.J. & Linch, D.C. (2001) Lack of pathogenic mutations in the 5'-untranslated region of the thrombopoietin gene in patients with non-familial essential thrombocythaemia. *Eur J Haematol*, **67**, 232-237.
- Allen, R.C., Zoghbi, H.Y., Moseley, A.B., Rosenblatt, H.M. & Belmont, J.W. (1992) Methylation of Hpall and Hhal sites near the polymorphic CAG repeat in the human androgenreceptor gene correlates with X chromosome inactivation. *Am J Hum Genet*, **51**, 1229-1239.
- Alvarez-Larran, A., Cervantes, F., Bellosillo, B., Giralt, M., Julia, A., Hernandez-Boluda, J.C., Bosch, A., Hernandez-Nieto, L., Clapes, V., Burgaleta, C., Salvador, C., Arellano-Rodrigo, E., Colomer, D. & Besses, C. (2007) Essential thrombocythemia in young individuals: frequency and risk factors for vascular events and evolution to myelofibrosis in 126 patients. *Leukemia*, 21, 1218-1223.
- Andrikovics, H., Nahajevszky, S., Koszarska, M., Meggyesi, N., Bors, A., Halm, G., Lueff, S., Lovas, N., Matrai, Z., Csomor, J., Rasonyi, R., Egyed, M., Varkonyi, J., Mikala, G., Sipos, A., Kozma, A., Adam, E., Fekete, S., Masszi, T. & Tordai, A. (2010) JAK2 46/1 haplotype analysis in myeloproliferative neoplasms and acute myeloid leukemia. *Leukemia*.
- Anger, B., Janssen, J.W., Schrezenmeier, H., Hehlmann, R., Heimpel, H. & Bartram, C.R. (1990)
 Clonal analysis of chronic myeloproliferative disorders using X-linked DNA polymorphisms. *Leukemia*, 4, 258-261.
- Antonioli, E., Guglielmelli, P., Pancrazzi, A., Bogani, C., Verrucci, M., Ponziani, V., Longo, G., Bosi, A. & Vannucchi, A.M. (2005) Clinical implications of the JAK2 V617F mutation in essential thrombocythemia. *Leukemia*, **19**, 1847-1849.
- Antonioli, E., Guglielmelli, P., Poli, G., Bogani, C., Pancrazzi, A., Longo, G., Ponziani, V., Tozzi, L., Pieri, L., Santini, V., Bosi, A., Vannucchi, A.M. & for the Myeloproliferative Disorders Research Consortium (2008) Influence of JAK2V617F allele burden on phenotype in essential thrombocythemia. *Haematologica*, **93**, 41-48.
- Antonioli, E., Carobbio, A., Pieri, L., Pancrazzi, A., Guglielmelli, P., Delaini, F., Ponziani, V., Bartalucci, N., Tozzi, L., Bosi, A., Rambaldi, A., Barbui, T. & Vannucchi, A.M. (2010)
 Hydroxyurea does not appreciably reduce JAK2 V617F allele burden in patients with polycythemia vera or essential thrombocythemia. *Haematologica*, **95**, 1435-1438.

- Avram, S., Lupu, A., Angelescu, S., Olteanu, N. & Mut-Popescu, D. (2001) Abnormalities of platelet aggregation in chronic myeloproliferative disorders. *Journal of Cellular and Molecular Medicine*, 5, 79-87.
- Axelrad, A.A., Eskinazi, D., Correa, P.N. & Amato, D. (2000) Hypersensitivity of circulating progenitor cells to megakaryocyte growth and development factor (PEG-rHu MGDF) in essential thrombocythemia. *Blood*, **96**, 3310-3321.
- Bacher, U., Haferlach, T., Kern, W., Hiddemann, W., Schnittger, S. & Schoch, C. (2005) Conventional cytogenetics of myeloproliferative diseases other than CML contribute valid information. *Annals of Hematology*, **84**, 250-257.
- Barbui, T., Barosi, G., Grossi, A., Gugliotta, L., Liberato, L., Marchetti, M., Mazzucconi, M., Rodeghiero, F. & Tura, S. (2004) Practice guidelines for the therapy of essential thrombocythemia. A statement from the Italian Society of Hematology, the Italian Society of Experimental Hematology and the Italian Group for Bone Marrow Transplantation. *Haematologica*, **89**, 215-232.
- Barbui, T., Carobbio, A., Rambaldi, A. & Finazzi, G. (2009) Perspectives on thrombosis in essential thrombocythemia and polycythemia vera: is leukocytosis a causative factor? *Blood*, **114**, 759-763.
- Barnhart, M.I., Kim, T.H., Evatt, B.L., Ragab, A.H., Lui, V.K., Herman, J. & Lusher, J.M. (1980) Essential thrombocythemia in a child: Platelet ultrastructure and function. *American Journal of Hematology*, **8**, 87-107.
- Battegay, E.J., Thomssen, C., Nissen, C., Gudat, F. & Speck, B. (1989) Endogenous megakaryocyte colonies from peripheral blood in precursor cell cultures of patients with myeloproliferative disorders. *European Journal of Haematology*, **42**, 321-326.
- Baxter, E.J., Scott, L.M., Campbell, P.J., East, C., Fourouclas, N., Swanton, S., Vassiliou, G.S., Bench, A.J., Boyd, E.M., Curtin, N., Scott, M.A., Erber, W.N. & Green, A.R. (2005) Acquired mutation of the tyrosine kinase JAK2 in human myeloproliferative disorders. *The Lancet*, 365, 1054-1061.
- Bazzan, M., Tamponi, G., Schinco, P., Vaccarino, A., Foli, C., Gallone, G. & Pileri, A. (1999) Thrombosis-free survival and life expectancy in 187 consecutive patients with essential thrombocythemia. *Annals of Hematology*, **78**, 539-543.
- Beer, P.A., Campbell, P.J., Scott, L.M., Bench, A.J., Erber, W.N., Bareford, D., Wilkins, B.S., Reilly, J.T., Hasselbalch, H.C., Bowman, R., Wheatley, K., Buck, G., Harrison, C.N. & Green, A.R. (2008) MPL mutations in myeloproliferative disorders: analysis of the PT-1 cohort. *Blood*, **112**, 141-149.
- Beer, P.A., Jones, A.V., Bench, A.J., Goday-Fernandez, A., Boyd, E.M., Vaghela, K.J., Erber, W.N., Odeh, B., Wright, C., McMullin, M.F., Cullis, J., Huntly, B.J., Harrison, C.N., Cross, N.C. & Green, A.R. (2009) Clonal diversity in the myeloproliferative neoplasms: independent origins of genetically distinct clones. *Br J Haematol*, **144**, 904-908.
- Beer, P.A., Delhommeau, F., LeCouedic, J.-P., Dawson, M.A., Chen, E., Bareford, D., Kusec, R., McMullin, M.F., Harrison, C.N., Vannucchi, A.M., Vainchenker, W. & Green, A.R. (2010a) Two routes to leukemic transformation after a JAK2 mutation-positive myeloproliferative neoplasm. *Blood*, **115**, 2891-2900.
- Beer, P.A., Ortmann, C.A., Campbell, P.J. & Green, A.R. (2010b) Independently acquired biallelic JAK2 mutations are present in a minority of patients with essential thrombocythemia. *Blood*, **116**, 1013-1014.

- Behnke, O. (1968) An electron microscope study of the megacaryocyte of the rat bone marrow: I. The development of the demarcation membrane system and the platelet surface coat. *Journal of Ultrastructure Research*, **24**, 412-433.
- Behnke, O. (1969) An electron microscope study of the rat megacaryocyte. II. Some aspects of platelet release and microtubules. *J Ultrastruct Res*, **26**, 111-129.
- Bellanne-Chantelot, C., Chaumarel, I., Labopin, M., Bellanger, F., Barbu, V., De Toma, C., Delhommeau, F., Casadevall, N., Vainchenker, W., Thomas, G. & Najman, A. (2006) Genetic and clinical implications of the Val617Phe JAK2 mutation in 72 families with myeloproliferative disorders. *Blood*, **108**, 346-352.
- Bellosillo, B., Martinez-Aviles, L., Gimeno, E., Florensa, L., Longaron, R., Navarro, G., Salar, A., Espinet, B., Sole, F., Serrano, S. & Besses, C. (2007) A higher JAK2 V617F-mutated clone is observed in platelets than in granulocytes from essential thrombocythemia patients, but not in patients with polycythemia vera and primary myelofibrosis. *Leukemia*, **21**, 1331-1332.
- Bellucci, S., Janvier, M., Tobelem, G., Flandrin, G., Charpak, Y., Berger, R. & Boiron, M. (1986) Essential thrombocythemias. Clinical evolutionary and biological data. *Cancer*, 58, 2440-2447.
- Berlin, N.I. (1975) Diagnosis and classification of the polycythemias. *Semin Hematol*, **12**, 339-351.
- Besses, C., Cervantes, F., Pereira, A., Florensa, L., Sole, F., Hernandez-Boluda, J.C., Woessner, S., Sans-Sabrafen, J., Rozman, C. & Montserrat, E. (1999) Major vascular complications in essential thrombocythemia: a study of the predictive factors in a series of 148 patients. *Leukemia*, 13, 150-154.
- Besses, C., Álvarez-Larrán, A., Martínez-Avilés, L., Mojal, S., Longarón, R., Salar, A., Florensa, L., Serrano, S. & Bellosillo, B. (2010) Modulation of JAK2 V617F allele burden dynamics by hydroxycarbamide in polycythaemia vera and essential thrombocythaemia patients. British Journal of Haematology, no-no.
- Beutler, E., Yeh, M. & Fairbanks, V.F. (1962) The normal human female as a mosaic of Xchromosome activity: studies using the gene for C-6-PD-deficiency as a marker. *Proc Natl Acad Sci U S A*, **48**, 9-16.
- Boyd, Y. & Fraser, N.J. (1990) Methylation patterns at the hypervariable X-chromosome locus DXS255 (M27 beta): correlation with X-inactivation status. *Genomics*, **7**, 182-187.
- Branehög, I., Ridell, B., Swolin, B. & Weinfeld, A. (1975) Megakaryocyte Quantifications in Relation to Thrombokinetics in Primary Thrombocythaemia and Allied Diseases. *Scandinavian Journal of Haematology*, **15**, 321-332.
- Brubaker, L.H., Wasserman, L.R., Goldberg, J.D., Pisciotta, A.V., McIntyre, O.R., Kaplan, M.E., Modan, B., Flannery, J. & Harp, R. (1984) Increased prevalence of polycythemia vera in parents of patients on polycythemia vera study group protocols. *American Journal of Hematology*, **16**, 367-373.
- Budde, U., Schaefer, G., Mueller, N., Egli, H., Dent, J., Ruggeri, Z. & Zimmerman, T. (1984) Acquired von Willebrand's disease in the myeloproliferative syndrome. *Blood*, **64**, 981-985.
- Busque, L., Mio, R., Mattioli, J., Brais, E., Blais, N., Lalonde, Y., Maragh, M. & Gilliland, D. (1996) Nonrandom X-inactivation patterns in normal females: lyonization ratios vary with age. *Blood*, **88**, 59-65.

- Campbell, P.J., Scott, L.M., Buck, G., Wheatley, K., East, C.L., Marsden, J.T., Duffy, A., Boyd,
 E.M., Bench, A.J., Scott, M.A., Vassiliou, G.S., Milligan, D.W., Smith, S.R., Erber, W.N.,
 Bareford, D., Wilkins, B.S., Reilly, J.T., Harrison, C.N. & Green, A.R. (2005) Definition of
 subtypes of essential thrombocythaemia and relation to polycythaemia vera based on
 JAK2 V617F mutation status: a prospective study. *The Lancet*, **366**, 1945-1953.
- Campbell, P.J., Baxter, E.J., Beer, P.A., Scott, L.M., Bench, A.J., Huntly, B.J.P., Erber, W.N., Kusec, R., Larsen, T.S., Giraudier, S., Le Bousse-Kerdiles, M.-C., Griesshammer, M., Reilly, J.T., Cheung, B.Y., Harrison, C.N. & Green, A.R. (2006a) Mutation of JAK2 in the myeloproliferative disorders: timing, clonality studies, cytogenetic associations, and role in leukemic transformation. *Blood*, **108**, 3548-3555.
- Campbell, P.J. & Green, A.R. (2006) The Myeloproliferative Disorders. *N Engl J Med*, **355**, 2452-2466.
- Campbell, P.J., Scott, L.M., Baxter, J.E., Bench, A.J., Green, A.R. & Erber, W.N. (2006b) Methods for the Detection of the JAK2 V617F Mutation in Human Myeloproliferative Disorders. pp. 253-264.
- Caramazza, D., Caracciolo, C., Barone, R., Malato, A., Saccullo, G., Cigna, V., Berretta, S., Schinocca, L., Quintini, G., Abbadessa, V., Di Raimondo, F. & Siragusa, S. (2009)
 Correlation between leukocytosis and thrombosis in Philadelphia-negative chronic myeloproliferative neoplasms. *Annals of Hematology*, 88, 967-971.
- Carbuccia, N., Murati, A., Trouplin, V., Brecqueville, M., Adelaide, J., Rey, J., Vainchenker, W., Bernard, O.A., Chaffanet, M., Vey, N., Birnbaum, D. & Mozziconacci, M.J. (2009) Mutations of ASXL1 gene in myeloproliferative neoplasms. *Leukemia*, **23**, 2183-2186.
- Cardier, J.E. & Dempsey, J. (1998) Thrombopoietin and Its Receptor, c-mpl, Are Constitutively Expressed by Mouse Liver Endothelial Cells: Evidence of Thrombopoietin as a Growth Factor for Liver Endothelial Cells. *Blood*, **91**, 923-929.
- Carobbio, A., Finazzi, G., Guerini, V., Spinelli, O., Delaini, F., Marchioli, R., Borrelli, G., Rambaldi, A. & Barbui, T. (2007) Leukocytosis is a risk factor for thrombosis in essential thrombocythemia: interaction with treatment, standard risk factors, and Jak2 mutation status. *Blood*, **109**, 2310-2313.
- Carobbio, A., Antonioli, E., Guglielmelli, P., Vannucchi, A.M., Delaini, F., Guerini, V., Finazzi, G., Rambaldi, A. & Barbui, T. (2008a) Leukocytosis and Risk Stratification Assessment in Essential Thrombocythemia. *J Clin Oncol*, **26**, 2732-2736.
- Carobbio, A., Finazzi, G., Antonioli, E., Guglielmelli, P., Vannucchi, A.M., Delaini, F., Guerini, V., Ruggeri, M., Rodeghiero, F., Rambaldi, A. & Barbui, T. (2008b) Thrombocytosis and leukocytosis interaction in vascular complications of essential thrombocythemia. *Blood*, **112**, 3135-3137.
- Carobbio, A., Finazzi, G., Antonioli, E., Guglielmelli, P., Vannucchi, A.M., Dellacasa, C.M., Salmoiraghi, S., Delaini, F., Rambaldi, A. & Barbui, T. (2009) JAK2V617F allele burden and thrombosis: A direct comparison in essential thrombocythemia and polycythemia vera. *Experimental Hematology*, **37**, 1016-1021.
- Case Jr, D.C. (1984) Absence of a specific chromosomal marker in essential thrombocythemia. *Cancer Genetics and Cytogenetics*, **12**, 163-165.
- Ceradini, D.J., Kulkarni, A.R., Callaghan, M.J., Tepper, O.M., Bastidas, N., Kleinman, M.E., Capla, J.M., Galiano, R.D., Levine, J.P. & Gurtner, G.C. (2004) Progenitor cell trafficking is regulated by hypoxic gradients through HIF-1 induction of SDF-1. *Nat Med*, **10**, 858-864.

- Cerutti, A., Custodi, P., Duranti, M., Noris, P. & Balduini, C.L. (1997) Thrombopoietin levels in patients with primary and reactive thrombocytosis. *British Journal of Haematology*, **99**, 281-284.
- Cervantes, F. & Rozman, C. (1982) A multivariate analysis of prognostic factors in chronic myeloid leukemia. *Blood*, **60**, 1298-1304.
- Cervantes, F., Alvarez-Larrán, A., Talarn, C., Gómez, M. & Montserrat, E. (2002) Myelofibrosis with myeloid metaplasia following essential thrombocythaemia: actuarial probability, presenting characteristics and evolution in a series of 195 patients. *British Journal of Haematology*, **118**, 786-790.
- Cervantes, F., Passamonti, F. & Barosi, G. (2008) Life expectancy and prognostic factors in the classic BCR//ABL-negative myeloproliferative disorders. *Leukemia*, **22**, 905-914.
- Champlin, R.E. & Golde, D.W. (1985) Chronic myelogenous leukemia: recent advances. *Blood*, **65**, 1039-1047.
- Cheung, B., Radia, D., Pantelidis, P., Yadegarfar, G. & Harrison, C. (2006) The presence of the JAK2 V617F mutation is associated with a higher haemoglobin and increased risk of thrombosis in essential thrombocythaemia. *British Journal of Haematology*, **132**, 244-245.
- Chim, C.-S., Kwong, Y.-L., Lie, A.K.-W., Ma, S.-K., Chan, C.-C., Wong, L.-G., San Kho, B.C., Lee, H.-K., Sim, J.P.-Y., Chan, C.-H., Chan, J.C.-W., Yeung, Y.-M., Law, M. & Liang, R. (2005)
 Long-term Outcome of 231 Patients With Essential Thrombocythemia: Prognostic
 Factors for Thrombosis, Bleeding, Myelofibrosis, and Leukemia. Arch Intern Med, 165, 2651-2658.
- Chim, C.S., Sim, J.P., Chan, C.C., Kho, B.C., Chan, J.C., Wong, L.G., Law, M., Liang, R. & Kwong, Y.L. (2010) Impact of JAK2V617F mutation on thrombosis and myeloid transformation in essential thrombocythemia: a multivariate analysis by Cox regression in 141 patients. *Hematology*, **15**, 187-192.
- Chiusolo, P., La Barbera, E.O., Laurenti, L., Piccirillo, N., Sorà, F., Giordano, G., Urbano, R., Mazzucconi, M.G., De Stefano, V., Leone, G. & Sica, S. (2001) Clonal hemopoiesis and risk of thrombosis in young female patients with essential thrombocythemia. *Experimental Hematology*, **29**, 670-676.
- Choi, E., Nichol, J., Hokom, M., Hornkohl, A. & Hunt, P. (1995) Platelets generated in vitro from proplatelet-displaying human megakaryocytes are functional. *Blood*, **85**, 402-413.
- Colombi, M., Radaelli, F., Zocchi, L. & Maiolo, A.T. (1991a) Thrombotic and hemorrhagic complications in essential thrombocythemia. A retrospective study of 103 patients. *Cancer*, **67**, 2926-2930.
- Colombi, M., Radaelli, F., Zocchi, L. & Maiolo, A.T. (1991b) Thrombotic and hemorrhagic complications in essential thrombocythemia. A retrospective study of 103 patients. *Cancer*, **67**, 2926-2930.
- Cortelazzo, S., Viero, P., Finazzi, G., D'Emilio, A., Rodeghiero, F. & Barbui, T. (1990) Incidence and risk factors for thrombotic complications in a historical cohort of 100 patients with essential thrombocythemia. *J Clin Oncol*, **8**, 556-562.
- Cortelazzo, S., Finazzi, G., Ruggeri, M., Vestri, O., Galli, M., Rodeghiero, F. & Barbui, T. (1995) Hydroxyurea for Patients with Essential Thrombocythemia and a High Risk of Thrombosis. *N Engl J Med*, **332**, 1132-1137.
- Coulondre, C., Miller, J.H., Farabaugh, P.J. & Gilbert, W. (1978) Molecular basis of base substitution hotspots in Escherichia coli. *Nature*, **274**, 775-780.

- Cramer, E.M., Norol, F., Guichard, J., Breton-Gorius, J., Vainchenker, W., Masse, J.-M. & Debili, N. (1997) Ultrastructure of Platelet Formation by Human Megakaryocytes Cultured With the Mpl Ligand. *Blood*, **89**, 2336-2346.
- Curnutte, J., Hopkins, P., Kuhl, W. & Beutler, E. (1992) Studying X inactivation. *The Lancet*, **339**, 749-749.
- Dameshek, W. (1951) Editorial: Some Speculations on the Myeloproliferative Syndromes. *Blood*, **6**, 372-375.
- Dan, K., Yamada, T., Kimura, Y., Usui, N., Okamoto, S., Sugihara, T., Takai, K., Masuda, M. & Mori, M. (2006) Clinical Features of Polycythemia Vera and Essential Thrombocythemia in Japan: Retrospective Analysis of a Nationwide Survey by the Japanese Elderly Leukemia and Lymphoma Study Group. International Journal of Hematology, 83, 443-449.
- de Sauvage, F.J., Carver-Moore, K., Luoh, S.M., Ryan, A., Dowd, M., Eaton, D.L. & Moore, M.W. (1996) Physiological regulation of early and late stages of megakaryocytopoiesis by thrombopoietin. *The Journal of Experimental Medicine*, **183**, 651-656.
- De Stefano, V., Za, T., Rossi, E., Fiorini, A., Ciminello, A., Luzzi, C., Chiusolo, P., Sica, S. & Leone, G. (2009) Influence of the JAK2 V617F mutation and inherited thrombophilia on the thrombotic risk among patients with essential thrombocythemia. *Haematologica*, **94**, 733-737.
- De Stefano, V., Za, T., Rossi, E., Vannucchi, A.M., Ruggeri, M., Elli, E., Micò, C., Tieghi, A., Cacciola, R.R., Santoro, C., Gerli, G., Guglielmelli, P., Pieri, L., Scognamiglio, F., Rodeghiero, F., Pogliani, E.M., Finazzi, G., Gugliotta, L., Leone, G., Barbui, T. & Party, F.t.G.C.M.N.W. (2010) Leukocytosis is a risk factor for recurrent arterial thrombosis in young patients with polycythemia vera and essential thrombocythemia. *American Journal of Hematology*, 85, 97-100.
- Debili, N., Coulombel, L., Croisille, L., Katz, A., Guichard, J., Breton-Gorius, J. & Vainchenker, W. (1996) Characterization of a bipotent erythro-megakaryocytic progenitor in human bone marrow. *Blood*, 88, 1284-1296.
- Delhommeau, F., Dupont, S., Valle, V.D., James, C., Trannoy, S., Masse, A., Kosmider, O., Le Couedic, J.-P., Robert, F., Alberdi, A., Lecluse, Y., Plo, I., Dreyfus, F.J., Marzac, C., Casadevall, N., Lacombe, C., Romana, S.P., Dessen, P., Soulier, J., Viguie, F., Fontenay, M., Vainchenker, W. & Bernard, O.A. (2009) Mutation in TET2 in Myeloid Cancers. N Engl J Med, 360, 2289-2301.
- Ding, J., Komatsu, H., Wakita, A., Kato-Uranishi, M., Ito, M., Satoh, A., Tsuboi, K., Nitta, M., Miyazaki, H., Iida, S. & Ueda, R. (2004) Familial essential thrombocythemia associated with a dominant-positive activating mutation of the c-MPL gene, which encodes for the receptor for thrombopoietin. *Blood*, **103**, 4198-4200.
- Dobo, I., Boiret, N., Lippert, E., Girodon, F., Mossuz, P., Donnard, M., Campos, L., Pineau, D., Bascans, E., Praloran, V. & Hermouet, S. (2004) A standardized endogenous megakaryocytic erythroid colony assay for the diagnosis of essential thrombocythemia. *Haematologica*, 89, 1207-1212.
- Druker, B.J., Guilhot, F., O'Brien, S.G., Gathmann, I., Kantarjian, H., Gattermann, N., Deininger, M.W.N., Silver, R.T., Goldman, J.M., Stone, R.M., Cervantes, F., Hochhaus, A., Powell, B.L., Gabrilove, J.L., Rousselot, P., Reiffers, J., Cornelissen, J.J., Hughes, T., Agis, H., Fischer, T., Verhoef, G., Shepherd, J., Saglio, G., Gratwohl, A., Nielsen, J.L., Radich, J.P., Simonsson, B., Taylor, K., Baccarani, M., So, C., Letvak, L. & Larson, R.A. (2006) Five-Year Follow-up of Patients Receiving Imatinib for Chronic Myeloid Leukemia. *New England Journal of Medicine*, 355, 2408-2417.

- Dubé, I.D., Gupta, C.M., Kalousek, D.K., Eaves, C.J. & Eaves, A.C. (1984a) Cytogenetic studies of early myeloid progenitor compartments in Ph1-positive chronic myeloid leukaemia (CML) I. Persistence of Ph1-negative committed progenitors that are suppressed from differentiating in vivo. *British Journal of Haematology*, 56, 633-644.
- Dubé, I.D., Kalousek, D.K., Coulombel, L., Gupta, C.M., Eaves, C.J. & Eaves, A.C. (1984b)
 Cytogenetic studies of early myeloid progenitor compartments in Ph1-positive chronic myeloid leukemia. II. Long-term culture reveals the persistence of Ph1-negative progenitors in treated as well as newly diagnosed patients. *Blood*, 63, 1172-1177.
- Dupont, S., Masse, A., James, C., Teyssandier, I., Lecluse, Y., Larbret, F., Ugo, V., Saulnier, P., Koscielny, S., Le Couedic, J.P., Casadevall, N., Vainchenker, W. & Delhommeau, F. (2007) The JAK2 617V>F mutation triggers erythropoietin hypersensitivity and terminal erythroid amplification in primary cells from patients with polycythemia vera. *Blood*, **110**, 1013-1021.
- El-Harith, E.-H.A., Roesl, C., Ballmaier, M., Germeshausen, M., Frye-Boukhriss, H., Neuhoff, N.v., Becker, C., Nürnberg, G., Nürnberg, P., Ahmed, M.A.M., Hübener, J., Schmidtke, J., Welte, K. & Stuhrmann, M. (2009) Familial thrombocytosis caused by the novel germ-line mutation p.Pro106Leu in the MPL gene. *British Journal of Haematology*, 144, 185-194.
- El-Kassar, N., Hetet, G., Briere, J. & Grandchamp, B. (1997) Clonality Analysis of Hematopoiesis in Essential Thrombocythemia: Advantages of Studying T Lymphocytes and Platelets. *Blood*, **89**, 128-134.
- Elagib, K.E., Racke, F.K., Mogass, M., Khetawat, R., Delehanty, L.L. & Goldfarb, A.N. (2003) RUNX1 and GATA-1 coexpression and cooperation in megakaryocytic differentiation. *Blood*, **101**, 4333-4341.
- Elmaagacli, A.H., Beelen, D.W., Opalka, B., Seeber, S. & Schaefer, U.W. (2000) The amount of BCR-ABL fusion transcripts detected by the real-time quantitative polymerase chain reaction method in patients with Philadelphia chromosome positive chronic myeloid leukemia correlates with the disease stage. *Annals of Hematology*, **79**, 424-431.
- Emadi, A. & Spivak, J.L. (2009) Anagrelide: 20 years later. *Expert Review of Anticancer Therapy*, **9**, 37-50.
- Emmons, R., Reid, D., Cohen, R., Meng, G., Young, N., Dunbar, C. & Shulman, N. (1996) Human thrombopoietin levels are high when thrombocytopenia is due to megakaryocyte deficiency and low when due to increased platelet destruction. *Blood*, **87**, 4068-4071.
- Epstein, E. & Goedel, A. (1934) Haemorrhagische thrombocythamie bei vascularer schrumpfmilz. *Virchows Arch A Pathol Anat Histopathol*, **293**, 233-236.
- Escoffre-Barbe, M., Amiot, L., Beaucournu, P., Jego, P., Grulois, I., Grosbois, B., Bernard, M., Fest, T., Lamy, T. & Fardel, O. (2006) Spontaneous megakaryocytic colony formation does not discriminate between essential thrombocythemia and polycythemia vera. *American Journal of Hematology*, **81**, 554-556.
- Esteller, M. (2003) Profiling aberrant DNA methylation in hematologic neoplasms: a view from the tip of the iceberg. *Clinical Immunology*, **109**, 80-88.
- Esteller, M. (2008) Epigenetics in Cancer. New England Journal of Medicine, 358, 1148-1159.
- Ezoe, S., Matsumura, I., Gale, K., Satoh, Y., Ishikawa, J., Mizuki, M., Takahashi, S., Minegishi, N., Nakajima, K., Yamamoto, M., Enver, T. & Kanakura, Y. (2005) GATA Transcription
 Factors Inhibit Cytokine-dependent Growth and Survival of a Hematopoietic Cell Line through the Inhibition of STAT3 Activity. *Journal of Biological Chemistry*, 280, 13163-13170.

- Falanga, A., Marchetti, M., Evangelista, V., Vignoli, A., Licini, M., Balicco, M., Manarini, S., Finazzi, G., Cerletti, C. & Barbui, T. (2000) Polymorphonuclear leukocyte activation and hemostasis in patients with essential thrombocythemia and polycythemia vera. *Blood*, 96, 4261-4266.
- Feinberg, A.P. & Vogelstein, B. (1983) Hypomethylation distinguishes genes of some human cancers from their normal counterparts. *Nature*, **301**, 89-92.
- Fialkow, P.J., Gartler, S.M. & Yoshida, A. (1967) Clonal origin of chronic myelocytic leukemia in man. *Proc Natl Acad Sci U S A*, **58**, 1468-1471.
- Fickers, M. & Speck, B. (1974) Thrombocythaemia. Acta Haematologica, 51, 257-265.
- Fielder, P., Gurney, A., Stefanich, E., Marian, M., Moore, M., Carver-Moore, K. & de Sauvage, F. (1996) Regulation of thrombopoietin levels by c-mpl-mediated binding to platelets. *Blood*, 87, 2154-2161.
- Finazzi, G., Ruggeri, M., Rodeghiero, F. & Barbui, T. (2000) Second malignancies in patients with essential thrombocythaemia treated with busulphan and hydroxyurea: long-term follow-up of a randomized clinical trial. *British Journal of Haematology*, **110**, 577-583.
- Finazzi, G., Rambaldi, A., Guerini, V., Carobbo, A. & Barbui, T. (2007) Risk of thrombosis in patients with essential thrombocythemia and polycythemia vera according to JAK2 V617F mutation status. *Haematologica*, **92**, 135-136.
- Florensa, L., Besses, C., Woessner, S., Sole, F., Acin, P., Pedro, C. & Sans-Sabrafen, J. (1995) Endogenous megakaryocyte and erythroid colony formation from blood in essential thrombocythaemia. *Leukemia*, 9, 271-273.
- Gaetani, G., Ferraris, A., Galiano, S., Giuntini, P., Canepa, L. & d'Urso, M. (1982) Primary thrombocythemia: clonal origin of platelets, erythrocytes, and granulocytes in a GdB/GdMediterranean subject. *Blood*, **59**, 76-79.
- Gaiger, A., Henn, T., Horth, E., Geissler, K., Mitterbauer, G., Maier-Dobersberger, T., Greinix, H., Mannhalter, C., Haas, O. & Lechner, K. (1995) Increase of bcr-abl chimeric mRNA expression in tumor cells of patients with chronic myeloid leukemia precedes disease progression. *Blood*, 86, 2371-2378.
- Gale, R. & Linch, D. (1994) Interpretation of X-chromosome inactivation patterns [letter; comment]. *Blood*, **84**, 2376-2378.
- Gale, R.E., Wheadon, H. & Linch, D.C. (1991) X-chromosome inactivation patterns using HPRT and PGK polymorphisms in haematologically normal and post-chemotherapy females. *British Journal of Haematology*, **79**, 193-197.
- Gale, R.E., Wheadon, H. & Linch, D.C. (1992) Assessment of X-chromosome inactivation patterns using the hypervariable probe M27 beta in normal hemopoietic cells and acute myeloid leukemic blasts. *Leukemia*, **6**, 649-655.
- Gale, R.E., Wheadon, H., Goldstone, A.H., Linch, D.C. & Burnett, A.K. (1993) Frequency of clonal remission in acute myeloid leukaemia. *The Lancet*, **341**, 138-142.
- Gale, R.E., Mein, C.A. & Linch, D.C. (1996) Quantification of X-chromosome inactivation patterns in haematological samples using the DNA PCR-based HUMARA assay. *Leukemia*, **10**, 362-367.
- Gale, R.E., Fielding, A.K., Harrison, C.N. & Linch, D.C. (1997) Acquired skewing of Xchromosome inactivation patterns in myeloid cells of the elderly suggests stochastic clonal loss with age. *British Journal of Haematology*, **98**, 512-519.

- Gale, R.E., Allen, A.J.R., Nash, M.J. & Linch, D.C. (2007) Long-term serial analysis of Xchromosome inactivation patterns and JAK2 V617F mutant levels in patients with essential thrombocythemia show that minor mutant-positive clones can remain stable for many years. *Blood*, **109**, 1241-1243.
- Gangat, N., Wolanskyj, A.P., McClure, R.F., Li, C.Y., Schwager, S., Wu, W. & Tefferi, A. (2007)
 Risk stratification for survival and leukemic transformation in essential
 thrombocythemia: a single institutional study of 605 patients. *Leukemia*, **21**, 270-276.
- Gangat, N., Tefferi, A., Thanarajasingam, G., Patnaik, M., Schwager, S., Ketterling, R. &
 Wolanskyj, A.P. (2009) Cytogenetic abnormalities in essential thrombocythemia:
 prevalence and prognostic significance. *European Journal of Haematology*, 83, 17-21.
- Geddis, A.E., Fox, N.E. & Kaushansky, K. (2001) Phosphatidylinositol 3-Kinase Is Necessary but Not Sufficient for Thrombopoietin-induced Proliferation in Engineered Mpl-bearing Cell Lines as Well as in Primary Megakaryocytic Progenitors. *Journal of Biological Chemistry*, **276**, 34473-34479.
- Geddis, A.E., Fox, N.E. & Kaushansky, K. (2006) The Mpl receptor expressed on endothelial cells does not contribute significantly to the regulation of circulating thrombopoietin levels. *Experimental Hematology*, **34**, 82-86.
- Gerrard, J.M., White, J.G., Rao, G.H. & Townsend, D. (1976) Localization of platelet prostaglandin production in the platelet dense tubular system. *Am J Pathol*, **83**, 283-298.
- Ghilardi, N., Wiestner, A. & Skoda, R.C. (1998) Thrombopoietin Production Is Inhibited by a Translational Mechanism. *Blood*, **92**, 4023-4030.
- Ghilardi, N., Wiestner, A., Kikuchi, M., Ohsaka, A. & Skoda, R.C. (1999) Hereditary thrombocythaemia in a Japanese family is caused by a novel point mutation in the thrombopoietin gene. *British Journal of Haematology*, **107**, 310-316.
- Gilliland, D.G., Blanchard, K.L., Levy, J., Perrin, S. & Bunn, H.F. (1991) Clonality in myeloproliferative disorders: analysis by means of the polymerase chain reaction. *Proc Natl Acad Sci U S A*, **88**, 6848-6852.
- Girodon, F., Schaeffer, C., Cleyrat, C., Mounier, M., Lafont, I., Santos, F.D., Vidal, A., Maynadie, M. & Hermouet, S. (2008) Frequent reduction or absence of detection of the JAK2-mutated clone in JAK2V617F-positive patients within the first years of hydroxyurea therapy. *Haematologica*, 93, 1723-1727.
- Girodon, F., Bonicelli, G., Schaeffer, C., Mounier, M., Carillo, S., Lafon, I., Carli, P.M., Janoray, I., Ferrant, E. & Maynadie, M. (2009) Significant increase in the apparent incidence of essential thrombocythemia related to new WHO diagnostic criteria: a populationbased study. *Haematologica*, **94**, 865-869.
- Glembotsky, A.C., Korin, L., Lev, P.R., Chazarreta, C.D., Marta, R.F., Molinas, F.C. & Heller, P.G.
 (2010) Screening for MPL mutations in essential thrombocythemia and primary
 myelofibrosis: normal Mpl expression and absence of constitutive STAT3 and STAT5
 activation in MPL W515L-positive platelets. *European Journal of Haematology*, 9999.
- Goerttler, P.S., Steimle, C., Marz, E., Johansson, P.L., Andreasson, B., Griesshammer, M., Gisslinger, H., Heimpel, H. & Pahl, H.L. (2005) The Jak2V617F mutation, PRV-1 overexpression, and EEC formation define a similar cohort of MPD patients. *Blood*, **106**, 2862-2864.
- Gregg, X.T., Kralovics, R. & Prchal, J.T. (2000) A polymorphism of the X-linked gene IDS increases the number of females informative for transcriptional clonality assays. *Am J Hematol*, **63**, 184-191.

- Griesshammer, M., Hornkohl, A., Nichol, J.L., Hecht, T., Raghavachar, A., Heimpel, H. & Schrezenmeier, H. (1998) High levels of thrombopoietin in sera of patients with essential thrombocythemia: Cause or consequence of abnormal platelet production? *Annals of Hematology*, **77**, 211-215.
- Griesshammer, M., Klippel, S., Strunck, E., Temerinac, S., Mohr, U., Heimpel, H. & Pahl, H.L.
 (2004) PRV-1 mRNA expression discriminates two types of essential thrombocythemia. Annals of Hematology, 83, 364-370.
- Guglielmelli, P., Biamonte, F., Spolverini, A., Pieri, L., Isgro, A., Antonioli, E., Pancrazzi, A., Bosi,
 A., Barosi, G. & Vannucchi, A.M. (2010) Frequency and clinical correlates of JAK2 46/1 (GGCC) haplotype in primary myelofibrosis. *Leukemia*, 24, 1533-1537.
- Haan, C., Is'harc, H., Hermanns, H.M., Schmitz-Van de Leur, H., Kerr, I.M., Heinrich, P.C., Grötzinger, J. & Behrmann, I. (2001) Mapping of a Region within the N Terminus of Jak1 Involved in Cytokine Receptor Interaction. *Journal of Biological Chemistry*, 276, 37451-37458.
- Haan, C., Heinrich, P.C. & Behrmann, I. (2002) Structural requirements of the interleukin-6 signal transducer gp130 for its interaction with Janus kinase 1: the receptor is crucial for kinase activation. *Biochem. J.*, **361**, 105-111.
- Hamelin, V., Letourneux, C., Romeo, P.-H., Porteu, F. & Gaudry, M. (2006) Thrombopoietin regulates IEX-1 gene expression through ERK-induced AML1 phosphorylation. *Blood*, **107**, 3106-3113.
- Harrison, C.N., Gale, R.E., Wiestner, A.C., Skoda, R.C. & Linch, D.C. (1998) The activating splice mutation in intron 3 of the thrombopoietin gene is not found in patients with non-familial essential thrombocythaemia. *British Journal of Haematology*, **102**, 00-00.
- Harrison, C.N., Gale, R.E., Machin, S.J. & Linch, D.C. (1999a) A large proportion of patients with a diagnosis of essential thrombocythemia do not have a clonal disorder and may be at lower risk of thrombotic complications. *Blood*, **93**, 417-424.
- Harrison, C.N., Gale, R.E., Pezella, F., Mire-Sluis, A., Machin, S.J. & Linch, D.C. (1999b) Platelet c-mpl expression is dysregulated in patients with essential thrombocythaemia but this is not of diagnostic value. *British Journal of Haematology*, **107**, 139-147.
- Harrison, C.N., Campbell, P.J., Buck, G., Wheatley, K., East, C.L., Bareford, D., Wilkins, B.S., van der Walt, J.D., Reilly, J.T., Grigg, A.P., Revell, P., Woodcock, B.E., Green, A.R. & the United Kingdom Medical Research Council Primary Thrombocythemia 1 Study (2005) Hydroxyurea Compared with Anagrelide in High-Risk Essential Thrombocythemia. *N Engl J Med*, **353**, 33-45.
- Harrison, C.N., Bareford, D., Butt, N., Campbell, P., Conneally, E., Drummond, M., Erber, W., Everington, T., Green, A.R., Hall, G.W., Hunt, B.J., Ludlam, C.A., Murrin, R., Nelson-Piercy, C., Radia, D.H., Reilly, J.T., Van der Walt, J., Wilkins, B., McMullin, M.F. & British Committee for Standards in, H. (2010) Guideline for investigation and management of adults and children presenting with a thrombocytosis. *British Journal of Haematology*, 149, 352-375.
- Hehlmann, R., Jahn, M., Baumann, B. & Kopcke, W. (1988) Essential thrombocythemia. Clinical characteristics and course of 61 cases. *Cancer*, **61**, 2487-2496.
- Heijnen, H.F., Debili, N., Vainchencker, W., Breton-Gorius, J., Geuze, H.J. & Sixma, J.J. (1998) Multivesicular bodies are an intermediate stage in the formation of platelet alphagranules. *Blood*, **91**, 2313-2325.
- Heller, P.G., Lev, P.R., Salim, J.P., Kornblihtt, L.I., Goette, N.P., Chazarreta, C.D., Glembotsky, A.C., Vassallu, P.S., Marta, R.F. & Molinas, F.C. (2006) JAK2 V617F mutation in platelets

from essential thrombocythemia patients: correlation with clinical features and analysis of STAT5 phosphorylation status. *European Journal of Haematology*, **77**, 210-216.

- Hirayama, Y., Sakamaki, S., Matsunaga, T., Kuga, T., Kuroda, H., Kusakabe, T., Sasaki, K.,
 Fujikawa, K., Kato, J., Kogawa, K., Koyama, R. & Niitsu, Y. (1998) Concentrations of
 thrombopoietin in bone marrow in normal subjects and in patients with idiopathic
 thrombocytopenic purpura, aplastic anemia, and essential thrombocythemia correlate
 with its mRNA expression of bone marrow stromal cells. *Blood*, **92**, 46-52.
- Hoffbrand, A.V., Catovsky, D. & Tuddenham, E.G.D. (2005) *Postgraduate haematology*. Blackwell, Oxford.
- Hollen, C.W., Henthorn, J., Koziol, J.A. & Burstein, S.A. (1991) Elevated serum interleukin-6 levels in patients with reactive thrombocytosis. *British Journal of Haematology*, **79**, 286-290.
- Horan, T., Wen, J., Narhi, L., Parker, V., Garcia, A., Arakawa, T. & Philo, J. (1996) Dimerization of the Extracellular Domain of Granulocyte-Colony Stimulating Factor Receptor by Ligand Binding: A Monovalent Ligand Induces 2:2 Complexes. *Biochemistry*, **35**, 4886-4896.
- Horikawa, Y., Matsumura, I., Hashimoto, K., Shiraga, M., Kosugi, S., Tadokoro, S., Kato, T., Miyazaki, H., Tomiyama, Y., Kurata, Y., Matsuzawa, Y. & Kanakura, Y. (1997) Markedly Reduced Expression of Platelet c-mpl Receptor in Essential Thrombocythemia. *Blood*, 90, 4031-4038.
- Houwerzijl, E., Louwes, H., Esselink, M., Smit, J., Vellenga, E. & de Wolf, J. (2005) Increased glycocalicin index and normal thrombopoietin levels in patients with idiopathic thrombocytopenic purpura with a decreased rate of platelet production. *Haematologica*, **90**, 710-711.
- Hsiao, H.H., Yang, M.Y., Liu, Y.C., Lee, C.P., Yang, W.C., Liu, T.C., Chang, C.S. & Lin, S.F. (2007) The association of JAK2V617F mutation and leukocytosis with thrombotic events in essential thrombocythemia. *Exp Hematol*, **35**, 1704-1707.
- Hussein, K., Bock, O., Theophile, K., von Neuhoff, N., Buhr, T., Schlué, J., Büsche, G. & Kreipe, H. (2009) JAK2V617F allele burden discriminates essential thrombocythemia from a subset of prefibrotic-stage primary myelofibrosis. *Experimental Hematology*, **37**, 1186-1193.e1187.
- Ishibashi, T., Miller, S. & Burstein, S. (1987) Type beta transforming growth factor is a potent inhibitor of murine megakaryocytopoiesis in vitro. *Blood*, **69**, 1737-1741.
- Jaffe, E.S. (2001) Pathology and genetics of tumours of haematopoietic and lymphoid tissues. IARC Press ; Oxford : Oxford University Press [distributor], Lyon.
- James, C., Ugo, V., Le Couedic, J.-P., Staerk, J., Delhommeau, F., Lacout, C., Garcon, L., Raslova, H., Berger, R., Bennaceur-Griscelli, A., Villeval, J.L., Constantinescu, S.N., Casadevall, N. & Vainchenker, W. (2005) A unique clonal JAK2 mutation leading to constitutive signalling causes polycythaemia vera. *Nature*, 434, 1144-1148.
- Janssen, J., Anger, B., Drexler, H., Bartram, C. & Heimpel, H. (1990) Essential thrombocythemia in two sisters originating from different stem cell levels. *Blood*, **75**, 1633-1636.
- Jantunen, R., Juvonen, E., Ikkala, E., Oksanen, K., Anttila, P. & Ruutu, T. (2001) The predictive value of vascular risk factors and gender for the development of thrombotic complications in essential thrombocythemia. *Annals of Hematology*, **80**, 74-78.
- Jelinek, J., Oki, Y., Gharibyan, V., Bueso-Ramos, C., Prchal, J.T., Verstovsek, S., Beran, M., Estey, E., Kantarjian, H.M. & Issa, J.P. (2005) JAK2 mutation 1849G>T is rare in acute

leukemias but can be found in CMML, Philadelphia chromosome-negative CML, and megakaryocytic leukemia. *Blood*, **106**, 3370-3373.

- Jensen, M.K., Brown, P.d.N., Nielsen, O.J. & Hasselbalch, H.C. (2000) Incidence, clinical features and outcome of essential thrombocythaemia in a well defined geographical area. *European Journal of Haematology*, **65**, 132-139.
- Johansson, P., Safai-Kutti, S., Lindstedt, G., Suurküla, M. & Kutti, J. (2002) Red Cell Mass, Spleen Size and Plasma Erythropoietin in Polycythaemia vera and Apparent Polycythaemia. *Acta Haematologica*, **108**, 1-7.
- Johansson, P., Kutti, J., Andréasson, B., Safai-Kutti, S., Vilén, L., Wedel, H. & Ridell, B. (2004) Trends in the incidence of chronic Philadelphia chromosome negative (Ph-) myeloproliferative disorders in the city of Göteborg, Sweden, during 1983–99. *Journal of Internal Medicine*, **256**, 161-165.
- Jones, A.V., Kreil, S., Zoi, K., Waghorn, K., Curtis, C., Zhang, L., Score, J., Seear, R., Chase, A.J., Grand, F.H., White, H., Zoi, C., Loukopoulos, D., Terpos, E., Vervessou, E.-C., Schultheis, B., Emig, M., Ernst, T., Lengfelder, E., Hehlmann, R., Hochhaus, A., Oscier, D., Silver, R.T., Reiter, A. & Cross, N.C.P. (2005) Widespread occurrence of the JAK2 V617F mutation in chronic myeloproliferative disorders. *Blood*, **106**, 2162-2168.
- Jones, A.V., Chase, A., Silver, R.T., Oscier, D., Zoi, K., Wang, Y.L., Cario, H., Pahl, H.L., Collins, A., Reiter, A., Grand, F. & Cross, N.C.P. (2009) JAK2 haplotype is a major risk factor for the development of myeloproliferative neoplasms. *Nature Genetics*, **41**, 446-449.
- Jones, A.V., Campbell, P.J., Beer, P.A., Schnittger, S., Vannucchi, A.M., Zoi, K., Percy, M.J.,
 McMullin, M.F., Scott, L.M., Tapper, W., Silver, R.T., Oscier, D., Harrison, C.N., Grallert,
 H., Kisialiou, A., Strike, P., Chase, A.J., Green, A.R. & Cross, N.C.P. (2010) The JAK2 46/1
 haplotype predisposes to MPL-mutated myeloproliferative neoplasms. *Blood*, 115, 4517-4523.
- Jorgensen, M., Raskind, W., Wolff, J. & Bachrach, H. (1998) Familial thrombocytosis associated with overproduction of thrombopoietin due to a novel splice donor site mutation. *Blood*, **92**, 205a (abstract).
- Juvonen, E., Partanen, S. & Ruutu, T. (1987) Colony formation by megakaryocytic progenitors in essential thrombocythaemia. *British Journal of Haematology*, **66**, 161-164.
- Kaser, A., Brandacher, G., Steurer, W., Kaser, S., Offner, F.A., Zoller, H., Theurl, I., Widder, W., Molnar, C., Ludwiczek, O., Atkins, M.B., Mier, J.W. & Tilg, H. (2001) Interleukin-6 stimulates thrombopoiesis through thrombopoietin: role in inflammatory thrombocytosis. *Blood*, **98**, 2720-2725.
- Kassar, N.E., Hetet, G., Li, Y., Briere, J. & Grandchamp, B. (1995) Clonal analysis of haemopoietic cells in essential thrombocythaemia. *British Journal of Haematology*, 90, 131-137.
- Kaushansky, K., Lok, S., Holly, R.D., Broudy, V.C., Lin, N., Bailey, M.C., Forstrom, J.W., Buddle, M.M., Oort, P.J., Hagen, F.S., Roth, G.J., Papayannopoulou, T. & Foster, D.C. (1994)
 Promotion of megakaryocyte progenitor expansion and differentiation by the c-Mpl ligand thrombopoietin. *Nature*, 369, 568-571.
- Kaushansky, K. (2005) The molecular mechanisms that control thrombopoiesis. *The Journal of Clinical Investigation*, **115**, 3339-3347.
- Kaushansky, K. (2009) Molecular mechanisms of thrombopoietin signaling. *J Thromb Haemost,* **7 Suppl 1,** 235-238.

- Kawamata, N., Ogawa, S., Yamamoto, G., Lehmann, S., Levine, R.L., Pikman, Y., Nannya, Y., Sanada, M., Miller, C.W., Gilliland, D.G. & Koeffler, H.P. (2008) Genetic profiling of myeloproliferative disorders by single-nucleotide polymorphism oligonucleotide microarray. *Experimental Hematology*, **36**, 1471-1479.
- Kawasaki, H., Nakano, T., Kohdera, U. & Kobayashi, Y. (2001) Hypersensitivity of megakaryocyte progenitors to thrombopoietin in essential thrombocythemia. *American Journal of Hematology*, **68**, 194-197.
- Kiladjian, J.-J., Cassinat, B., Chevret, S., Turlure, P., Cambier, N., Roussel, M., Bellucci, S., Grandchamp, B., Chomienne, C. & Fenaux, P. (2008) Pegylated interferon-alfa-2a induces complete hematologic and molecular responses with low toxicity in polycythemia vera. *Blood*, **112**, 3065-3072.
- Kiladjian, J.J., Elkassar, N., Hetet, G., Briere, J., Grandchamp, B. & Gardin, C. (1997) Study of the thrombopoitin receptor in essential thrombocythemia. *Leukemia*, **11**, 1821-1826.
- Kiladjian, J.J., Elkassar, N., Cassinat, B., Hetet, G., Giraudier, S., Balitrand, N., Conejero, C., Briere, J., Fenaux, P., Chomienne, C. & Grandchamp, B. (2006) Essential thrombocythemias without V617F JAK2 mutation are clonal hematopoietic stem cell disorders. *Leukemia*, **20**, 1181-1183.
- Kilpivaara, O., Mukherjee, S., Schram, A.M., Wadleigh, M., Mullally, A., Ebert, B.L., Bass, A., Marubayashi, S., Heguy, A., Garcia-Manero, G., Kantarjian, H., Offit, K., Stone, R.M., Gilliland, D.G., Klein, R.J. & Levine, R.L. (2009) A germline JAK2 SNP is associated with predisposition to the development of JAK2V617F-positive myeloproliferative neoplasms. *Nature Genetics*, **41**, 455-459.
- Kirito, K., Osawa, M., Morita, H., Shimizu, R., Yamamoto, M., Oda, A., Fujita, H., Tanaka, M., Nakajima, K., Miura, Y., Ozawa, K. & Komatsu, N. (2002a) A functional role of Stat3 in in vivo megakaryopoiesis. *Blood*, **99**, 3220-3227.
- Kirito, K., Watanabe, T., Sawada, K.-i., Endo, H., Ozawa, K. & Komatsu, N. (2002b) Thrombopoietin Regulates Bcl-xL Gene Expression through Stat5 and Phosphatidylinositol 3-Kinase Activation Pathways. *Journal of Biological Chemistry*, 277, 8329-8337.
- Kirito, K., Fox, N., Komatsu, N. & Kaushansky, K. (2005) Thrombopoietin enhances expression of vascular endothelial growth factor (VEGF) in primitive hematopoietic cells through induction of HIF-1{alpha}. *Blood*, **105**, 4258-4263.
- Kishimoto, T. (1989) The biology of interleukin-6. *Blood*, **74**, 1-10.
- Kittur, J., Knudson, R.A., Lasho, T.L., Finke, C.M., Gangat, N., Wolanskyj, A.P., Li, C.-Y., Wu, W., Ketterling, R.P., Pardanani, A. & Tefferi, A. (2007) Clinical correlates of JAK2 V617F allele burden in essential thrombocythemia. *Cancer*, **109**, 2279-2284.
- Klippel, S., Strunck, E., Busse, C.E., Behringer, D. & Pahl, H.L. (2002) Biochemical characterization of PRV-1, a novel hematopoietic cell surface receptor, which is overexpressed in polycythemia rubra vera. *Blood*, **100**, 2441-2448.
- Kojima, H., Shinagawa, A., Shimizu, S., Kanada, H., Hibi, M., Hirano, T. & Nagasawa, T. (2001)
 Role of phosphatidylinositol-3 kinase and its association with Gab1 in thrombopoietinmediated up-regulation of platelet function. *Experimental Hematology*, 29, 616-622.
- Kosugi, S., Kurata, Y., Omiyama, Y.T., Ahara, T.T., Kato, T., Tadokoro, S., Shiraga, M., Onda, S.H.,
 Kanakura, Y. & Matsuzawa, Y. (1996) Circulating thrombopoietin level in chronic
 immune thrombocytopenic purpura. *British Journal of Haematology*, 93, 704-706.

- Kralovics, R., Guan, Y. & Prchal, J.T. (2002) Acquired uniparental disomy of chromosome 9p is a frequent stem cell defect in polycythemia vera. *Exp Hematol*, **30**, 229-236.
- Kralovics, R., Buser, A.S., Teo, S.-S., Coers, J., Tichelli, A., van der Maas, A.P.C. & Skoda, R.C.
 (2003) Comparison of molecular markers in a cohort of patients with chronic myeloproliferative disorders. *Blood*, **102**, 1869-1871.
- Kralovics, R., Passamonti, F., Buser, A.S., Teo, S.-S., Tiedt, R., Passweg, J.R., Tichelli, A., Cazzola, M. & Skoda, R.C. (2005) A Gain-of-Function Mutation of JAK2 in Myeloproliferative Disorders. N Engl J Med, 352, 1779-1790.
- Kralovics, R., Teo, S.-S., Li, S., Theocharides, A., Buser, A.S., Tichelli, A. & Skoda, R.C. (2006) Acquisition of the V617F mutation of JAK2 is a late genetic event in a subset of patients with myeloproliferative disorders. *Blood*, **108**, 1377-1380.
- Kuter, D. & Rosenberg, R. (1995) The reciprocal relationship of thrombopoietin (c-Mpl ligand) to changes in the platelet mass during busulfan-induced thrombocytopenia in the rabbit. *Blood*, **85**, 2720-2730.
- Landgren, O., Goldin, L.R., Kristinsson, S.Y., Helgadottir, E.A., Samuelsson, J. & Bjorkholm, M. (2008) Increased risks of polycythemia vera, essential thrombocythemia, and myelofibrosis among 24 577 first-degree relatives of 11 039 patients with myeloproliferative neoplasms in Sweden. *Blood*, **112**, 2199-2204.
- Landolfi, R., Marchioli, R., Kutti, J., Gisslinger, H., Tognoni, G., Patrono, C., Barbui, T. & the European Collaboration on Low-Dose Aspirin in Polycythemia Vera Investigators (2004)
 Efficacy and Safety of Low-Dose Aspirin in Polycythemia Vera. N Engl J Med, 350, 114-124.
- Landolfi, R., Di Gennaro, L., Barbui, T., De Stefano, V., Finazzi, G., Marfisi, R., Tognoni, G., Marchioli, R. & for the European Collaboration on Low-Dose Aspirin in Polycythemia Vera (2007) Leukocytosis as a major thrombotic risk factor in patients with polycythemia vera. *Blood*, **109**, 2446-2452.
- Lane, W.J., Dias, S., Hattori, K., Heissig, B., Choy, M., Rabbany, S.Y., Wood, J., Moore, M.A. & Rafii, S. (2000) Stromal-derived factor 1-induced megakaryocyte migration and platelet production is dependent on matrix metalloproteinases. *Blood*, **96**, 4152-4159.
- Larsen, T.S., Pallisgaard, N., Møller, M.B. & Hasselbalch, H.C. (2007) The JAK2 V617F allele burden in essential thrombocythemia, polycythemia vera and primary myelofibrosis – impact on disease phenotype. *European Journal of Haematology*, **79**, 508-515.
- Larsen, T.S., Pallisgaard, N., Stricker, K.d., Møller, M.B. & CarlHasselbalch, H. (2009) Limited efficacy of hydroxyurea in lowering of the JAK2 V617F allele burden. *Hematology*, **14**, 11-15.
- Lecine, P., Villeval, J.-L., Vyas, P., Swencki, B., Xu, Y. & Shivdasani, R.A. (1998) Mice Lacking Transcription Factor NF-E2 Provide In Vivo Validation of the Proplatelet Model of Thrombocytopoiesis and Show a Platelet Production Defect That Is Intrinsic to Megakaryocytes. *Blood*, **92**, 1608-1616.
- Lengfelder, E., Griesshammer, M. & Hehlmann, R. (1996) Interferon-alpha in the Treatment of Essential Thrombocythemia. *Leukemia & Lymphoma*, **22**, 135-142.
- Levine, R.L., Wadleigh, M., Cools, J., Ebert, B.L., Wernig, G., Huntly, B.J.P., Boggon, T.J.,
 Wlodarska, I., Clark, J.J., Moore, S., Adelsperger, J., Koo, S., Lee, J.C., Gabriel, S.,
 Mercher, T., D'Andrea, A., Fröhling, S., Döhner, K., Marynen, P., Vandenberghe, P.,
 Mesa, R.A., Tefferi, A., Griffin, J.D., Eck, M.J., Sellers, W.R., Meyerson, M., Golub, T.R.,
 Lee, S.J. & Gilliland, D.G. (2005) Activating mutation in the tyrosine kinase JAK2 in

polycythemia vera, essential thrombocythemia, and myeloid metaplasia with myelofibrosis. *Cancer Cell*, **7**, 387-397.

- Levine, R.L., Belisle, C., Wadleigh, M., Zahrieh, D., Lee, S., Chagnon, P., Gilliland, D.G. & Busque,
 L. (2006) X-inactivation-based clonality analysis and quantitative JAK2V617F
 assessment reveal a strong association between clonality and JAK2V617F in PV but not
 ET/MMM, and identifies a subset of JAK2V617F-negative ET and MMM patients with
 clonal hematopoiesis. *Blood*, **107**, 4139-4141.
- Li, J., Xia, Y. & Kuter, D.J. (2000) The platelet thrombopoietin receptor number and function are markedly decreased in patients with essential thrombocythaemia. *British Journal of Haematology*, **111**, 943-953.
- Lieu, C.H., Wu, H.S., Hon, Y.C., Tsai, W.H., Yang, C.F., Wang, C.C., Lin, Y.C., Shih, C.H. & Hsu, H.C.
 (2008) Prevalence of the JAK2-V617F mutation in Taiwanese patients with chronic myeloproliferative disorders. *Internal Medicine Journal*, 38, 422-426.
- Lindauer, K., Loerting, T., Liedl, K.R. & Kroemer, R.T. (2001) Prediction of the structure of human Janus kinase 2 (JAK2) comprising the two carboxy-terminal domains reveals a mechanism for autoregulation. *Protein Eng.*, **14**, 27-37.
- Lippert, E., Boissinot, M., Kralovics, R., Girodon, F., Dobo, I., Praloran, V., Boiret-Dupre, N., Skoda, R.C. & Hermouet, S. (2006) The JAK2-V617F mutation is frequently present at diagnosis in patients with essential thrombocythemia and polycythemia vera. *Blood*, **108**, 1865-1867.
- Liu, E., Jelinek, J., Pastore, Y.D., Guan, Y., Prchal, J.F. & Prchal, J.T. (2003) Discrimination of polycythemias and thrombocytoses by novel, simple, accurate clonality assays and comparison with PRV-1 expression and BFU-E response to erythropoietin. *Blood*, **101**, 3294-3301.
- Lok, S., Kaushansky, K., Holly, R.D., Kuijper, J.L., Lofton-Day, C.E., Oort, P.J., Grant, F.J., Heipel, M.D., Burkhead, S.K., Kramer, J.M. & et al. (1994) Cloning and expression of murine thrombopoietin cDNA and stimulation of platelet production in vivo. *Nature*, **369**, 565-568.
- Lucas, G.S., Padua, R.A., Masters, G.S., Oscier, D.G. & Jacobs, A. (1989) The application of Xchromosome gene probes to the diagnosis of myeloproliferative disease. *British Journal of Haematology*, **72**, 530-533.
- Luhovy, M., Liu, Y., Belickova, M., Prchal, J.F. & Prchal, J.T. (1995) A novel clonality assay based on transcriptional polymorphism of X chromosome gene p55. *Biol Blood Marrow Transplant*, **1**, 81-87.
- Lussana, F., Caberlon, S., Pagani, C., Kamphuisen, P.W., Büller, H.R. & Cattaneo, M. (2009) Association of V617F Jak2 mutation with the risk of thrombosis among patients with essential thrombocythaemia or idiopathic myelofibrosis: A systematic review. *Thrombosis Research*, **124**, 409-417.
- Lyon, M.F. (1961) Gene Action in the X-chromosome of the Mouse (Mus musculus L.). *Nature*, **190**, 372-373.
- McNally, R.J.Q., Rowland, D., Roman, E. & Cartwright, R.A. (1997) Age and sex distributions of hematological malignancies in the U.K. *Hematological Oncology*, **15**, 173-189.
- Mesa, R.A., Silverstein, M.N., Jacobsen, S.J., Wollan, P.C. & Tefferi, A. (1999) Population-based incidence and survival figures in essential thrombocythemia and agnogenic myeloid metaplasia: An Olmsted county study, 1976-1995. *American Journal of Hematology*, 61, 10-15.
- Mesa, R.A., Hanson, C.A., Li, C.-Y., Yoon, S.-Y., Rajkumar, S.V., Schroeder, G. & Tefferi, A. (2002) Diagnostic and prognostic value of bone marrow angiogenesis and megakaryocyte c-Mpl expression in essential thrombocythemia. *Blood*, **99**, 4131-4137.
- Messinezy, M., Westwood, N.B., Woodcock, S.P., Strong, R.M. & Pearson, T.C. (1995) Low serum erythropoietin--a strong diagnostic criterion of primary polycythaemia even at normal haemoglobin levels. *Clin Lab Haematol*, **17**, 217-220.
- Mi, J.Q., Blanc-Jouvan, F., Wang, J., Sotto, M.F., Cousin, F., Castinel, A., Chauvet, M., Sotto, J.J., Polack, B. & Mossuz, P. (2001) Endogenous Megakaryocytic Colony Formation and Thrombopoietin Sensitivity of Megakaryocytic Progenitor Cells Are Useful to Distinguish Between Essential Thrombocythemia and Reactive Thrombocytosis. Journal of Hematotherapy & Stem Cell Research, 10, 405-409.
- Michiels, J., Commandeur, S., Hoogenboom, G., Wegman, J., Scholten, L., van Rijssel, R. & De Raeve, H. (2007) JAK2 V617F positive early stage myeloproliferative disease (essential thrombocythemia) as the cause of portal vein thrombosis in two middle-aged women: therapeutic implications in view of the literature. *Annals of Hematology*, **86**, 793-800.
- Miyakawa, Y., Rojnuckarin, P., Habib, T. & Kaushansky, K. (2001) Thrombopoietin Induces Phosphoinositol 3-Kinase Activation through SHP2, Gab, and Insulin Receptor Substrate Proteins in BAF3 Cells and Primary Murine Megakaryocytes. *Journal of Biological Chemistry*, **276**, 2494-2502.
- Mohandas, T., Sparkes, R. & Shapiro, L. (1981) Reactivation of an inactive human X chromosome: evidence for X inactivation by DNA methylation. *Science*, **211**, 393-396.
- Moliterno, A.R., Hankins, W.D. & Spivak, J.L. (1998) Impaired Expression of the Thrombopoietin Receptor by Platelets from Patients with Polycythemia Vera. *N Engl J Med*, **338**, 572-580.
- Moliterno, A.R., Williams, D.M., Gutierrez-Alamillo, L.I., Salvatori, R., Ingersoll, R.G. & Spivak, J.L. (2004) Mpl Baltimore: A thrombopoietin receptor polymorphism associated with thrombocytosis. *Proceedings of the National Academy of Sciences of the United States of America*, **101**, 11444-11447.
- Moliterno, A.R., Williams, D.M., Rogers, O. & Spivak, J.L. (2006) Molecular mimicry in the chronic myeloproliferative disorders: reciprocity between quantitative JAK2 V617F and Mpl expression. *Blood*, **108**, 3913-3915.
- Mossuz, P., Girodon, F., Donnard, M., Latger-Cannard, V., Dobo, I., Boiret, N., Lecron, J., Binquet, C., Barro, C., Hermouet, S. & Praloran, V. (2004) Diagnostic value of serum erythropoietin level in patients with absolute erythrocytosis. *Haematologica*, **89**, 1194-1198.
- Muheim, R., Buterin, T., Colgate, K.C., Kolbanovsij, A., Geacintov, N.E. & Naegeli, H. (2003)
 Modulation of Human Nucleotide Excision Repair by 5-Methylcytosines⁺. *Biochemistry*, 42, 3247-3254.
- Mullighan, C.G. (2009) TET2 mutations in myelodysplasia and myeloid malignancies. *Nature Genetics*, **41**, 766-767.
- Muntean, A.G. & Crispino, J.D. (2005) Differential requirements for the activation domain and FOG-interaction surface of GATA-1 in megakaryocyte gene expression and development. *Blood*, **106**, 1223-1231.
- Murphy, S., Iland, H., Rosenthal, D. & Laszlo, J. (1986) Essential thrombocythemia: an interim report from the Polycythemia Vera Study Group. *Semin Hematol*, **23**, 177-182.

- Murphy, S., Peterson, P., Iland, H. & Laszlo, J. (1997) Experience of the Polycythemia Vera Study Group with essential thrombocythemia: a final report on diagnostic criteria, survival, and leukemic transition by treatment. *Semin Hematol*, **34**, 29-39.
- Nakao, K. & Angrist, A.A. (1968) Membrane Surface Specialization of Blood Platelet and Megakaryocyte. *Nature*, **217**, 960-961.
- Nakao, T., Geddis, A.E., Fox, N.E. & Kaushansky, K. (2008) PI3K/Akt/FOXO3a pathway contributes to thrombopoietin-induced proliferation of primary megakaryocytes in vitro and in vivo via modulation of p27(Kip1). *Cell Cycle*, **7**, 257-266.
- Nowell, P.C. & Hungerford, D.A. (1960) Chromosome studies on normal and leukemic human leukocytes. *J Natl Cancer Inst*, **25**, 85-109.
- Ohyashiki, K., Ito, Y., Hori, K., Sato, K., Makino, T. & Ohyashiki, J.H. (2007) Thrombosis can occur at any phase of essential thrombocythemia with JAK2V617F mutation: a single institutional study in Japan. *Leukemia*, **21**, 1570-1571.
- Olcaydu, D., Harutyunyan, A., Jager, R., Berg, T., Gisslinger, B., Pabinger, I., Gisslinger, H. & Kralovics, R. (2009) A common JAK2 haplotype confers susceptibility to myeloproliferative neoplasms. *Nature Genetics*, **41**, 450-454.
- Olcaydu, D., Rumi, E., Harutyunyan, A., Passamonti, F., Pietra, D., Pascutto, C., Berg, T., Jager, R., Hammond, E., Cazzola, M. & Kralovics, R. (2010) The role of the JAK2 GGCC haplotype and the TET2 gene in familial myeloproliferative neoplasms. *Haematologica*, haematol.2010.034488.
- Palandri, F., Catani, L., Testoni, N., Ottaviani, E., Polverelli, N., Fiacchini, M., Vivo, A.D., Salmi,
 F., Lucchesi, A., Baccarani, M. & Vianelli, N. (2009a) Long-term follow-up of 386
 consecutive patients with essential thrombocythemia: Safety of cytoreductive therapy.
 American Journal of Hematology, 84, 215-220.
- Palandri, F., Ottaviani, E., Salmi, F., Catani, L., Polverelli, N., Fiacchini, M., Martinelli, G., Baccarani, M. & Vianelli, N. (2009b) JAK2V617F mutation in essential thrombocythemia: correlation with clinical characteristics, response to therapy and long-term outcome in a cohort of 275 patients. *Leukemia & Lymphoma*, **50**, 247-253.
- Panteli, K.E., Hatzimichael, E.C., Bouranta, P.K., Katsaraki, A., Seferiadis, K., Stebbing, J. & Bourantas, K.L. (2005) Serum interleukin (IL)-1, IL-2, sIL-2Ra, IL-6 and thrombopoietin levels in patients with chronic myeloproliferative diseases. *British Journal of Haematology*, **130**, 709-715.
- Pardanani, A., Lasho, T.L., Finke, C., Mesa, R.A., Hogan, W.J., Ketterling, R.P., Gilliland, D.G. & Tefferi, A. (2007) Extending Jak2V617F and MpIW515 mutation analysis to single hematopoietic colonies and B and T lymphocytes. *Stem Cells*, **25**, 2358-2362.
- Pardanani, A., Fridley, B.L., Lasho, T.L., Gilliland, D.G. & Tefferi, A. (2008) Host genetic variation contributes to phenotypic diversity in myeloproliferative disorders. *Blood*, **111**, 2785-2789.
- Pardanani, A., Lasho, T.L., Finke, C.M., Gangat, N., Wolanskyj, A.P., Hanson, C.A. & Tefferi, A. (2010) The JAK2 46/1 haplotype confers susceptibility to essential thrombocythemia regardless of JAK2V617F mutational status-clinical correlates in a study of 226 consecutive patients. *Leukemia*, **24**, 110-114.
- Pardanani, A., Gotlib, J.R., Jamieson, C., Cortes, J.E., Talpaz, M., Stone, R.M., Silverman, M.H., Gilliland, D.G., Shorr, J. & Tefferi, A. (2011) Safety and Efficacy of TG101348, a Selective JAK2 Inhibitor, in Myelofibrosis. *Journal of Clinical Oncology*, **29**, 789-796.

- Pardanani, A.D., Levine, R.L., Lasho, T., Pikman, Y., Mesa, R.A., Wadleigh, M., Steensma, D.P., Elliott, M.A., Wolanskyj, A.P., Hogan, W.J., McClure, R.F., Litzow, M.R., Gilliland, D.G. & Tefferi, A. (2006) MPL515 mutations in myeloproliferative and other myeloid disorders: a study of 1182 patients. *Blood*, **108**, 3472-3476.
- Parker, C.J. (2007) The pathophysiology of paroxysmal nocturnal hemoglobinuria. *Exp Hematol*, **35**, 523-533.
- Partanen, S., Ruutu, T. & Vuopio, P. (1983) Haematopoietic Progenitors in Essential Thrombocythaemia. *Scandinavian Journal of Haematology*, **30**, 130-134.
- Passamonti, F., Rumi, E., Pungolino, E., Malabarba, L., Bertazzoni, P., Valentini, M., Orlandi, E., Arcaini, L., Brusamolino, E., Pascutto, C., Cazzola, M., Morra, E. & Lazzarino, M. (2004)
 Life expectancy and prognostic factors for survival in patients with polycythemia vera and essential thrombocythemia. *The American Journal of Medicine*, **117**, 755-761.
- Passamonti, F., Rumi, E., Pietra, D., Porta, M.G.D., Boveri, E., Pascutto, C., Vanelli, L., Arcaini, L., Burcheri, S., Malcovati, L., Lazzarino, M. & Cazzola, M. (2006) Relation between JAK2 (V617F) mutation status, granulocyte activation, and constitutive mobilization of CD34+ cells into peripheral blood in myeloproliferative disorders. *Blood*, **107**, 3676-3682.
- Passamonti, F., Rumi, E., Arcaini, L., Boveri, E., Elena, C., Pietra, D., Boggi, S., Astori, C., Bernasconi, P., Varettoni, M., Brusamolino, E., Pascutto, C. & Lazzarino, M. (2008)
 Prognostic factors for thrombosis, myelofibrosis, and leukemia in essential thrombocythemia: a study of 605 patients. *Haematologica*, **93**, 1645-1651.
- Patel, S.R., Hartwig, J.H. & Italiano, J.E. (2005) The biogenesis of platelets from megakaryocyte proplatelets. *The Journal of Clinical Investigation*, **115**, 3348-3354.
- Patnaik, M.M., Lasho, T.L., Finke, C.M., Gangat, N., Caramazza, D., Siragusa, S., Hanson, C.A., Pardanani, A. & Tefferi, A. (2010) MPL mutation effect on JAK2 46/1 haplotype frequency in JAK2V617F-negative myeloproliferative neoplasms. *Leukemia*.
- Pellegrini, S. & Dusanter-Fourt, I. (1997) The Structure, Regulation and Function of the Janus Kinases (JAKs) and the Signal Transducers and Activators of Transcription (STATs). *European Journal of Biochemistry*, **248**, 615-633.
- Pemmaraju, N., Moliterno, A.R., Williams, D.M., Rogers, O. & Spivak, J.L. (2007) The quantitative JAK2 V617F neutrophil allele burden does not correlate with thrombotic risk in essential thrombocytosis. *Leukemia*, **21**, 2210-2212.
- Perez-Encinas, M., Bello, J.L., Perez-Crespo, S., De Miguel, R. & Tome, S. (1994) Familial myeloproliferative syndrome. *Am J Hematol*, **46**, 225-229.
- Pikman, Y., Lee, B.H., Mercher, T., McDowell, E., Ebert, B.L., Gozo, M., Cuker, A., Wernig, G., Moore, S., Galinsky, I., DeAngelo, D.J., Clark, J.J., Lee, S.J., Golub, T.R., Wadleigh, M., Gilliland, D.G. & Levine, R.L. (2006) MPLW515L Is a Novel Somatic Activating Mutation in Myelofibrosis with Myeloid Metaplasia. *PLoS Med*, **3**, e270.
- Prchal, J.F. & Axelrad, A.A. (1974) Letter: Bone-marrow responses in polycythemia vera. *N Engl J Med*, **290**, 1382.
- Prchal, J.T. & Guan, Y.L. (1993) A novel clonality assay based on transcriptional analysis of the active X chromosome. *Stem Cells*, **11 Suppl 1**, 62-65.
- Radley, J. & Haller, C. (1982) The demarcation membrane system of the megakaryocyte: a misnomer? *Blood*, **60**, 213-219.

- Randi, M.L., Fabris, F. & Girolami, A. (2000) Leukemia and Myelodysplasia Effect of Multiple Cytotoxic Therapy in Essential Thrombocythemia. *Leukemia and Lymphoma*, **37**, 379 -385.
- Richter, M.F., Duménil, G., Uzé, G., Fellous, M. & Pellegrini, S. (1998) Specific Contribution of Tyk2 JH Regions to the Binding and the Expression of the Interferon α/β Receptor Component IFNAR1. *Journal of Biological Chemistry*, **273**, 24723-24729.
- Rojnuckarin, P., Drachman, J.G. & Kaushansky, K. (1999) Thrombopoietin-Induced Activation of the Mitogen-Activated Protein Kinase (MAPK) Pathway in Normal Megakaryocytes:
 Role in Endomitosis. *Blood*, **94**, 1273-1282.
- Rolovic, Z., Basara, N., Gotic, M., Sefer, D. & Bogdanovic, A. (1995) The determination of spontaneous megakaryocyte colony formation is an unequivocal test for discrimination between essential thrombocythaemia and reactive thrombocytosis. *British Journal of Haematology*, **90**, 326-331.
- Ruan, G.-R., Jiang, B., Li, L.-D., Niu, J.-H., Li, J.-L., Xie, M., Qin, Y.-Z., Liu, Y.-R., Huang, X.-J. & Chen, S.-S. (2010) MPL W515L/K mutations in 343 Chinese adults with JAK2V617F mutation-negative chronic myeloproliferative disorders detected by a newly developed RQ-PCR based on TaqMan MGB probes. *Hematological Oncology*, 28, 33-39.
- Rumi, E., Passamonti, F., Pietra, D., Porta, M.G.D., Arcaini, L., Boggi, S., Elena, C., Boveri, E., Pascutto, C., Lazzarino, M. & Cazzola, M. (2006) JAK2 (V617F) as an acquired somatic mutation and a secondary genetic event associated with disease progression in familial myeloproliferative disorders. *Cancer*, **107**, 2206-2211.
- Rumi, E., Passamonti, F., Della Porta, M.G., Elena, C., Arcaini, L., Vanelli, L., Del Curto, C., Pietra, D., Boveri, E., Pascutto, C., Cazzola, M. & Lazzarino, M. (2007) Familial Chronic
 Myeloproliferative Disorders: Clinical Phenotype and Evidence of Disease Anticipation.
 J Clin Oncol, 25, 5630-5635.
- Sacchi, S., Vinci, G., Gugliotta, L., Rupoli, S., Gargantini, L., Martinelli, V., Baravelli, S., Lazzarino, M. & Finazzi, G. (2000) Diagnosis of essential thrombocythemia at platelet counts between 400 and 600x10(9)/L. Gruppo Italiano Malattie Mieloproliferative Croniche(GIMMC). *Haematologica*, 85, 492-495.
- Saint-Martin, C., Leroy, G., Delhommeau, F., Panelatti, G., Dupont, S., James, C., Plo, I., Bordessoule, D., Chomienne, C., Delannoy, A., Devidas, A., Gardembas-Pain, M., Isnard, F., Plumelle, Y., Bernard, O., Vainchenker, W., Najman, A., Bellanne-Chantelot, C. & the French Group of Familial Myeloproliferative Disorders (2009) Analysis of the Ten-Eleven Translocation 2 (TET2) gene in familial myeloproliferative neoplasms. *Blood*, **114**, 1628-1632.
- Schaub, F.X., Jager, R., Looser, R., Hao-Shen, H., Hermouet, S., Girodon, F., Tichelli, A., Gisslinger, H., Kralovics, R. & Skoda, R.C. (2009) Clonal analysis of deletions on chromosome 20q and JAK2-V617F in MPD suggests that del20q acts independently and is not one of the predisposing mutations for JAK2-V617F. *Blood*, **113**, 2022-2027.
- Schaub, F.X., Looser, R., Li, S., Hao-Shen, H., Lehmann, T., Tichelli, A. & Skoda, R.C. (2010)
 Clonal analysis of TET2 and JAK2 mutations suggests that TET2 can be a late event in the progression of myeloproliferative neoplasms. *Blood*, **115**, 2003-2007.
- Schnittger, S., Bacher, U., Haferlach, C., Beelen, D., Bojko, P., Burkle, D., Dengler, R., Distelrath, A., Eckart, M., Eckert, R., Fries, S., Knoblich, J., Kochling, G., Laubenstein, H.-P., Petrides, P., Planker, M., Pihusch, R., Weide, R., Kern, W. & Haferlach, T. (2009)
 Characterization of 35 new cases with four different MPLW515 mutations and essential thrombocytosis or primary myelofibrosis. *Haematologica*, 94, 141-144.

- Schulze, H. & Shivdasani, R.A. (2005) Mechanisms of thrombopoiesis. *J Thromb Haemost*, **3**, 1717-1724.
- Scott, L.M., Scott, M.A., Campbell, P.J. & Green, A.R. (2006) Progenitors homozygous for the V617F mutation occur in most patients with polycythemia vera, but not essential thrombocythemia. *Blood*, **108**, 2435-2437.
- Sessarego, M., Defferrari, R., Dejana, A.M., Rebuttato, A.M., Fugazza, G., Salvidio, E. & Ajmar,
 F. (1989) Cytogenetic analysis in essential thrombocythemia at diagnosis and at transformation: A 12-year study. *Cancer Genetics and Cytogenetics*, 43, 57-65.
- Shen, Y.M., Chao, H.Y., Zhang, R., Feng, Y.F., Cen, J.N., Yao, L., Shen, H.J., Zhu, Z.L. & Xue, Y.Q. (2009) Quantitative Assay for Janus Kinase 2 (JAK2) Mutation in Chinese Patients with Chronic Myeloproliferative Disorders. *The Journal of International Medical Research*, 37, 37-46.
- Shih, L.-Y., Lin, T.-L., Lai, C.-L., Dunn, P., Wu, J.-H., Wang, P.-N., Kuo, M.-C. & Lee, L.-C. (2002) Predictive values of X-chromosome inactivation patterns and clinicohematologic parameters for vascular complications in female patients with essential thrombocythemia. *Blood*, **100**, 1596-1601.
- Shih, L. & Lee, C. (1994) Identification of masked polycythemia vera from patients with idiopathic marked thrombocytosis by endogenous erythroid colony assay. *Blood*, **83**, 744-748.
- Shivdasani, R.A., Fujiwara, Y., McDevitt, M.A. & Orkin, S.H. (1997) A lineage-selective knockout establishes the critical role of transcription factor GATA-1 in megakaryocyte growth and platelet development. *EMBO J*, **16**, 3965-3973.
- Singal, U., Prasad, A.S., Halton, D.M. & Bishop, C. (1983) Essential thrombocythemia: a clonal disorder of hematopoietic stem cell. *Am J Hematol*, **14**, 193-196.
- Skoda, R.C. (2009) Thrombocytosis. Hematology Am Soc Hematol Educ Program, 159-167.
- Snow, J.W., Abraham, N., Ma, M.C., Abbey, N.W., Herndier, B. & Goldsmith, M.A. (2002) STAT5 promotes multilineage hematolymphoid development in vivo through effects on early hematopoietic progenitor cells. *Blood*, **99**, 95-101.
- Soranzo, N., Spector, T.D., Mangino, M., Kuhnel, B., Rendon, A., Teumer, A., Willenborg, C., Wright, B., Chen, L., Li, M., Salo, P., Voight, B.F., Burns, P., Laskowski, R.A., Xue, Y., Menzel, S., Altshuler, D., Bradley, J.R., Bumpstead, S., Burnett, M.-S., Devaney, J., Doring, A., Elosua, R., Epstein, S.E., Erber, W., Falchi, M., Garner, S.F., Ghori, M.J.R., Goodall, A.H., Gwilliam, R., Hakonarson, H.H., Hall, A.S., Hammond, N., Hengstenberg, C., Illig, T., Konig, I.R., Knouff, C.W., McPherson, R., Melander, O., Mooser, V., Nauck, M., Nieminen, M.S., O'Donnell, C.J., Peltonen, L., Potter, S.C., Prokisch, H., Rader, D.J., Rice, C.M., Roberts, R., Salomaa, V., Sambrook, J., Schreiber, S., Schunkert, H., Schwartz, S.M., Serbanovic-Canic, J., Sinisalo, J., Siscovick, D.S., Stark, K., Surakka, I., Stephens, J., Thompson, J.R., Volker, U., Volzke, H., Watkins, N.A., Wells, G.A., Wichmann, H.E., Van Heel, D.A., Tyler-Smith, C., Thein, S.L., Kathiresan, S., Perola, M., Reilly, M.P., Stewart, A.F.R., Erdmann, J., Samani, N.J., Meisinger, C., Greinacher, A., Deloukas, P., Ouwehand, W.H. & Gieger, C. (2009) A genome-wide meta-analysis identifies 22 loci associated with eight hematological parameters in the HaemGen consortium. *Nature Genetics*, **41**, 1182-1190.
- Stegelmann, F., Bullinger, L., Griesshammer, M., Holzmann, K., Habdank, M., Kuhn, S., Maile, C., Schauer, S., Dohner, H. & Dohner, K. (2010) High-resolution single-nucleotide polymorphism array-profiling in myeloproliferative neoplasms identifies novel genomic aberrations. *Haematologica*, **95**, 666-669.

- Sterkers, Y., Preudhomme, C., Lai, J.-L., Demory, J.-L., Caulier, M.-T., Wattel, E., Bordessoule, D., Bauters, F. & Fenaux, P. (1998) Acute Myeloid Leukemia and Myelodysplastic
 Syndromes Following Essential Thrombocythemia Treated With Hydroxyurea: High Proportion of Cases With 17p Deletion. *Blood*, **91**, 616-622.
- Sungaran, R., Markovic, B. & Chong, B.H. (1997) Localization and Regulation of Thrombopoietin mRNA Expression in Human Kidney, Liver, Bone Marrow, and Spleen Using In Situ Hybridization. *Blood*, 89, 101-107.
- Swerdlow, S.H., Campo, E., Harris, N.L., Jaffe, E.S., Pileri, S.A., Stein, H., Thiele, J. & Vardiman, J.W. (2008) WHO Classification of Tumours of Haematopoietic and Lymphoid Tissues, Fourth Edition. IARC, Lyon.
- Tablin, F., Jain, N.C., Mandell, C.P., Hopper, P.E. & Zinkl, J.G. (1989) Ultrastructural Analysis of Platelets and Megakaryocytes from a Dog with Probable Essential Thrombocythemia. *Veterinary Pathology Online*, **26**, 289-293.
- Tahara, T., Usuki, K., Sato, H., Ohashi, H., Morita, H., Tsumura, H., Matsumoto, A., Miyazaki, H., Urabe, A. & Kato, T. (1996) A sensitive sandwich ELISA for measuring thrombopoietin in human serum: serum thrombopoietin levels in healthy volunteers and in patients with haemopoietic disorders. *Br J Haematol*, **93**, 783-788.
- Tahiliani, M., Koh, K.P., Shen, Y., Pastor, W.A., Bandukwala, H., Brudno, Y., Agarwal, S., Iyer, L.M., Liu, D.R., Aravind, L. & Rao, A. (2009) Conversion of 5-methylcytosine to 5hydroxymethylcytosine in mammalian DNA by MLL partner TET1. *Science*, **324**, 930-935.
- Taksin, A.L., Couedic, J.-P.L., Dusanter-Fourt, I., Masse, A., Giraudier, S., Katz, A., Wendling, F., Vainchenker, W., Casadevall, N. & Debili, N. (1999) Autonomous Megakaryocyte
 Growth in Essential Thrombocythemia and Idiopathic Myelofibrosis Is Not Related to a c-mpl Mutation or to an Autocrine Stimulation by Mpl-L. *Blood*, **93**, 125-139.
- Tefferi, A., Ho, T.C., Ahmann, G.J., Katzmann, J.A. & Greipp, P.R. (1994) Plasma interleukin-6 and C-reactive protein levels in reactive versus clonal thrombocytosis. *The American Journal of Medicine*, **97**, 374-378.
- Tefferi, A., Fonseca, R., Pereira, D.L. & Hoagland, H.C. (2001) A long-term retrospective study of young women with essential thrombocythemia. *Mayo Clinic Proceedings*, **76**, 22-28.
- Tefferi, A., Lasho, T.L., Wolanskyj, A.P. & Mesa, R.A. (2004) Neutrophil PRV-1 expression across the chronic myeloproliferative disorders and in secondary or spurious polycythemia. *Blood*, **103**, 3547-3548.
- Tefferi, A., Thiele, J., Orazi, A., Kvasnicka, H.M., Barbui, T., Hanson, C.A., Barosi, G., Verstovsek, S., Birgegard, G., Mesa, R., Reilly, J.T., Gisslinger, H., Vannucchi, A.M., Cervantes, F., Finazzi, G., Hoffman, R., Gilliland, D.G., Bloomfield, C.D. & Vardiman, J.W. (2007)
 Proposals and rationale for revision of the World Health Organization diagnostic criteria for polycythemia vera, essential thrombocythemia, and primary myelofibrosis: recommendations from an ad hoc international expert panel. *Blood*, **110**, 1092-1097.
- Tefferi, A. (2008) Essential thrombocythemia, polycythemia vera, and myelofibrosis: Current management and the prospect of targeted therapy. *American Journal of Hematology*, **83**, 491-497.
- Tefferi, A. & Vardiman, J.W. (2008) Classification and diagnosis of myeloproliferative neoplasms: the 2008 World Health Organization criteria and point-of-care diagnostic algorithms. *Leukemia*, **22**, 14-22.
- Tefferi, A., Pardanani, A., Lim, K.H., Abdel-Wahab, O., Lasho, T.L., Patel, J., Gangat, N., Finke, C.M., Schwager, S., Mullally, A., Li, C.Y., Hanson, C.A., Mesa, R., Bernard, O.,

Delhommeau, F., Vainchenker, W., Gilliland, D.G. & Levine, R.L. (2009) TET2 mutations and their clinical correlates in polycythemia vera, essential thrombocythemia and myelofibrosis. *Leukemia*, **23**, 905-911.

- Temerinac, S., Klippel, S., Strunck, E., Roder, S., Lubbert, M., Lange, W., Azemar, M., Meinhardt, G., Schaefer, H.E. & Pahl, H.L. (2000) Cloning of PRV-1, a novel member of the uPAR receptor superfamily, which is overexpressed in polycythemia rubra vera. *Blood*, **95**, 2569-2576.
- Teofili, L., Martini, M., Luongo, M., Di Mario, A., Leone, G., De Stefano, V. & Larocca, L.M.
 (2002) Overexpression of the Polycythemia Rubra Vera-1 Gene in Essential
 Thrombocythemia. J Clin Oncol, 20, 4249-4254.
- Teofili, L., Giona, F., Martini, M., Cenci, T., Guidi, F., Torti, L., Palumbo, G., Amendola, A., Foa, R.
 & Larocca, L.M. (2007) Markers of Myeloproliferative Diseases in Childhood
 Polycythemia Vera and Essential Thrombocythemia. J Clin Oncol, 25, 1048-1053.
- Teofili, L., Giona, F., Torti, L., Cenci, T., Ricerca, B.M., Rumi, C., Nunes, V., Foa, R., Leone, G., Martini, M. & Larocca, L.M. (2010) Hereditary thrombocytosis caused by MPLSer505Asn is associated with a high thrombotic risk, splenomegaly and progression to bone marrow fibrosis. *Haematologica*, **95**, 65-70.
- Theocharides, A., Boissinot, M., Girodon, F., Garand, R., Teo, S.-S., Lippert, E., Talmant, P., Tichelli, A., Hermouet, S. & Skoda, R.C. (2007) Leukemic blasts in transformed JAK2-V617F-positive myeloproliferative disorders are frequently negative for the JAK2-V617F mutation. *Blood*, **110**, 375-379.
- Theocharides, A., Passweg, J.R., Medinger, M., Looser, R., Li, S., Hao-Shen, H., Buser, A.S., Gratwohl, A., Tichelli, A. & Skoda, R.C. (2008) The allele burden of JAK2 mutations remains stable over several years in patients with myeloproliferative disorders. *Haematologica*, **93**, 1890-1893.
- Thiele, J., Kvasnicka, H.M., Diehl, V., Fischer, R. & Michiels, J.J. (1999) Clinicopathological Diagnosis and Differential Criteria of Thrombocythemias in Various Myeloproliferative Disorders by Histopathology, Histochemistry and Immunostaining from Bone Marrow Biopsies. Leukemia and Lymphoma, 33, 207 - 218.
- Tiedt, R., Hao-Shen, H., Sobas, M.A., Looser, R., Dirnhofer, S., Schwaller, J. & Skoda, R.C. (2008) Ratio of mutant JAK2-V617F to wild-type Jak2 determines the MPD phenotypes in transgenic mice. *Blood*, **111**, 3931-3940.
- Tiedt, R., Coers, J., Ziegler, S., Wiestner, A., Hao-Shen, H., Bornmann, C., Schenkel, J., Karakhanova, S., de Sauvage, F.J., Jackson, C.W. & Skoda, R.C. (2009) Pronounced thrombocytosis in transgenic mice expressing reduced levels of Mpl in platelets and terminally differentiated megakaryocytes. *Blood*, **113**, 1768-1777.
- Trifa, A., Cucuianu, A., Petrov, L., Urian, L., Militaru, M., Dima, D., Pop, I. & Popp, R. (2010) The G allele of the JAK2 rs10974944 SNP, part of JAK2 46/1 haplotype, is strongly associated with JAK2 V617F-positive myeloproliferative neoplasms. *Annals of Hematology*, **89**, 979-983.
- Trowbridge, E.A., Martin, J.F., Slater, D.N., Kishk, Y.T., Warren, C.W., Harley, P.J. & Woodcock, B. (1984) The origin of platelet count and volume. *Clin Phys Physiol Meas*, **5**, 145-170.
- Tsang, A.P., Visvader, J.E., Turner, C.A., Fujiwara, Y., Yu, C., Weiss, M.J., Crossley, M. & Orkin,
 S.H. (1997) FOG, a multitype zinc finger protein, acts as a cofactor for transcription
 factor GATA-1 in erythroid and megakaryocytic differentiation. *Cell*, **90**, 109-119.
- Tsantes, A.E., Dimoula, A., Bonovas, S., Mantzios, G., Tsirigotis, P., Zoi, K., Kalamara, E., Kardoulaki, A., Sitaras, N., Travlou, A., Dervenoulas, J. & Vaiopoulos, G. (2010) The role

of the Platelet Function Analyzer (PFA)-100 and platelet aggregometry in the differentiation of essential thrombocythemia from reactive thrombocytosis. *Thrombosis Research*, **125**, 142-146.

- Tsukamoto, N., Morita, K., Maehara, T., Okamoto, K., Sakai, H., Karasawa, M., Naruse, T. & Omine, M. (1994) Clonality in chronic myeloproliferative disorders defined by X-chromosome linked probes: demonstration of heterogeneity in lineage involvement. British Journal of Haematology, **86**, 253-258.
- Ugo, V., Marzac, C., Teyssandier, I., Larbret, F., Lécluse, Y., Debili, N., Vainchenker, W. & Casadevall, N. (2004) Multiple signaling pathways are involved in erythropoietinindependent differentiation of erythroid progenitors in polycythemia vera. *Experimental Hematology*, **32**, 179-187.
- van Genderen, P.J.J., Michiels, J.J., van der Poel-van de Luytgaarde, S.C.P.A.M. & van Vliet, H.H.D.M. (1994) Acquired von Willebrand disease as a cause of recurrent mucocutaneous bleeding in primary thrombocythemia: Relationship with platelet count. *Annals of Hematology*, **69**, 81-84.
- vanGenderen, P.J.J., Budde, U., Michiels, J.J., Strik, R.v. & Vliet, H.H.D.M.v. (1996) The reduction of large von Willebrand factor multimers in plasma in essential thrombocythaemia is related to the platelet count. *British Journal of Haematology*, **93**, 962-965.
- Vannucchi, A.M., Antonioli, E., Guglielmelli, P., Rambaldi, A., Barosi, G., Marchioli, R., Marfisi, R.M., Finazzi, G., Guerini, V., Fabris, F., Randi, M.L., De Stefano, V., Caberlon, S., Tafuri, A., Ruggeri, M., Specchia, G., Liso, V., Rossi, E., Pogliani, E., Gugliotta, L., Bosi, A., Barbui, T. & for the Italian Group for Malignant Hematologic Disorders of the Adult Myeloproliferative Disorder Working Party (2007) Clinical profile of homozygous JAK2 617V>F mutation in patients with polycythemia vera or essential thrombocythemia. *Blood*, **110**, 840-846.
- Vannucchi, A.M., Antonioli, E., Guglielmelli, P., Pancrazzi, A., Guerini, V., Barosi, G., Ruggeri, M., Specchia, G., Lo-Coco, F., Delaini, F., Villani, L., Finotto, S., Ammatuna, E., Alterini, R., Carrai, V., Capaccioli, G., Lollo, S.D., Liso, V., Rambaldi, A., Bosi, A. & Barbui, T. (2008a) Characteristics and clinical correlates of MPL 515W>L/K mutation in essential thrombocythemia. *Blood*, **112**, 844-847.
- Vannucchi, A.M., Antonioli, E., Guglielmelli, P., Pardanani, A. & Tefferi, A. (2008b) Clinical correlates of JAK2V617F presence or allele burden in myeloproliferative neoplasms: a critical reappraisal. *Leukemia*, 22, 1299-1307.
- Verstovsek, S., Kantarjian, H., Mesa, R.A., Pardanani, A.D., Cortes-Franco, J., Thomas, D.A.,
 Estrov, Z., Fridman, J.S., Bradley, E.C., Erickson-Viitanen, S., Vaddi, K., Levy, R. & Tefferi,
 A. (2010) Safety and Efficacy of INCB018424, a JAK1 and JAK2 Inhibitor, in
 Myelofibrosis. New England Journal of Medicine, 363, 1117-1127.
- Vizmanos, J.L., Ormazabal, C., Larrayoz, M.J., Cross, N.C. & Calasanz, M.J. (2006) JAK2 V617F mutation in classic chronic myeloproliferative diseases: a report on a series of 349 patients. *Leukemia*, **20**, 534-535.
- Vogelstein, B., Fearon, E., Hamilton, SR & Feinberg, A. (1985) Use of restriction fragment length polymorphisms to determine the clonal origin of human tumors. *Science*, **227**, 642-645.
- Vogelstein, B., Fearon, E.R., Hamilton, S.R., Preisinger, A.C., Willard, H.F., Michelson, A.M., Riggs, A.D. & Orkin, S.H. (1987) Clonal analysis using recombinant DNA probes from the X-chromosome. *Cancer Res*, **47**, 4806-4813.

- Vyas, P., Ault, K., Jackson, C.W., Orkin, S.H. & Shivdasani, R.A. (1999) Consequences of GATA-1 deficiency in megakaryocytes and platelets. *Blood*, **93**, 2867-2875.
- Walsh, P.S., Erlich, H.A. & Higuchi, R. (1992) Preferential PCR amplification of alleles: mechanisms and solutions. *Genome Research*, **1**, 241-250.
- Wang, J.C., Chen, C., Novetsky, A.D., Lichter, S.M., Ahmed, F. & Friedberg, N.M. (1998) Blood Thrombopoietin Levels in Clonal Thrombocytosis and Reactive Thrombocytosis. *The American Journal of Medicine*, **104**, 451-455.
- Wernig, G., Mercher, T., Okabe, R., Levine, R.L., Lee, B.H. & Gilliland, D.G. (2006) Expression of Jak2V617F causes a polycythemia vera-like disease with associated myelofibrosis in a murine bone marrow transplant model. *Blood*, **107**, 4274-4281.
- Wiestner, A., Schlemper, R.J., van der Maas, A.P.C. & Skoda, R.C. (1998) An activating splice donor mutation in the thrombopoietin gene causes hereditary thrombocythaemia. *Nature Genetics*, **18**, 49-52.
- Wolanskyj, A.P., Lasho, T.L., Schwager, S.M., McClure, R.F., Wadleigh, M., Lee, S.J., Gilliland, D.G. & Tefferi, A. (2005) JAK2 V617F mutation in essential thrombocythaemia: clinical associations and long-term prognostic relevance. *British Journal of Haematology*, **131**, 208-213.
- Wolanskyj, A.P., Schwager, S.M., McClure, R.F., Larson, D.R. & Tefferi, A. (2006) Essential Thrombocythemia Beyond the First Decade: Life Expectancy, Long-term Complication Rates, and Prognostic Factors. *Mayo Clinic Proceedings*, **81**, 159-166.
- Wong, G.C., Kam, G.L. & Koay, E.S. (2010) JAK2 Mutations in Asian Patients with Essential Thrombocythaemia. *Internal Medicine Journal*, no-no.
- Xu, W., Li, J.-Y., Xia, J., Zhang, S.-J., Fan, L. & Qiao, C. (2008) MPL W515L mutation in Chinese patients with myeloproliferative diseases. *Leukemia & Lymphoma*, **49**, 955-958.
- Xu, X., Zhang, Q., Luo, J., Xing, S., Li, Q., Krantz, S.B., Fu, X. & Zhao, Z.J. (2007) JAK2V617F: prevalence in a large Chinese hospital population. *Blood*, **109**, 339-342.
- Zanjani, E.D., Lutton, J.D., Hoffman, R. & Wasserman, L.R. (1977) Erythroid colony formation by polycythemia vera bone marrow in vitro. Dependence on erythropoietin. *The Journal of Clinical Investigation*, **59**, 841-848.
- Zauli, G., Visani, G., Catani, L., Vianelli, N., Gugliotta, L. & Capitani, S. (1993) Reduced responsiveness of bone marrow megakaryocyte progenitors to platelet-derived transforming growth factor beta 1, produced in normal amount, in patients with essential thrombocythaemia. *Br J Haematol*, **83**, 14-20.
- Zhao, R., Xing, S., Li, Z., Fu, X., Li, Q., Krantz, S.B. & Zhao, Z.J. (2005) Identification of an Acquired JAK2 Mutation in Polycythemia Vera. *Journal of Biological Chemistry*, **280**, 22788-22792.